**T2R2**東京工業大学リサーチリポジトリ Tokyo Tech Research Repository

## 論文 / 著書情報 Article / Book Information

題目(和文)	いくつかの基本的なH 制御問題に対する低次元ロバスト補償器の設計 に関する研究
Title(English)	Designing low-order robust controllers with solving some fundamental H control problems
著者(和文)	渡辺隆男
Author(English)	
出典(和文)	学位:博士(工学), 学位授与機関:東京工業大学, 報告番号:乙第3327号, 授与年月日:1999年6月30日, 学位の種別:論文博士, 審査員:美多 勉
Citation(English)	Degree:Doctor (Engineering), Conferring organization: Tokyo Institute of Technology, Report number:乙第3327号, Conferred date:1999/6/30, Degree Type:Thesis doctor, Examiner:
学位種別(和文)	
Type(English)	Doctoral Thesis

A Thesis Submitted to Tokyo Institute of Technology

## Designing Low-order Robust Controllers with Solving Some Fundamental $H_{\infty}$ Control Problems

Takao Watanabe

May 1999

## Abstract

 $H_{\infty}$  control is one of the most effective methodologies in robust control, and much attention has been paid to it in recent years. So far, the  $H_{\infty}$  controller has been given with a dynamical order that is equivalent to the sum of the orders of a plant and weighting functions. However it is frequently too high to use in practice. Under these circumstances, this thesis considers the problem of designing low-order  $H_{\infty}$  controllers. It considers two kinds of approaches: a *direct* one in which a low-order controller is derived directly in the process of the controller design, and an *indirect* one in which a high-order controller is designed first and then approximated to a lower-order one.

First, this thesis starts with non-standard  $H_{\infty}$  control problems, which have certain key structures for reducing the dynamical order of the controllers. A class of low-order controllers for the non-standard  $H_{\infty}$  problems is derived directly based on an algebraic operation in a class of full-order controllers. This derivation indicates that it is possible to design low-order controllers for problems that are reducible to one of the non-standard  $H_{\infty}$  problems. So far, people have usually avoided treating the non-standard  $H_{\infty}$  problems and they have been solved after being transformed into standard one. This study is a new attempt to utilize the non-standard  $H_{\infty}$  problems.

Second, this thesis also shows that some representative control system design problems are reducible to the non-standard problems, and proposes new formulations for solving these problems. It then proposes methods of designing low-order controllers for two kinds of problems: robust servo controller design and two-degree-of-freedom controller design. It is found that the low-order controllers can be designed based on the new formulations. Besides being used to design low-order controllers, the new formulations make it easier to design controllers than formulations based on the standard  $H_{\infty}$  problems.

Lastly, this thesis also considers an *indirect* approach that treats a specialized solution of the numerically solved  $H_{\infty}$  controller, which is represented by a linear fractional transformation of an optimized Youla parameter. In many cases the numerical approach yields an extremely high-order Youla parameter and the  $H_{\infty}$  controller becomes high order. We propose reducing the order of the  $H_{\infty}$  controller by reducing the order of the Youla parameter. One of advantages of the reduction method is that the resultant reduced-order controller satisfies closed loop properties: internal stability and closed loop pole specification.

## Acknowledgments

The present thesis is the outcome of my research done at Tokyo Metropolitan University, and it was completed with the support of many people.

First, I would like to express my gratitude to Professor Keiichiro Yasuda in Tokyo Metropolitan University for his constant support, daily guidance, encouragement, and understanding. I would like to thank him for giving me many valuable comments and discussions on this research, and for revising the manuscript. Also, I very much appreciate his guidance about studying in the field of control engineering.

I would like to acknowledge Professor Tsutomu Mita in Tokyo Institute of Technology for many valuable comments and discussions throughout the thesis and for revising the manuscript several times. A key idea in this study is due to his presentation and comments at the SICE symposium on control theory. Also, I would like to thank him for giving me an opportunity to submit my thesis to Tokyo Institute of Technology, and for judging the thesis as the chief investigator of the judging committee for my doctoral degree.

I would also like to acknowledge the members of the judging committee in Tokyo Institute of Technology: Professor Katsuhisa Furuta for his enthusiastic discussions and many valuable comments, Professor Tetsuya Iwasaki for essential comments on the low-order controller design and for numerical examples, Professor Yoshiki Matsuo for a very important suggestion from the view point of practical use of the results, and Professor Masaki Yamakita for many valuable discussions and helpful comments.

Finally, I would like to thank my parents for their support and encouragement and Ms. Kumiko Nakada for her encouragement.

> Takao Watanabe May 1999

## Notation

$\mathbb{Z}$	set of integers
$\mathbb{R}$	set of real numbers
$\mathbb{C}$	set of complex numbers
$\mathbb{F}^{m  imes n}$	set of $m \times n$ matrices on $\mathbb{F}$ , where $\mathbb{F}$ might be $\mathbb{Z}$ or $\mathbb{R}$ or $\mathbb{C}$ .
$\operatorname{Re}(c)$	real part of $c \in \mathbb{C}$
$\operatorname{Im}(c)$	imaginary part of $c \in \mathbb{C}$
$I_n$	$n \times n$ identity matrix
0	zero matrix
$\operatorname{diag}(a_1,\ldots,a_n)$	an $n \times n$ diagonal matrix with $a_i$ as its <i>i</i> -th diagonal element
$M^T$	transpose of a matrix $M \in \mathbb{R}^{m \times n}$
$M^{\dagger}$	pseudo inverse matrix of $M$
$M^{\perp}$	orthogonal complement of $M$
$\ker M$	$\{x\in\mathbb{R}^n     Mx=0\}$
$\operatorname{range} M$	$\{y \in \mathbb{R}^m \mid y = Mx, x \in \mathbb{R}^n\}$
$\lambda_i(A)$	an eigenvalue of a square matrix $A \in \mathbb{R}^{n \times n}$ , where $i = [1, 2, \cdots, n]$
$\lambda_{max}(A)$	the largest eigenvalue of a matrix $A$
$\lambda_{min}(A)$	the smallest eigenvalue of a matrix $A$
$\rho(A)$	spectral radius of a matrix $A$
$\sigma_{max}(A)$	the largest singular value of a matrix $A$
$\operatorname{trace} A$	trace of a matrix $A$
$\sim$	equivalent under nonsingular transformations
$\mathcal{RH}_\infty$	set of proper stable real rational transfer functions
$\ G(s)\ _{\infty}$	$H_{\infty}$ norm of a transfer function $G(s) \in \mathcal{RH}_{\infty}$
$\mathcal{BH}_\infty$	set of bounded transfer functions in $\mathcal{RH}_{\infty}$
	defined as
$ \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix} $	$\begin{cases} a \text{ system described with the state space equation} \\ x = Ax + Bu \\ y = Cx + Du \\ a \text{ system descrived with the transfer function } D + C(sI - A)^{-1}B \end{cases}$
Dom(C)	$\{s \in \mathbb{C} \mid  s + \alpha  = \beta, \operatorname{Re}(s) < 0\}$
Dom(D)	$\{s \in \mathbb{C} \mid  s + \alpha  < \beta, \operatorname{Re}(s) < 0\}$

Some of transformations that will be used in this thesis are defined as follows.

For a matrix partitioned as

$$G = \left[ \begin{array}{cc} G_{11} & G_{12} \\ G_{21} & G_{22} \end{array} \right],$$

we define a linear fractinal transformation (LFT), a homographic transformation (HMT), and a dual homographic transformation (DHM) as follows.

• If  $det(I - G_{22}Q) \neq 0$ , then the LFT is defined and is denoted as

$$\mathcal{F}_l(G,Q) = G_{11} + G_{12}Q(I - G_{22}Q)^{-1}G_{21}.$$

• If  $det(G_{21}Q + G_{22}) \neq 0$ , then the HMT is defined and is denoted as

$$\mathcal{HM}(G,Q) = (G_{11}Q + G_{12})(G_{21}Q + G_{22})^{-1}.$$

• If  $det(G_{11} + QG_{21}) \neq 0$ , then the DHM is defined and is denoted as

$$\mathcal{DHM}(G,Q) = (G_{11} + QG_{21})^{-1}(G_{12} + QG_{22}).$$

Those transformations are related with each other by an equation:

$$\mathcal{F}_{l}(G,Q) = \mathcal{H}\mathcal{M}\left(\begin{bmatrix} G_{12} - G_{11}G_{21}^{-1}G_{22} & G_{11}G_{21}^{-1} \\ -G_{21}^{-1}G_{22} & G_{21}^{-1} \end{bmatrix}, Q\right)$$
$$= \mathcal{D}\mathcal{H}\mathcal{M}\left(\begin{bmatrix} G_{12}^{-1} & G_{12}^{-1}G_{11} \\ -G_{22}G_{12}^{-1} & G_{21} - G_{22}G_{12}^{-1}G_{11} \end{bmatrix}, Q\right).$$

viii

## Contents

A	bstra	stract		
A	ckno	wledgements	vi	
N	Notation			
1	Intr	ntroduction		
	1.1	Overview of the thesis	1	
	1.2	Studies of the low-order controller design	2	
	1.3	The non-standard $H_{\infty}$ problems $\ldots \ldots \ldots$	3	
		1.3.1 Problem description	3	
		1.3.2 Assumptions	4	
		1.3.3 Past studies on the non-standard $H_{\infty}$ problems	5	
	1.4	Related issue to the non-standard problem	6	
		1.4.1 Fundamental non-standard solution via reduced-order observer design	6	
		1.4.2 Integral-type low-order robust controller design $\ldots \ldots \ldots \ldots$	8	
		1.4.3 Low-order TDF controller design	11	
	1.5	$H_{\infty}$ controller reduction	12	
	1.6	Contribution	14	
	1.7	Organization	15	
<b>2</b>	Rec	luced-order non-standard $H_\infty$ controller design	17	
	2.1	Introduction	17	
	2.2	Formulation	18	
2.3 Preliminaries		Preliminaries	19	
		2.3.1 Pseudo inverse matrix and orthogonal complement matrix	19	
		2.3.2 Invariant zeros	19	
		2.3.3 Canonical transformation of the generalized plant	20	
	2.4	The non-standard $H_{\infty}$ problem of case 2	22	
		2.4.1 Characterization of zeros in $G_{21}(s)$	22	
		2.4.2 A necessary condition for the solvability	24	

#### CONTENTS

		2.4.3	Lossless factorization of $G(s)$	27
		2.4.4	Parametrization of full-order $H_{\infty}$ controller $\ldots \ldots \ldots \ldots \ldots$	29
		2.4.5	Derivation of reduced-order $H_{\infty}$ controller	33
		2.4.6	The controller structure	35
	2.5	The n	on-standard $H_{\infty}$ problem of case 1	36
		2.5.1	Characterization of zeros in $G_{12}(s)$	37
		2.5.2	A necessary condition for the solvability $\ldots \ldots \ldots \ldots \ldots \ldots$	39
		2.5.3	Parametrization of full-order $H_{\infty}$ controller $\ldots \ldots \ldots \ldots \ldots$	41
		2.5.4	Derivation of reduced-order $H_{\infty}$ controller	42
		2.5.5	The controller structure	44
	2.6	The n	on-standard $H_{\infty}$ problem of case 3	46
		2.6.1	Parametrization of full-order $H_{\infty}$ controller $\ldots \ldots \ldots \ldots \ldots$	46
		2.6.2	Derivation of reduced-order $H_{\infty}$ controllers based on characterization	
			of zeros in $G_{21}(s)$	52
		2.6.3	Derivation of reduced-order $H_{\infty}$ controllers based on characterization	
			of zeros in $G_{12}(s)$	54
	2.7	A nun	nerical example and discussions	58
		2.7.1	Magnetic levitation system	58
		2.7.2	An uncertain plant with partial state measurement	60
		2.7.3	Low order robust controller design	61
		2.7.4	Discussion	63
	2.8	Summ	nary	65
3	A s	ynthes	is of low-order integral-type controller	67
	3.1	Introd	luction	67
	3.2	Robus	st servo controller design	68
		3.2.1	Specifications	68
		3.2.2	Formulation with the mixed sensitivity problem	69
	3.3	In the	e case $P(s)$ has no $j\omega$ -poles $\ldots \ldots \ldots$	70
		3.3.1	Formulation	70
		3.3.2	A high-order controller design	72
		3.3.3	A low-order controller design	73
		3.3.4	A direct derivation of an integral-type $H_{\infty}$ controller	76
	3.4	In the	e case $P(s)$ has $j\omega$ -poles	83
		3.4.1	The $j\omega$ invariant zeros of $G_{21}(s)$	83
		3.4.2	Design	83
	3.5	Nume	rical examples	89
		3.5.1	$P(s)$ has no $j\omega$ -poles	89
		3.5.2	$P(s)$ has $j\omega$ -poles	90

#### CONTENTS

	3.6	Summary	90
4	Tra	de-off analysis of a low-order TDF control system	93
	4.1	Introduction	93
	4.2	Basic analysis and design of the TDF control system	94
		4.2.1 Basic analysis of the TDF control system	94
		4.2.2 A basic design of the TDF controller	96
	4.3	An idea of sharing common dynamics	98
		4.3.1 Sharing common dynamics between $K(s)$ and $F(s)$	98
		4.3.2 A basic design of a low-order TDF controller	99
	4.4	Trade-off analysis in the low-order TDF control system	101
		4.4.1 Trade-off between the feedback performance and the feedforward per-	
		formance	101
	4.5	Summary	103
<b>5</b>	$\mathbf{A} \mathbf{s}$	ynthesis of low-order TDF controller	105
	5.1	Introduction	105
	5.2	Problem descriptions and comparison	105
		5.2.1 Two-step design $\ldots$	106
		5.2.2 Simultaneous design	108
	5.3	Low-order TDF controller design	110
		5.3.1 Control specification and construction of the generalized plant $\ldots$	110
		5.3.2 State-space representation of the generalized plant	112
		5.3.3 Parametrization of all $H_{\infty}$ controllers	114
		5.3.4 Reduced-order TDF controller design	117
	5.4	A numerical example	121
		5.4.1 Magnetic levitation system	121
		5.4.2 Description of perturbed models	122
		5.4.3 Designing of low-order integral-type TDF controller	123
		5.4.4 Comparison with another method	125
	5.5	Summary	130
6	$H_{\infty}$	controller approximation	131
	6.1	Introduction	131
	6.2	A numerical approach to $H_{\infty}$ controller design $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	132
		6.2.1 Youla parametrization	132
		6.2.2 A numerical approach to controller design	134
		6.2.3 A defect of the numerical approach	135
	6.3	The controller reduction by the approximation of $Q(s)$	136

xi

#### CONTENTS

		6.3.1	Balanced truncation	137
		6.3.2	Model approximation with constraint of pole position $\ldots \ldots \ldots$	138
		6.3.3	The $H_{\infty}$ controller reduction $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	142
	6.4	A num	erical example	144
	6.5	Summ	ary and discussion	145
7	Con	clusio	a	149
Bi	bliog	raphy		155
$\mathbf{A}_{\mathbf{j}}$	ppen	dix		155
A Preliminary results				
	A.1	Invaria	ant zeros	157
	A.2	Param	etrization of stabilizing controllers	157
	A.3	A mat	rix equation $\ldots$	158
	A.4	Star p	roduct	158
	A.5	Inner	function $\ldots$	159
	A.6	Lossle	ss system	159
	A.7	Distur	bance feedforward problem	160
	A.8	ARE (	Algebraic Riccati Equation)	160
	A.9	Solutio	on to Full control problem	161
	A.10	Reduc	tion mode of an ARE for the standard problem $\ldots \ldots \ldots \ldots \ldots$	162
	A.11	Soluti	ons to the mixed-sensitivity problem	163
в	Pub	licatio	ns	165

#### xii

## Chapter 1 Introduction

#### 1.1 Overview of the thesis

With the advances in technology, the role of control systems has become more important and they are required to act more precise and to perform robust in the presence of uncertainties of plants. In recent years, a robust control theory has developed and high-performance controllers have become easy to design. On the other hand, the resultant controllers are derived with high order [55], and they are sometimes hard to implement [1]. Thus, designing a low-order controller is an important problem that must be settled [1]. Although the designing of low-order robust controllers has been the subject of many studies, it is still an unsettled problem. In this thesis, we are devoted to the issue of designing the low-order robust controllers.

In the first place, this thesis focuses on a non-standard  $H_{\infty}$  control problem that is a key to the low-order robust controller design. In the non-standard problem, a class of plants for which we consider the controller design includes plants that can be stabilized with minimalorder-observer-based feedback controllers [41, 42, 30]. Hence, it is hoped that approaching the non-standard problem from the viewpoint of the minimal-order-observer-design enables us to develop a method for a wider class of low-order robust controller designs. Thus, the main result of this thesis provides a complete characterization of low-order  $H_{\infty}$  controllers for the non-standard problem.

Although, in this thesis, the non-standard problem is a fundamental topic to give methods for designing the low-order robust controllers, it is too abstract to make the most of practically. Hence, a question "What sorts of practical problems of robust controller design reduce to the non-standard  $H_{\infty}$  problem ?" may arise. To answer this question, the class of the practical problems that are reducible to the non-standard problem should be clarified. This thesis investigates the problems of an integral-type robust controller design and a two-degree-offreedom (TDF) controller design. It then proposes methods for designing the low-order  $H_{\infty}$ controllers in those two kinds of problems.

This thesis also considers an *indirect approach*, in which a model approximation method

is used to reduce the order of the controller. In the studies of control application, there are many cases that have adopted the indirect approach. Nevertheless, there are few theoretical results about the reduction of the robust controller. Hence, there is no choice but to rely on the rule of trial and error for reducing the controller. In this thesis, we shall focus on a numerical solution of the  $H_{\infty}$  controller that is derived with the so-called Youla-parameter approach [69, 3, 43], where it is known that a resultant  $H_{\infty}$  controller becomes high-order. However, there is no theoretical result for reducing the order of such a controller. Under the circumstance, this thesis proposes a way to reduce the order of the controller and gives a sufficient condition for the approximated controller satisfies the closed loop properties: internal stability and some closed loop pole specification.

#### 1.2 Studies of the low-order controller design

A problem of designing the low-order controller has been an interest of many researchers for long time, however it is still an open problem. As stated in an earlier monograph [1], there are two approaches to designing low-order controllers for high-order plants. One is a *direct approach*, in which the parameters of the low-order controller are computed by some optimization procedure [2, 33]. The other is an *indirect approach*, in which a high-order controller is designed first and then the dynamical order of the controller is reduced by using some model approximation method [52, 18, 56].

Designing low-order controllers with the *direct approach* [2, 28, 57, 8, 9, 46] is a difficult problem. However, in recent years, a numerical method [3, 16, 32] for control system design has provided a new way for designing low-order controllers directly, and in this way the condition for existence of a fixed-order controller is expressed with an LMI (Linear Matrix Inequality) with a constraint of matrix rank [33, 63, 31]. Nevertheless, it is still difficult to solve the inequality numerically because solving the inequality reduces to a non-convex feasibility problem. Hence the approach via the numerical method requires further studies of the numerical algorithm [57, 8, 4, 46], and it is still important that the way to derive directly the low-order robust controller be studied from another point of view. On the other hand, the present study slightly differs from the earlier studies. It first clarifies a class of problems for which the low-order controllers can be designed and then analyzes some synthesis problems that are reducible to that class.

Essentially, the controller design with evaluation of the closed loop performances and the controller reduction with evaluation of the input-output properties of the controller itself are independent problems [1]. Hence, it is not always true that the good approximation of the controller in a sense of input-output properties means the good approximation of the closed loop properties. Although many model reduction methods [1, 52, 72, 73, 62, 63] are available, there are difficulties in preserving the closed loop performances of an original controller. In recent years, Goddard [20] proposed a method for the controller approximation

with preserving closed loop performances. This thesis also considers an *indirect approach* to designing the low-order controller, and it treats the controller order reduction in such a way that the required closed loop performances are assured.

#### **1.3** The non-standard $H_{\infty}$ problems

This section briefly introduces the non-standard  $H_{\infty}$  problems, which give a key to the designing of the low-order robust controllers, and refers to past studies concerned with the non-standard  $H_{\infty}$  problems.

#### 1.3.1 Problem description

This thesis treats the following linear, time-invariant, finite dimensional system:

$$\begin{cases} \dot{x}(t) = Ax(t) + B_1 w(t) + B_2 u(t) \\ z(t) = C_1 x(t) + D_{12} u(t) \\ y(t) = C_2 x(t) + D_{21} w(t) \end{cases}$$
(1.1)

where  $x(t) \in \mathbb{R}^n$  is the state variable,  $z(t) \in \mathbb{R}^{p_1}$  is the controlled output,  $y(t) \in \mathbb{R}^{p_2}$  is the measurement output,  $w(t) \in \mathbb{R}^{m_1}$  is the exogenous (disturbance) input and  $u(t) \in \mathbb{R}^{m_2}$ is the control input. The matrices  $A, B_1, B_2, C_1, C_2, D_{12}, D_{21}$  are real, constant and of the appropriate dimensions. The system in (1.1) is called a generalized plant, and the system is represented as follows.

$$G(s) = \begin{bmatrix} G_{11}(s) & G_{12}(s) \\ G_{21}(s) & G_{22}(s) \end{bmatrix} = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix}$$
(1.2)

The purpose of control is to design a controller K(s) that is connected with the generalized plant as

$$u = K(s)y.$$

A closed loop system composed of the generalized plant and the controller is illustrated in Figure 1.1. Then, the transfer function of the closed loop system is represented as



Figure 1.1: A closed loop system

$$T_{zw}(s) \triangleq \mathcal{F}_l(G(s), K(s))$$

**Remark 1.3.1** The components of w(t) are all the exogenous inputs: references, disturbances, sensor noise, and so on. The components of z(t) are all the signals we wish to control: tracking errors between reference signals and plant outputs, actuator signals whose values must be kept between certain limits, and so on. The vector y(t) contains the outputs of all sensors. Finally u(t) contains all controlled inputs to the generalized plant.

The  $H_{\infty}$  control problem is stated as follows.

**Definition 1.3.1** (The  $H_{\infty}$  control problem) Determine whether or not there exists a controller K(s) such that the closed loop system is internally stable and satisfies

$$||T_{zw}(s)||_{\infty} < 1,$$

where the  $H_{\infty}$  norm of a transfer function  $F(s) \in \mathcal{RH}_{\infty}$  is defined as

$$\|F\|_{\infty} \triangleq \sup_{\operatorname{Re}(s)>0} \sigma_{max} \left[F(s)\right] = \sup_{\omega \in \mathbb{R}} \sigma_{max} \left[F(j\omega)\right].$$

Parametrize all such controllers when one exists.

**Remark 1.3.2** Many of the synthesis problems in the control engineering can be reduced to the  $H_{\infty}$  control problem [10, 12]. For example, problems of designing controllers which provides robust stability, robust performances, low-sensitivity, and so on are all reducible to the  $H_{\infty}$  problem.

#### 1.3.2 Assumptions

In earlier papers [11, 19], the  $H_{\infty}$  problem is considered under the following assumptions with regard to the system (1.1).

#### $\mathbf{A1}$

1)  $(A, B_2)$  is stabilizable.

**2)**  $(A, C_2)$  is detectable.

#### $\mathbf{A2}$

1)  $\operatorname{rank}(D_{12}) = m_2$ . ( $D_{12}$  is of full column rank.)

**2)**  $rank(D_{21}) = p_2$ . ( $D_{21}$  is of full row rank.)

#### **A3**

- 1) the system  $G_{12}(s)$  has no invariant zeros on the imaginary axis.
- 2) the system  $G_{21}(s)$  has no invariant zeros on the imaginary axis.

#### 1.3. THE NON-STANDARD $H_{\infty}$ PROBLEMS

Assumption A1 concerns the existence of a stabilizing controller for the system (1.1). The two parts of assumption A2 mean that there are no inputs in u which have the same effect on the output, that the exogenous signals in w affect the states, and that the true states are not directly measurable. Under the assumptions of A2, the assumption A3-1) is equivalent to the condition such that

$$\operatorname{rank}\left(\left[\begin{array}{cc} A-j\omega I & B_2\\ C_1 & D_{12} \end{array}\right]\right)=n+m_2, \ \forall \omega \in \mathbb{R}$$

and the assumption A3-2) is equivalent to the condition such that

$$\operatorname{rank}\left(\left[\begin{array}{cc} A-j\omega I & B_1\\ C_2 & D_{21} \end{array}\right]\right)=n+p_2, \ \forall \omega \in \mathbb{R}$$

(See appendix A.1.)

The system (1.1) satisfying these assumptions is called **the standard generalized plant** and the corresponding  $H_{\infty}$  problem is called **the standard**  $H_{\infty}$  **problem**. If one of these assumptions is not satisfied, the system (1.1) is called **the non-standard generalized plant** and the corresponding  $H_{\infty}$  problem is called **the non-standard**  $H_{\infty}$  **problem**.

#### **1.3.3** Past studies on the non-standard $H_{\infty}$ problems

Many studies of the  $H_{\infty}$  control treat the standard  $H_{\infty}$  problem [11, 19], however we are sometimes faced with the non-standard problems in which the generalized plant does not satisfy one of the assumptions: **A1**, **A2** and **A3**. This thesis especially treats the nonstandard  $H_{\infty}$  control problem where the assumption **A2** is not satisfied, that is, the rank conditions of the direct feed-through terms of the subsystems  $G_{12}(s)$  and  $G_{21}(s)$  are not satisfied.

The non-standard problems are treated in several papers [35, 51, 60, 50, 64, 65]. Stoorvogel [65] and Sampei et al. [60] studied the non-standard  $H_{\infty}$  control problems by using the Riccati inequalities, and they derived an  $H_{\infty}$  controller of McMillan degree *n* which is identical with that of the generalized plant. Kimura et al. [35] treated the non-standard  $H_{\infty}$ control problems by using the concept of J-lossless, and they derived the  $H_{\infty}$  controller which is represented with a free parameter. Furthermore, the LMI approach [16, 32, 31] enabled us to solve the controller design problems without being conscious of almost all the assumptions.

The correspondence between the non-standard  $H_{\infty}$  problem and the low-order controller design was first pointed out by Zhang and Hosoe [71], and Stoorvogel, Sabeli and Chen [65] respectively. Zhang et al. derived a low-order controller for the  $H_{\infty}$  problem where the partial states of the real plant are measurable without noise. The discussion in their article is, however, restricted to the plant in which partial states are measurable without noise and is fairly difficult. On the other hand, Stoorvogel et al. derived a low-order controller for the non-standard  $H_{\infty}$  problem. Their discussion bases a matrix inequality approach and the minimal-order observer design, however, the resultant controller is not parametrized. Recently, Xin et al. [68] studied the designing of the low-order controller for continuous and discrete time singular  $H_{\infty}$  control problem based on LMI. However, the controller is not parametrized with free parameters and its structure is unclear.

Besides the problem of designing the low-order controller, problems in utilizing the nonstandard  $H_{\infty}$  control for practical problems have not been tackled in any study, so far. In the application of the standard  $H_{\infty}$  problem, it is known what kind of the control problems is reducible to the standard problem. On the other hand, the problem in the application of the non-standard  $H_{\infty}$  control is an open problem. Thus, first, this thesis aims at constructing a general way of designing the low-order  $H_{\infty}$  controller for the problem of the non-standard  $H_{\infty}$  control, second, it aims at clarifying what kind of practical control problems is reducible to the non-standard problem, then a method of designing the low-order robust controller is to be shown.

#### 1.4 Related issue to the non-standard problem

This section looks into the non-standard  $H_{\infty}$  problem from the viewpoint of the low-order controller design and its applications. It will give us understanding of backgrounds of discussions in later sections.

#### 1.4.1 Fundamental non-standard solution via reduced-order observer design

Consider a generalized plant that is simplified as

$$\begin{cases} \begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} B_{11} \\ B_{12} \end{bmatrix} w + \begin{bmatrix} B_{21} \\ B_{22} \end{bmatrix} u \\ z = C_1 \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + D_{12}u & , \qquad (1.3) \\ \begin{bmatrix} y_1 \\ y_2 \end{bmatrix} = \begin{bmatrix} I_{p_2-m_1} & O \\ O & C_{22} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} O \\ I_{m_1} \end{bmatrix} w$$

where it is assumed that the triple  $\begin{pmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{pmatrix}$ ,  $\begin{bmatrix} B_{21} \\ B_{22} \end{bmatrix}$ ,  $\begin{bmatrix} I_{p_2-m_1} & O \\ O & C_{22} \end{bmatrix}$ ) is stabilizable and detectable. Here, it should be noted that the matrix  $\begin{bmatrix} O \\ I_{m_1} \end{bmatrix}$  is of full column rank, hence the system is a generalized plant of the non-standard problem where the assumption **A2-2**) is not satisfied. Although this system seems like a special case of the non-standard generalized plant, it can be given by an equivalent transformation of the generalized plant (1.1). In this system, it should be noted that a partial state variable  $x_1 \in \mathbb{R}^{p_2-m_1}$  is directly observed through the observation of a measurement output  $y_1$ . In order to use information of each state variable, there is therefore no need to estimate the partial state variable  $x_1$  and the dynamics of the observer can be reduced. Actually, since the state variable  $x_1$  and the

#### 1.4. RELATED ISSUE TO THE NON-STANDARD PROBLEM

input u are available, we can describe the system in the following form.

$$\begin{cases} \dot{x}_2 = A_{22}x_2 + B_{12}w + \begin{bmatrix} A_{21} & B_{22} \end{bmatrix} \begin{bmatrix} y_1 \\ u \end{bmatrix} \\ y_2 = C_{22}x_2 + I_{m_1}w \end{cases}$$
(1.4)

This system represents a dynamical system of the state variable  $x_2$ . The detectability of the pair  $\begin{pmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{pmatrix}$ ,  $\begin{bmatrix} I_{p_2-m_1} & O \\ O & C_{22} \end{bmatrix}$ ) means that the pair  $(A_{22}, C_{22})$  is detectable. Hence, all the state variables in the system (1.3) can be estimated by using a reduced-order observer of the form:

$$\begin{cases} \dot{\hat{x}}_2 = (A_{22} - HC_{22})\,\hat{x}_2 + \begin{bmatrix} A_{21} & H \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \end{bmatrix} + B_{22}u \\ \hat{x} = \begin{bmatrix} O \\ I \end{bmatrix} \hat{x}_2 + \begin{bmatrix} I & O \\ O & O \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \end{bmatrix} , \qquad (1.5)$$

where the matrix  $A_{22} - HC_{22}$  is stable. On the other hand, from the stabilizability assumed above, if all the states of the system (1.3) are estimated, the system is stabilized with a state feedback controller of the form:

$$u = \begin{bmatrix} F_1 & F_2 \end{bmatrix} \hat{x}$$

where  $\hat{x}$  is an estimate of the state and the feedback gain  $\begin{bmatrix} F_1 & F_2 \end{bmatrix}$  stabilizes the matrix  $\begin{bmatrix} A_{11} + B_{21}F_1 & A_{12} + B_{21}F_2 \\ A_{21} + B_{22}F_1 & A_{22} + B_{22}F_2 \end{bmatrix}$ . From these, it can be verified that a reduced-order-observer-



Figure 1.2: A closed loop system

based output feedback controller:

$$\begin{pmatrix} \dot{\hat{x}}_2 = (A_{22} - HC_{22}) \hat{x}_2 + \begin{bmatrix} A_{21} & H \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \end{bmatrix} + B_{22}u \\ u = F_2 \hat{x}_2 + \begin{bmatrix} F_1 & O \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \end{bmatrix}$$
(1.6)

stabilizes the system (1.3), because the A-matrix of the closed loop system composed of (1.3) and (1.6) is represented as

$$\begin{bmatrix} A_{11} + B_{21}F_1 & A_{12} & B_{21}F_2 \\ A_{21} + B_{22}F_1 & A_{22} & B_{22}F_2 \\ A_{21} + B_{22}F_1 & HC_{22} & A_{22} - HC_{22} + B_{22}F_2 \end{bmatrix} \\ \sim \begin{bmatrix} A_{11} + B_{21}F_1 & A_{12} + B_{21}F_2 & B_{21}F_2 \\ A_{21} + B_{22}F_1 & A_{22} + B_{22}F_2 & B_{22}F_2 \\ \hline O & O & A_{22} - HC_{22} \end{bmatrix},$$

and is verified to be stable. By using the observer design, we can obtain an  $H_{\infty}$  controller [15]. The closed loop system is illustrated in Figure 1.2.

Thus, the non-standard problem corresponds to a problem of the minimal-order-observerbased controller design and is a key problem in designing a low-order robust controller. It is well known that the standard  $H_{\infty}$  controller can be given with an LFT (Linear Fractional Transformation) of a central solution and a free parameter, and that the McMillan degree of the solution is more than that of the generalized plant [11]. A recent paper [65] pointed out that McMillan degree of the non-standard  $H_{\infty}$  controller can be reduced to an order lower than that of the standard controller, and that paper derived a reduced-order  $H_{\infty}$  controller by using Riccati inequality approach. This thesis aims at deriving the class of the reduced-order  $H_{\infty}$  controllers by way of Riccati equation approach.

#### 1.4.2 Integral-type low-order robust controller design

In the controller design, there are many cases where the controller is required to contain integrators. The integral-type robust controller design is indispensable for a synthesis of the robust servo system. However, in general, the framework of the  $H_{\infty}$  control doesn't produce the integral-type controller. The problem of designing the integral-type robust controller has been studied by many researchers and many approaches based on the standard  $H_{\infty}$  control have been proposed. On the other hand, approaches based on the non-standard  $H_{\infty}$  control have not been studied well even though the problem naturally reduces to some non-standard  $H_{\infty}$  control is unpopular is that the solution is complicated compared with the approach of the standard  $H_{\infty}$  control.

Let us consider an SISO system in Figure 1.3, where P(s) is a plant assumed to have neither poles nor zeros on the imaginary axis. This assumption is relaxed in a later chapter. Consider a simple reference tracking problem, where reference signal r is supposed to be the step-formed input. To let the output y track the reference signal r, we require the controller K(s) to contain an integrator. One of the effective ways to meet this requirement is to introduce the integrator into the weight  $W_s(s)$  and to let the transfer function  $T_{zr}(s) \in \mathcal{RH}_{\infty}$ . This is done by letting  $T_{zr}(s)$  satisfy the following inequality:

$$\|T_{zr}(s)\|_{\infty} < \gamma, \tag{1.7}$$



Figure 1.3: A closed loop system

where  $\gamma$  is a positive real number given beforehand. Thus we consider the  $H_{\infty}$  problem stated as: find internally stabilizing controller K(s) such that inequality (1.7) is satisfied. Since no states of the weight  $W_s(s)$  are observed by K(s), the generalized plant in Figure 1.3 doesn't satisfy the assumptions A1-2) nor A3-2), and the problem is thus a non-standard  $H_{\infty}$  problem. For the problems of this type, many approaches based on the conversion to the standard  $H_{\infty}$  problem have been proposed. The book of Zhou, Doyle and Glover [73] lists two types of approaches.

One is based on factorization of the weight  $W_s(s)$  into the parts of integrators and another. The factorization makes the problem the standard one. This approach derives an integraltype controller by a series of process: (1) factorizing the weight, (2) solving a problem of the standard  $H_{\infty}$  control, and (3) modifying a controller derived from the problem to an integral type. Thus, this approach requires one to follow several steps. The other is an approximative one, in which the weight  $W_s(s)$  is approximated in such a way that the integral part of  $W_s(s)$ is replaced as follows:

$$\frac{1}{s+\epsilon} \leftarrow \frac{1}{s}$$

where  $\epsilon$  is a small positive real number. By the replacement of the integrator in  $W_s(s)$ , the generalized plant in Figure 1.3 satisfies the assumptions A1-2) and A3-2), and the problem reduces to the standard  $H_{\infty}$  problem. This approach has convenience, on the other hand, it derives an incomplete integral-type controller. The other famous one is an approach that uses the generalized  $H_{\infty}$  control [39, 40, 49, 47], where it is required to derive a pseudo-stabilizing solution of ARE. Thus, there are many approaches to solve the problem, however,

the solutions are complex or incomplete. This is because the solutions are restricted by the standard problem.

On the other hand, we treat the problem of the integral-type robust controller design as the non-standard  $H_{\infty}$  problem for the generalized plant described in Figure 1.4, where the output from  $W_s(s)$  is measured by K(s). In the generalized plant, while the assumptions A1-



Figure 1.4: An alternative generalized plant

2) and A3-2) are satisfied, the assumption A2-2) is not satisfied because the dimension of the measurement output is greater than that of the external input. Thus the  $H_{\infty}$  problem for the system in Figure 1.4 is the non-standard problem where the matrix  $D_{21}$  is of full column rank, and the problem can be solved by using the solution of the non-standard problem we focus on. Although we can solve the problem of integral-type robust controller design, there is another problem; that is, the resultant integral-type controller  $K_I(s)$  becomes high-order than the system G(s) by the dimension of  $W_s(s)$ . For this problem, we show that our study on the designing of the reduced-order  $H_{\infty}$  controller contributes. Compared with previous works, while the problem description is essentially identical to the description in the works of Zhang, et al. [71] and Hozumi et al. [26], our study takes a different approach to solving the problem of the integral-type  $H_{\infty}$  controller design, and enables us to extend the result for TDF control system design to be stated below.

#### 1.4.3 Low-order TDF controller design

TDF control is one effective way to attain feedback performances and tracking performance simultaneously. Figure 1.5 illustrates the TDF control system where K(s) is a feedback controller and F(s) is a feedforward controller. A past study [66, 44] on the TDF control



Figure 1.5: TDF control system

has shown a property such that the feedback and tracking performances are determined independently by the feedback and feedforward controllers. Based on the property, the TDF controller is frequently designed in two steps: the feedback controller design and the feedforward controller design. Here, suppose that the output feedback controller is designed in the first step. Then the feedforward controller is designed such that it satisfies the specification:

$$\left\|W_{\rho}(s)\left(M(s) - T_{yr}(s)\right)\right\|_{\infty} < \gamma,$$

where M(s) is a model of the tracking signal,  $W_{\rho}(s)$  is a weight, and  $T_{yr}(s)$  is the closed loop transfer function from the signal r to y. This specification is illustrated in Figure 1.6. In



Figure 1.6: A feedforward controller design in the second step

this design, it should be noted that the order of F(s) becomes no less than twice the order of P(s). Hence, the order of the TDF controller becomes not less than 3 times as much as that

of the plant model. That is, the two-step design of the TDF controller results in a high-order controller. In the previous studies [17, 5, 6, 24], in order to design a reduced-order TDF controller, the dynamics of the feedback controller which is designed beforehand is used as the dynamics in the feedforward controller.

However, in these studies, it isn't noted that the above property is guaranteed in the only case where the controllers are independently designed. Specifically, the low-order TDF controller design inevitably causes any correlation in the structures of the feedback controller and the feedforward controller. Hence, it is important to analyze the trade-off between these performances of the TDF control system in which the controllers have any correlation. This thesis first analyzes the trade-off between these performances of the low-order TDF control system, in which the controllers share the same dynamics, and shows that the independence of the performances is not maintained in this case.



Figure 1.7: Simultaneous design of TDF controller

From the above reason, designing of the TDF controller is considered with a simultaneous approach rather than the two-step approach. Figure 1.7 illustrates a generalized plant for the simultaneous design of the TDF controller. In this formulation, the feedback controller and the feedforward controller are designed simultaneously. It should be noted that the resultant controllers share the common dynamics, and the degree of the controller is not increased unnecessarily. On the other hand, it is interesting to note that the system in Figure 1.7 is the non-standard generalized plant discussed in the previous sections. Hence, it is expected that the order of the controller can be further reduced.

#### 1.5 $H_{\infty}$ controller reduction

There are many model reduction methods that are applicable for the controller order reduction. However it is difficult to reduce the order of the controller with preserving the closed loop performances of the  $H_{\infty}$  controller. Let us consider the closed loop system illustrated in Figure 1.1, where the controller stabilizes the system and satisfies the constraint such that

$$\left\|\mathcal{F}_{l}\left(G(s), K(s)\right)\right\|_{\infty} < 1.$$

$$(1.8)$$

#### 1.5. $H_{\infty}$ CONTROLLER REDUCTION

Then we reduce the order of the controller with some model reduction method, if the order of the controller is too high to implement in practical use. However, neither the constraint of (1.8) nor the stability of closed loop system is necessarily satisfied even if the reduction error of the controller is small. This indicates that good approximation in the sense of open loop property doesn't mean good approximation in the sense of closed loop property. Hence, when we evaluate the properties of closed loop systems, the controller approximation should be considered. On the other hand, it is hard to evaluate the degradations in the closed loop properties when we approximate the controller.



Figure 1.8: Closed loop system

In this thesis, we consider the method for the  $H_{\infty}$  controller reduction by way of the reduction of the Youla parameter Q(s) that is designed such that the closed loop performances are satisfied. We focus on the  $H_{\infty}$  controller that is designed based on a numerical approach where the Youla parameter is optimized under some specification. The closed loop system is illustrated in Figure 1.8. It should be noted that the controller is represented as

$$K(s) = \mathcal{F}_l(M(s), Q(s)),$$

where the system M(s) is given a priori. It should also be noted that the closed loop transfer function  $T_{zw}(s)$  can be represented as

$$T_{zw}(s) = T_1 - T_2(s)Q(s)T_3(s),$$

where  $T_1(s), T_2(s)$  and  $T_3(s)$  are stable transfer functions which depend on factorization of the plant and control specifications. This representation is useful for evaluating the closed loop performances.

Let us define the optimal Youla parameter as

$$Q^*(s) \triangleq \arg \min_{Q(s) \in \mathcal{RH}_{\infty}} \|T_{zw}(s)\|_{\infty}.$$

If we approximate  $Q^*(s)$  with  $Q_r(s)$ , an inequality:

$$\min \|T_{zw}(s)\|_{\infty} = \|T_1 - T_2 Q^*(s) T_3(s)\|_{\infty}$$
  
$$\leq \|T_1 - T_2 Q_r(s) T_3(s)\|_{\infty} + \|T_2\|_{\infty} \|T_3\|_{\infty} \|Q^*(s) - Q_r(s)\|_{\infty}$$

holds. Letting closed loop transfer functions as

$$T_{zw}^*(s) \triangleq T_1 - T_2 Q^*(s) T_3(s)$$
  

$$\tilde{T}_{zw}(s) \triangleq T_1 - T_2 Q_r(s) T_3(s)$$
  

$$\gamma \triangleq ||T_2||_{\infty} ||T_3||_{\infty} ||Q^*(s) - Q_r(s)||_{\infty}$$

the inequality is written

$$||T_{zw}^*(s)||_{\infty} \le ||\tilde{T}_{zw}(s)||_{\infty} + \gamma.$$
(1.9)

As shown in Figure 1.9, the inequality (1.9) indicates that at every frequency, the graph of



Figure 1.9: An interpretation of inequality 1.9

 $T_{zw}^*(j\omega)$  lies in a disk of center  $\tilde{T}_{zw}(j\omega)$ , radius  $\gamma$ . From the inequality, it can be seen that good approximation of  $Q^*(s)$  with  $Q_r(s)$  suppresses the degradation in the closed loop performance. Furthermore, it should be noted that the approximation  $Q_r(s) \in \mathcal{RH}_{\infty}$  guarantees the closed loop stability; moreover, the closed loop poles depend directly on the poles of  $Q_r(s)$ . Hence, if we could obtain an approximation method that preserves the region in which poles of the system are located, we can propose an approximation method of the controller that satisfies a constraint of closed loop pole position.

These observations motivate to approximate the parameter Q(s) in order to approximate the  $H_{\infty}$  controller. Thus, we propose a method of  $H_{\infty}$  controller reduction based on the approximation of the Youla parameter.

#### **1.6** Contribution

The new results and the original points of this thesis are listed as follows.

1. The reduced-order  $H_{\infty}$  controllers are derived in the non-standard  $H_{\infty}$  problems. The class of the reduced-order controllers belongs to a generalized class of the minimal order observer based output feedback controllers. Derivation of the reduced-order controllers is based on an algebraic operation; that is, it is based on the selection of the free parameters in the general representation of the non-standard solutions.

- 2. As an application of the non-standard problems, a method for designing an integral-type robust controller is proposed. This method makes it possible to design the low-order integral-type robust controller easily and is also applicable to a low-order TDF controller design.
- 3. In the analysis of the low-order TDF control system, we point out a necessity to consider the trade-off between the feedback performances and the tracking performance. Specifically, in the case of the low-order TDF controller, in which controllers share common dynamics, it is clarified that the independent property among those performances is not maintained.
- 4. Based on the above analysis, it is pointed out that the low-order TDF controller design problem should be treated as a simultaneous optimization problem. In the synthesis of the low-order TDF control system, an approach based on the non-standard  $H_{\infty}$ problem is proposed. Specifically, the result of the reduced-order  $H_{\infty}$  controller design is applicable to this approach and a method for designing some kind of low-order TDF controller is proposed.
- 5. An  $H_{\infty}$  controller reduction method based on the reduction of the high-order Youla parameter is proposed. This method makes it easy to take closed loop specifications into account. That is, the reduced-order controller guarantees the closed loop properties such as internal stability and closed loop pole specification, and the degradation in the closed loop performance: the  $H_{\infty}$  norm of the closed loop transfer function can be evaluated.

#### 1.7 Organization

The organization of this thesis is as follows. Chapter 2 gives two-types of reduced-order controllers of the non-standard  $H_{\infty}$  problem. The results given in this chapter are fundamental to chapters of 3, 4 and 5. Chapter 3 considers a synthesis of the low-order robust servo control. It treats the problem as the non-standard  $H_{\infty}$  problem and derives the integral-type  $H_{\infty}$ controller. Chapters 4 and 5 consider the problem of the low-order Two-Degree-of-Freedom controller design. Chapter 4 analyzes the low-order TDF control system and shows that there is a trade-off between the feedback performance and the tracking performance. Based on the analysis, the chapter discusses the formulations of the low-order TDF controller. Chapter 5 proposes a method for designing the low-order TDF controller. It is based on the nonstandard  $H_{\infty}$  control. Chapter 6 considers the method for reducing the order of the  $H_{\infty}$ controller, which is a special solution of a numerical approach. Chapter 7 is the conclusion. The structure of the thesis is illustrated in Figure 1.10, where a symbol C indicates a chapter.



Figure 1.10: Organization of the thesis

### Chapter 2

# Reduced-order non-standard $H_{\infty}$ controller design

#### 2.1 Introduction

This chapter treats a fundamental issue to give a reduced-order  $H_{\infty}$  controller. In particular, problems that are focused on are the non-standard  $H_{\infty}$  control problems, where the dimensions of the control inputs are greater than those of the controlled outputs, and the dimensions of the measurement outputs are greater than those of the disturbance inputs. As a special case, those types of the non-standard  $H_{\infty}$  problems include the  $H_{\infty}$  problems of systems with partial state feedback [71] and redundant inputs [59]. It is known that the dynamical order of the observer-based feedback controller for a plant whose partial state variables are measured without noise can be reduced by the number of the partial state variables [30, 64, 65, 68, 70]. We thus hope that we can derive the reduced-order controller in the non-standard  $H_{\infty}$  problems.

Thus, this chapter aims to generalize the reduced-order controller design based on the non-standard  $H_{\infty}$  control problems. It starts from deriving the class of full-order  $H_{\infty}$  controllers [45, 51], in which the controllers are parametrized with free parameters. The remarkable difference between the classes of the  $H_{\infty}$  controllers in the non-standard problem and in the standard problem lies in the parametrization of these controllers. The  $H_{\infty}$  controllers of the non-standard problems are represented with larger number of free parameters than that of the standard problem. The difference in the structure of  $H_{\infty}$  problems reflects the difference in the parametrizations of these controllers. Hence, the free parameters in the  $H_{\infty}$  controllers.

In fact, the approach taken here is based on the selection of the free parameters in the full-order  $H_{\infty}$  controllers. The results given in this chapter are stated as follows. Two types of reduced-order  $H_{\infty}$  controllers are derived. One of the controllers is the minimal order observer type, where the order of the controller is less than that of the generalized plant by the number of the disturbance-free outputs. The other one is the dual version of the

aforementioned case, where the order of the controller is less than that of the generalized plant by the number of the redundant inputs. Results given in this chapter will be utilized in the later chapters.

#### 2.2 Formulation

Let us consider the generalized plant described in (1.1) or (1.2). This section considers the class of non-standard  $H_{\infty}$  control problems where the following assumptions are satisfied.

A1 1)  $(A, B_2)$  is stabilizable, and 2)  $(A, C_2)$  is detectable.

A2' One of the following cases is satisfied.

**Case 1**  $D_{12}$  and  $D_{21}$  are of full row rank.

**Case 2**  $D_{12}$  and  $D_{21}$  are of full column rank.

**Case 3**  $D_{12}$  is of full row rank and  $D_{21}$  is of full column rank.

**A3'** 
$$\forall \omega \in \mathbb{R}; \begin{bmatrix} A - j\omega I & B_2 \\ C_1 & D_{12} \end{bmatrix}$$
 and  $\begin{bmatrix} A - j\omega I & B_1 \\ C_2 & D_{21} \end{bmatrix}$  are of full rank.

In this thesis, we call the  $H_{\infty}$  problem, where the assumptions A1, A2'-Case i and A3' hold, the non-standard  $H_{\infty}$  problem of case i.

In the  $H_{\infty}$  problems of case 1 and case 3, we put the assumption such that the matrix  $\begin{bmatrix} B_2 \\ D_{12} \end{bmatrix}$  is of full column rank. In these cases,  $B_2$  and  $D_{12}$  are supposed to be of the form:

$$\begin{bmatrix} -\frac{B_2}{D_{12}} \end{bmatrix} = \begin{bmatrix} -\frac{B_{21}}{O} - \frac{B_{22}}{I_{p_1}} \end{bmatrix}, \quad B_{22} \in \mathbb{R}^{n \times p_1},$$
(2.1)

where  $B_{21}$  is of full column rank.

In the problems of case 2 and case 3, we put the assumption such that the matrix  $\begin{bmatrix} C_2 & D_{21} \end{bmatrix}$  is of full row rank. In these cases,  $C_2$  and  $D_{21}$  are supposed to be of the form:

$$\begin{bmatrix} C_2 & D_{21} \end{bmatrix} = \begin{bmatrix} C_{21} & O \\ C_{22} & I_{m_1} \end{bmatrix}, \quad C_{22} \in \mathbb{R}^{m_1 \times n},$$
(2.2)

where  $C_{21}$  is of full row rank.

These assumptions imposed on the system don't lose the generality in the system (1.1), because an equivalent transformation on the output variable:  $\bar{y} = My, |M| \neq 0$ , and an input transformation on the input variable:  $\bar{u} = N^{-1}u, |N| \neq 0$  in the system (1.1) yield the matrices in the equations (2.1) and (2.2). In this case, if a control low  $\bar{u} = K(s)\bar{y}$  is obtained, the real control law is reconstructed as u = NK(s)My.

#### 2.3 Preliminaries

#### 2.3.1 Pseudo inverse matrix and orthogonal complement matrix

For a singular and full-rank matrix, a pseudo inverse matrix and an orthogonal complement matrix are defined as follows.

1. Let D be of full column rank, the matrices  $D^{\dagger}$  and  $D^{\perp}$  are defined according to the following equalities:

$$\begin{bmatrix} D^{\dagger} \\ D^{\perp} \end{bmatrix} \begin{bmatrix} D & (D^{\perp})^T \end{bmatrix} = I,$$
$$DD^{\dagger} + (D^{\perp})^T D^{\perp} = I.$$

2. Let D be of full row rank, the matrices  $D^{\dagger}$  and  $D^{\perp}$  are defined according to the following equalities:

$$\begin{bmatrix} D\\ (D^{\perp})^T \end{bmatrix} \begin{bmatrix} D^{\dagger} & D^{\perp} \end{bmatrix} = I,$$
$$D^{\dagger}D + D^{\perp} (D^{\perp})^T = I.$$

We call the matrix  $D^{\dagger}$  pseudo inverse matrix of D, and call the matrix  $D^{\perp}$  orthogonal complement matrix.

#### 2.3.2 Invariant zeros

Let us consider the system

$$T(s) = \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix},$$

where the matrix D is nonsingular and is of full rank. Then the invariant zeros of T(s) are characterized as follows.

- **Lemma 2.3.1** 1. Let D be of full column rank, then the invariant zeros of T(s) are the unobservable modes of the pair  $(A BD^{\dagger}C, D^{\perp}C)$ .
  - 2. Let D be of full row rank, then the invariant zeros of T(s) are the uncontrollable modes of the pair  $(A BD^{\dagger}C, BD^{\perp})$ .

**Proof.** See section A.1.

#### 2.3.3 Canonical transformation of the generalized plant

Our subsequent analysis is greatly simplified if the generalized plant is transformed so that the invariant zeros are explicitly represented. Therefore, the generalized plant is transformed to a canonical form:

$$G(s) = \begin{bmatrix} T^{-1}AT & T^{-1}B_1 & T^{-1}B_2 \\ \hline C_1T & O & D_{12} \\ C_2T & D_{21} & O \end{bmatrix},$$

where the matrix T is supposed to be nonsingular. Then, the following lemma holds.

**Lemma 2.3.2** (1) Assume the generalized plant of case 1 or case 3, i.e.,  $D_{12}$  is of full row rank, let the stable and unstable zero modes of the subsystem  $G_{12}(s)$  be  $A_{-}$  and  $A_{+} \in \mathbb{R}^{r \times r}$ . Then the following transformation is possible

$$\begin{bmatrix} T^{-1}(A - B_2 D_{12}^{\dagger} C_1) T & T^{-1} B_2 D_{12}^{\perp} & T^{-1} B_2 D_{12}^{\dagger} \end{bmatrix}$$
$$= \begin{bmatrix} A_{11} & A_{12} & A_{21} & A_{22} & I_{m_2 - p_1} & B_{22uu} \\ A_{13} & A_{14} & A_{23} & A_{24} & O & B_{22ul} \\ O & O & A_- & O & O & B_{22lu} \\ O & O & O & A_+ & O & B_{22ll} \end{bmatrix},$$
(2.3)

where  $\lambda_i(A_{14}) < 0$ , both of the pairs  $(A_{11}, I_{m2-p1})$  and  $(A_+, B_{22ll})$  are controllable, and  $A_+$  has no  $j\omega$  eigenvalues.

(2) Assume the generalized plant of case 2 or case 3, i.e.,  $D_{21}$  is of full column rank, let the stable and unstable zero modes of the subsystem  $G_{21}(s)$  be  $\bar{A}_{-}$  and  $\bar{A}_{+} \in \mathbb{R}^{l \times l}$ . Then the following transformation is possible

$$\begin{bmatrix} \bar{T}^{-1}(A - B_1 D_{21}^{\dagger} C_2) \bar{T} \\ \bar{D}_{21}^{-} \bar{C}_2 \bar{T} \\ \bar{D}_{21}^{+} \bar{C}_2 \bar{T} \end{bmatrix} = \begin{bmatrix} A_{11} & A_{12} & O & O \\ \bar{A}_{13} & \bar{A}_{14} & O & O \\ \bar{A}_{31} & \bar{A}_{32} & \bar{A}_{-} & O \\ \bar{A}_{33} & \bar{A}_{34} & O & \bar{A}_{+} \\ \bar{I}_{p2-m_1} & \bar{O} & \bar{O} & \bar{O} \\ -\bar{C}_{22ll} & \bar{C}_{22rl} & \bar{C}_{22rr} \end{bmatrix},$$
(2.4)

where  $\lambda_i(\bar{A}_{14}) < 0$ , both of the pairs  $(\bar{A}_{11}, I_{p_2-m_1})$  and  $(\bar{A}_+, \bar{C}_{22rr})$  are observable, and  $\bar{A}_+$  has no j $\omega$  eigenvalues.

**Proof.** (1) Choose a basis V for the controllable subspace of  $(A - B_2 D_{12}^{\dagger} C_1, B_2 D_{12}^{\perp})$ . Furthermore, choose a basis U such that the transformation matrix

$$M \triangleq \begin{bmatrix} V & U \end{bmatrix}$$

is nonsingular. Then the following transformation is possible

$$M^{-1}(A - B_2 D_{12}^{\dagger} C_1) M = \begin{bmatrix} A_1 & A_2 \\ O & A_4 \end{bmatrix}$$
$$M^{-1} B_2 D_{12}^{\perp} = \begin{bmatrix} \hat{B}_{21u} \\ O \end{bmatrix}$$
$$M^{-1} B_2 D_{12}^{\dagger} = \begin{bmatrix} \hat{B}_{22u} \\ \hat{B}_{22l} \end{bmatrix},$$

where  $(A_1, \hat{B}_{21u})$  is controllable and  $(A_4, \hat{B}_{22l})$  is stabilizable.

By the previous assumption,  $\hat{B}_{21u}$  is of full column rank and we put  $\hat{B}_{21u}$  as

$$\hat{B}_{21u} = \left[ \begin{array}{c} I_{m_2 - p_1} \\ O \end{array} \right]$$

Let us define a nonsingular transformation matrix J:

$$J \triangleq \left[ \begin{array}{cc} J_1 & O \\ O & J_2 \end{array} \right] \in \mathbb{R}^{n \times n}$$

Then  $J_1$  and  $J_2$  are nonsingular and the following equation holds:

$$J^{-1}M^{-1}(A - B_2 D_{12}^{\dagger}C_1)MJ = \begin{bmatrix} J_1^{-1}A_1J_1 & J_1^{-1}A_2J_2 \\ O & J_2^{-1}A_4J_2 \end{bmatrix}$$

Putting  $J_1$  and  $A_1$  as

$$J_1 = \begin{bmatrix} I & X \\ O & I \end{bmatrix}, A_1 = \begin{bmatrix} A_{111} & A_{112} \\ A_{121} & A_{122} \end{bmatrix}$$

and substituting these into  $J_1^{-1}A_1J_1$ , the following equation is obtained

$$J_1^{-1}A_1J_1 = \begin{bmatrix} A_{111} - XA_{121} & A_{111}X - XA_{121}X + A_{112} - XA_{122} \\ A_{121} & A_{121}X + A_{122} \end{bmatrix},$$

where  $(A_{122}, A_{121})$  is controllable, since  $(A_1, \hat{B}_{21u})$  is controllable. Hence the arbitrary eigenvalues of  $A_{121}X + A_{122}$  can be specified. Choose matrix X such that all the eigenvalues of  $A_{121}X + A_{122}$  lie in the left half plane and define the stable matrix  $A_{14}$  as follows.

$$A_{121}X + A_{122} \triangleq A_{14}$$

From this definition  $J_1^{-1}A_1J_1$  can be represented as

$$J_1^{-1}A_1J_1 = \left[ \begin{array}{cc} A_{11} & A_{12} \\ A_{13} & A_{14} \end{array} \right].$$

On the other hand, choosing a matrix  $J_2$  that does permutation and diagonalization,  $J_2^{-1}A_4J_2$  becomes

$$J_2^{-1}A_4J_2 = \left[\begin{array}{cc} A_- & O\\ O & A_+ \end{array}\right],$$

where the matrix  $A_{-}$  is a stable zero-mode of  $G_{12}(s)$  and the matrix  $A_{+}$  is an unstable zero-mode of  $G_{12}(s)$ . Hence by putting  $J_{1}^{-1}A_{2}J_{2}$  as

$$J_1^{-1}A_2J_2 = \left[ \begin{array}{cc} A_{21} & A_{22} \\ A_{23} & A_{24} \end{array} \right],$$

the following equations are obtained

$$J^{-1}M^{-1}(A - B_2 D_{12}^{\dagger}C_1)MJ = \begin{bmatrix} A_{11} & A_{12} & A_{21} & A_{22} \\ A_{13} & A_{14} & A_{23} & A_{24} \\ O & O & A_- & O \\ O & O & O & A_+ \end{bmatrix}$$
$$J_1^{-1}\hat{B}_{21u} = \begin{bmatrix} I_{m_2-p_1} \\ O \end{bmatrix}.$$

Partitioning  $J^{-1}M^{-1}B_2D_{12}^{\dagger}$  as

$$J^{-1}M^{-1}B_2D_{12}^{\dagger} = \begin{bmatrix} J_1^{-1} & O \\ O & J_2^{-1} \end{bmatrix} \begin{bmatrix} \hat{B}_{22u} \\ \hat{B}_{22l} \end{bmatrix} = \begin{bmatrix} J_1^{-1}\hat{B}_{22u} \\ J_2^{-1}\hat{B}_{22l} \end{bmatrix} = \begin{bmatrix} \hat{B}_{22uu} \\ \hat{B}_{22ul} \\ \hat{B}_{22lu} \\ \hat{B}_{22lu} \\ \hat{B}_{22ll} \end{bmatrix},$$

and putting

$$MJ \triangleq T$$

follow the results in equation (2.4).

(2) This result follows by applying the dual argument of the proof (1).

#### 2.4 The non-standard $H_{\infty}$ problem of case 2

This section derives full-order and reduced-order controllers of the non-standard  $H_{\infty}$  problem of case 2. That is, we consider the  $H_{\infty}$  problem for a generalized plant:

$$G(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ \hline C_{21} & O & O \\ \hline C_{22} & I_{m_1} & O \end{bmatrix},$$

where the assumptions A1, A2'-case 2 and A3' are satisfied. We put variables as follows.

$$\begin{bmatrix} C_2 & D_{21} \end{bmatrix} = \begin{bmatrix} C_{21} & O \\ C_{22} & I_{m_1} \end{bmatrix}$$

#### **2.4.1** Characterization of zeros in $G_{21}(s)$

Let us consider the non-standard problem of case 2. Since  $D_{21}$  is of full column rank, it is assumed that the generalized plant has been put into a basis corresponding to the canonical form in equation (2.4). Hence, the following equations can be assumed without loss of generality

$$A - B_1 D_{21}^{\dagger} C_2 = \begin{bmatrix} -\frac{A_{11}}{A_{13}} & \frac{A_{12}}{A_{14}} & 0 & 0 \\ -\frac{A_{31}}{A_{31}} & A_{32} & A_- & 0 \\ A_{33} & A_{34} & 0 & A_+ \end{bmatrix}, A_{11} \in \mathbb{R}^{(p_2 - m_1) \times (p_2 - m_1)}$$

$$D_{21}^{\perp} C_2 = \begin{bmatrix} I_{p_2 - m_1} & 0 & 0 & 0 \\ C_{22lr} & C_{22r} & C_{22rr} \end{bmatrix} = \begin{bmatrix} C_{22ll} & C_{22r} \end{bmatrix}, C_{22ll} \in \mathbb{R}^{p_2 \times (p_2 - m_1)}$$

$$(2.5)$$

where  $\lambda_i(A_{14}) < 0$ , both of the pairs  $(A_{11}, I_{p_2-m_1})$  and  $(A_+, C_{22rr})$  are observable, and  $A_+ \in \mathbb{R}^{l \times l}$  has no  $j\omega$  eigenvalues. We call  $A_-$  the stable zero mode of  $G_{21}(s)$  and  $A_+$  the unstable zero mode of  $G_{21}(s)$ .

For notational ease, let us partition the following matrices.

$$B_{1} = \begin{bmatrix} -B_{1u} \\ B_{1l} \end{bmatrix}, B_{1u} \in \mathbb{R}^{(p_{2}-m_{1})\times m_{1}}$$

$$B_{2} = \begin{bmatrix} -B_{2u} \\ B_{2l} \end{bmatrix}, B_{2u} \in \mathbb{R}^{(p_{2}-m_{1})\times m_{2}}$$

$$C_{1} = \begin{bmatrix} C_{1ll} & C_{1lr} & C_{1rl} & C_{1rr} \end{bmatrix} \triangleq \begin{bmatrix} C_{1ll} & C_{1r} \end{bmatrix}, C_{1ll} \in \mathbb{R}^{p_{1}\times(p_{2}-m_{1})}$$
(2.6)

Under the above preparations, let us choose  $L_H$  such that the observable subspace of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is stabilized, and put  $L_H$  as follows

$$L_{H} \triangleq \begin{bmatrix} L_{H_{1}} \\ -A_{13} \\ -A_{31} \\ L_{H_{33}} \end{bmatrix} \in \mathbb{R}^{n \times (p_{2} - m_{1})},$$
(2.7)

where  $L_{H_1} \in \mathbb{R}^{(p_2-m_1)\times(p_2-m_1)}$  is chosen such that  $A_{11}+L_{F_1}$  is stable, and  $L_{H_{33}} \in \mathbb{R}^{l\times(p_2-m_1)}$  is an arbitrary matrix. Then we define  $A_{ZH}$  as follows:

$$A_{ZH} = A - B_1 D_{21}^{\dagger} C_2 + L_H D_{21}^{\perp} C_2 = \begin{bmatrix} A_{11} + L_{H_1} & A_{12} & O + O \\ O & A_{14} & O + O \\ O & A_{14} & O + O \\ O & A_{33} + L_{H_{33}} & A_{34} & O + A_{4} \end{bmatrix},$$

where the (1,1)-block of the matrix  $A_{ZH}$  is stable.

Next, let us choose  $E_H$  such that the unstable zero-mode of  $G_{21}(s)$  is stabilized without effecting the stable mode of  $A_{ZH}$ . This implies that  $E_H$  is chosen such that a matrix  $A_H \triangleq A_{ZH} + E_H D_{21}^{\dagger} C_2$  is stable under the following constraint:

$$UE_H = O, (2.8)$$

where U is a row-basis of the stable subspace in  $A_{ZH}$ , that is, U satisfies

$$UA_{ZH} = \Lambda U,$$

where  $\Lambda$  is a stable mode of  $A_{ZH}$ .
Hence, temporarily, we choose  $E_H$  as

$$E_{H} \triangleq \begin{bmatrix} O \\ O \\ O \\ E_{H22} \end{bmatrix}.$$
 (2.9)

Then  $A_H$  is represented as follows

$$A_{H} = A_{ZH} + E_{H} D_{21}^{\dagger} C_{2}$$

$$= \begin{bmatrix} A_{11} + L_{H_{1}} & A_{12} & O & O \\ O & A_{14} & O & O \\ O & A_{32} & A_{-} & O \\ A_{33} + L_{H_{33}} + E_{H22} C_{22ll} & A_{34} + E_{H22} C_{22lr} & E_{H22} C_{22rl} & A_{+} + E_{H22} C_{22rr} \end{bmatrix},$$

$$(2.10)$$

where  $E_{H22}$  can be chosen such that  $A_+ + E_{H22}C_{22rr}$  is stable, because the pair  $(A_+, C_{22rr})$  is observable. Then the matrix  $A_H$  is stable.

## 2.4.2 A necessary condition for the solvability

First, we introduce Full Control problem, which will be used to derive a necessary condition for the solvability in the non-standard  $H_{\infty}$  problem of case 2.

**Definition 2.4.1** Let us consider the generalized plant (1.1), where the assumption A1 is satisfied. Suppose that the matrices  $B_2$  and  $D_{12}$  are defined as follows

$$B_2 = \begin{bmatrix} I_n & O \end{bmatrix}, \quad D_{12} = \begin{bmatrix} O & I_{p_1} \end{bmatrix}.$$
(2.11)

Then the  $H_{\infty}$  problem for a generalized plant

$$G_{FC}(s) = \begin{bmatrix} A & B_1 & I_n & O \\ \hline C_1 & O & O & I_{p_1} \\ \hline C_2 & D_{21} & O & O \end{bmatrix},$$

is called FC (Full Control) problem.

The following lemma is useful for solving the non-standard  $H_{\infty}$  problems.

**Lemma 2.4.1** Suppose that the  $H_{\infty}$  problem with the assumption A1 is solvable. Then it is necessary that the FC problem is solvable.

**Proof.** If the output feedback controller of u = K(s)y solves the  $H_{\infty}$  problem, it is necessary that the controller of the form:

$$\left[\begin{array}{c} u_1\\ u_2 \end{array}\right] = \left[\begin{array}{c} B_2\\ D_{12} \end{array}\right] K(s)y$$

solves the FC problem.

From Lemma 2.4.1, it is necessary that the non-standard  $H_{\infty}$  problem for G(s) where  $B_2$  and  $D_{12}$  are supposed to hold (2.11) is solvable. Hence we consider the non-standard problem for the generalized plant  $G_{FC}(s)$  where the assumptions A1, A2'-case 2 and A3' are satisfied.

If we choose an observer gain H and a feedback gain F such that

$$\begin{split} H &\triangleq -B_1 D_{21}^{\dagger} + L_H D_{21}^{\perp} + E_H D_{21}^{\dagger} \\ F &\triangleq \begin{bmatrix} H C_2 \\ -C_1 \end{bmatrix}, \end{split}$$

then  $A + HC_2 = A + B_2F = A_H$  is stable. From Lemma A.2.1, the stabilizing controllers K(s) for  $G_{22}(s)$  are parametrized with a free parameter  $Q(s) \in \mathcal{RH}_{\infty}$ , and the closed loop transfer function  $G_{zw} = \mathcal{F}_l(G_{FC}(s), K(s))$  is represented as follows.

$$G_{zw}(s) = \mathcal{F}_l \left( M_2(s), -QD_{21} \right), \quad Q(s) \in \mathcal{RH}_{\infty}$$
$$M_2(s) = \begin{bmatrix} A_H & E_H & O \\ \hline C_1 & O & D_{12} \\ D_{21}^{\dagger}C_2 & I & O \end{bmatrix}$$

From the above discussion, it can be seen that the  $H_{\infty}$  problem for  $G_{FC}(s)$  is solvable, if and only if the  $H_{\infty}$  problem for  $M_2(s)$  is solvable. Hence we consider the  $H_{\infty}$  problem for  $M_2(s)$ .

**Lemma 2.4.2** Suppose that the matrices  $L_H$  and  $E_H$  are selected as in (2.7) and (2.9) such that  $A_H$  is stable. Then, if the non-standard  $H_{\infty}$  problem of case 2 is solvable, it is necessary that the ARE

$$YA_{ZH}^{T} + A_{ZH}Y + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}Y = O,$$
(2.12)

where

$$A_{ZH} \triangleq A - B_1 D_{21}^{\dagger} C_2 + L_H D_{21}^{\perp} C_2$$

has a stabilizing solution  $Y \ge O$ , which stabilizes a matrix

$$A_{Y} \triangleq A_{ZH} + Y \left\{ C_{1}^{T} C_{1} - \left( D_{21}^{\dagger} C_{2} \right)^{T} D_{21}^{\dagger} C_{2} \right\}.$$
(2.13)

**Proof.** Consider the  $H_{\infty}$  problem for the generalized plant  $M_2(s)$  where  $A_H$  is stable. Seeing the matrix  $D_{12}$  given in (2.11), it can be verified that  $M_2(s)$  holds the assumptions **A1**, **A2** and **A3**, hence the  $H_{\infty}$  problem for  $M_2(s)$  is the standard problem. The solvability condition for the standard problem reduces to the condition such that the ARE in (2.12) has a nonnegative definite stabilizing solution Y.

**Theorem 2.4.1** Suppose that the matrices  $L_H$  and  $E_H$  are selected as in (2.7) and (2.9) such that  $A_H$  is stable, and that there exists a stabilizing solution  $Y \ge O$  in the ARE (2.12). Then the solution Y can be represented as follows:

$$Y = \begin{bmatrix} O & O \\ O & Y_r \end{bmatrix} \in \mathbb{R}^{n \times n}, \tag{2.14}$$

where  $Y_r \in \mathbb{R}^{l \times l}$  is a nonnegative definite stabilizing solution of reduced-order ARE:

$$Y_r A_+^T + A_+ Y_r + Y_r \left( C_{1rr}^T C_{1rr} - C_{22rr}^T C_{22rr} \right) Y_r = 0.$$
(2.15)

**Proof.** (Necessity:) Suppose that there exists a stabilizing solution  $Y \ge O$  in the ARE (2.12). The ARE can be represented as follows.

$$YA_{ZH}^T + A_Y Y = 0 (2.16)$$

Let U be a row-basis of the stable subspace of the matrix  $A_{ZH}$ . Then the basis U satisfies the following equation:

$$UA_{ZH} = \Lambda U, \tag{2.17}$$

where

$$\Lambda = \begin{bmatrix} A_{11} + L_{H_1} & A_{12} & O \\ O & A_{14} & O \\ O & A_{32} & A_{-} \end{bmatrix}.$$
 (2.18)

After post-multiplication by  $U^T$ , the equation (2.16) becomes

$$YA_{ZH}^{T}U^{T} + A_{Y}YU^{T} = (YU^{T})\Lambda^{T} + A_{Y}(YU^{T}) = O.$$

Since both  $\Lambda$  and  $A_Y$  are stable matrices, one can apply Lemma A.3.1 to this equation and deduce that

$$YU^T = O. (2.19)$$

Since Y is a symmetric matrix, it must be of the form in (2.14). By substituting this Y into the ARE (2.12), the reduced-order ARE (2.15) can be deduced. Hence it is necessary that the ARE has a nonnegative definite stabilizing solution  $Y_r$ .

(Sufficiency:) Suppose that there exists a stabilizing solution  $Y_r \ge O$  in the reducedorder ARE (2.15). Then the matrix

$$A_{Y_r} \triangleq A_+ + Y_r \left( C_{1rr}^T C_{1rr} - C_{22rr}^T C_{22rr} \right)$$

is stable. If we select Y as in (2.14), it can be verified that

$$A_{Y} = A_{ZH} + Y \left\{ C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2} \right\}$$
$$= \begin{bmatrix} A_{11} + L_{H_{1}} & A_{12} & O & O \\ O & A_{14} & O & O \\ O & A_{32} & A_{-} & O \\ \tilde{A}_{33} & \tilde{A}_{34} & \tilde{A}_{43} & A_{Y_{r}} \end{bmatrix}, \qquad (2.20)$$

where

$$\tilde{A}_{33} = A_{33} + L_{H_{33}} + Y_r \left( C_{1rr}^T C_{1ll} - C_{22rr}^T C_{22ll} \right)$$
$$\tilde{A}_{34} = A_{34} + Y_r \left( C_{1rr}^T C_{1lr} - C_{22rr}^T C_{22lr} \right)$$
$$\tilde{A}_{43} = Y_r \left( C_{1rr}^T C_{1rl} - C_{22rr}^T C_{22rl} \right)$$

is a stable matrix, and that the matrix Y satisfies the ARE in (2.12). Hence the ARE has a nonnegative definite stabilizing solution Y.

#### **2.4.3** Lossless factorization of G(s)

Up to this part, the matrix  $E_H$  is fixed as in (2.9). Again, let us choose  $E_H$  as follows:

$$E_H = -Y \left( D_{21}^{\dagger} C_2 \right)^T.$$
(2.21)

Since UY = O,  $E_H$  satisfies (2.8). Then the following lemma is stated.

**Lemma 2.4.3** Suppose that the ARE in (2.12) has a nonnegative definite stabilizing solution Y. Then the following statements are equivalent.

1. A controller K(s) stabilizes a closed loop system (G(s), K(s)), and satisfies

$$\|\mathcal{F}_l(G(s), K(s))\|_{\infty} < 1.$$

2. A controller K(s) stabilizes a closed loop system  $(G_{tmp}(s), K(s))$ , and satisfies

$$\|\mathcal{F}_l(G_{tmp}(s), K(s))\|_{\infty} < 1,$$

where  $G_{tmp}(s)$  is represented as follows:

$$G_{tmp}(s) = \begin{bmatrix} \hat{A} & B_1 - E_H & \hat{B}_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix}$$
(2.22)

and  $\hat{A}$  and  $\hat{B}_2$  are

$$\hat{A} = A + YC_1^T C_1$$
$$\hat{B}_2 = B_2 + YC_1^T D_{12}.$$

**Proof.** We can factorize G(s) as follows:

$$G(s) = \Theta(s) \star G_{tmp}(s),$$

where  $\star$  is star-product which is defined in section A.4, and  $\Theta(s)$  is represented as follows

$$\Theta(s) = \begin{bmatrix} \Theta_{11}(s) & \Theta_{12}(s) \\ \Theta_{21}(s) & \Theta_{22}(s) \end{bmatrix} = \begin{bmatrix} A_{\infty} & E_H & -YC_1^T \\ \hline C_1 & O & I \\ D_{21}^{\dagger}C_2 & I & O \end{bmatrix} \in \mathcal{RH}_{\infty}$$



Figure 2.1: Lossless factorization

where

$$A_{\infty} = A + H_{\infty}C_{2}$$
  
$$H_{\infty} = -B_{1}D_{21}^{\dagger} + E_{H}D_{21}^{\dagger} + L_{H}D_{21}^{\perp}.$$
 (2.23)

Since the ARE in (2.12) has a nonnegative definite stabilizing solution Y, it can be verified that the following equations hold.

$$\left\{ \begin{array}{l} YA_{\infty}^{T}+A_{\infty}Y+B_{\Theta}B_{\Theta}^{T}=O\\ YC_{\Theta}^{T}+B_{\Theta}D_{\Theta}^{T}=O\\ D_{\Theta}D_{\Theta}^{T}=I \end{array} \right. \label{eq:alpha}$$

Where  $B_{\Theta}, C_{\Theta}$  and  $D_{\Theta}$  are defined as follows:

$$B_{\Theta} = \begin{bmatrix} E_H & -YC_1^T \end{bmatrix}, C_{\Theta} = \begin{bmatrix} C_1 \\ D_{21}^{\dagger}C_2 \end{bmatrix}, D_{\Theta} = \begin{bmatrix} O & I \\ I & O \end{bmatrix}.$$

From the above fact and Lemma A.5.1, we can verify that  $\Theta^T(s)$  is inner function. Furthermore, it can be verified that  $\Theta_{12}^{-1}(s) \in \mathcal{RH}_{\infty}$  and  $\Theta_{22}(\infty) = O$ , hence the system  $\Theta^T(s)$  is lossless, where lossless-ness of a system is listed in section A.6.

Denote the system  $G_{zw}$  as

$$G_{zw} = \mathcal{F}_l(\Theta, \mathcal{F}_l(G_{tmp}, K))$$

From the definition of the  $H_{\infty}$ -norm, it is verified that

$$\|G_{zw}\|_{\infty} < 1 \quad \Leftrightarrow \quad \left\|G_{zw}^{T}\right\|_{\infty} < 1.$$
(2.24)

On the other hand,  $\|G_{zw}^T\|_{\infty}$  satisfies

$$\begin{aligned} \left\| G_{zw}^{T} \right\|_{\infty} &= \left\| \mathcal{F}_{l} \left( \Theta, \mathcal{F}_{l} \left( G_{tmp}, K \right) \right)^{T} \right\|_{\infty} \\ &= \left\| \mathcal{F}_{l} \left( \Theta^{T}, \mathcal{F}_{l} \left( G_{tmp}, K \right)^{T} \right) \right\|_{\infty}. \end{aligned}$$

Since  $\Theta^T$  is lossless, from Lemma A.6.1 the following equivalence is verified.

$$\left\|G_{zw}^{T}\right\|_{\infty} < 1 \quad \Leftrightarrow \quad \left\|\mathcal{F}_{l}\left(G_{tmp}, K\right)^{T}\right\|_{\infty} < 1 \quad \Leftrightarrow \quad \left\|\mathcal{F}_{l}\left(G_{tmp}, K\right)\right\|_{\infty} < 1 \tag{2.25}$$

Hence, from (2.24) and (2.25) it is shown that the statements 1 and 2 are equivalent.

## 2.4.4 Parametrization of full-order $H_{\infty}$ controller

Under the necessary condition such that the ARE in (2.12) has a nonnegative definite stabilizing solution Y, the  $H_{\infty}$  problem for G(s) is reduced to the  $H_{\infty}$  problem for  $G_{tmp}(s)$ . Hence, in this section, we consider the  $H_{\infty}$  problem for  $G_{tmp}(s)$ . First, let us parametrize the stabilizing controllers for  $G_{tmp}(s)$ .

Suppose that  $(\hat{A}, \hat{B}_2)$  is stabilizable. Then we can choose a feedback gain F such that

$$A_F \triangleq \hat{A} + \hat{B}_2 F$$

is stable, and let us choose an observer gain  $H_{\infty}$  as in (2.23), then

$$A_H \triangleq \hat{A} + H_\infty C_2 = A_Y$$

is stable. From Lemma A.2.1, the class of stabilizing controllers is represented as follows.

$$K(s) = \mathcal{F}_l \left( \begin{bmatrix} A_Y + \hat{B}_2 F & -H_\infty & -\hat{B}_2 \\ F & O & -I \\ -C_2 & I & O \end{bmatrix}, Q(s) \right), Q(s) \in \mathcal{RH}_\infty$$
(2.26)

Then the closed loop transfer function is represented as follows

$$\mathcal{F}_l\left(G_{tmp}(s), K(s)\right) = \mathcal{F}_l\left(\tilde{M}_2(s), -Q(s)D_{21}\right),$$

where

$$\tilde{M}_{2}(s) = \begin{bmatrix} \hat{A} + \hat{B}_{2}F & B_{1} - E_{H} & \hat{B}_{2} \\ \hline C_{1} + D_{12}F & O & D_{12} \\ O & I & O \end{bmatrix}.$$
(2.27)

From (2.27), the solvability conditions of the  $H_{\infty}$  problems for  $G_{tmp}(s)$  and for  $\tilde{M}_2(s)$  are equivalent, hence we consider the  $H_{\infty}$  problem for  $\tilde{M}_2(s)$ .

**Theorem 2.4.2** The non-standard  $H_{\infty}$  problem for the generalized plant which satisfies assumptions A1, A2'-case 2 and A3' is solvable if and only if an ARE:

$$X\left(A - B_2 D_{12}^{\dagger} C_1\right) + \left(A - B_2 D_{12}^{\dagger} C_1\right)^T X + X\left\{B_1 B_1^T - B_2 D_{12}^{\dagger} \left(B_2 D_{12}^{\dagger}\right)^T\right\} X + \left(D_{12}^{\perp} C_1\right)^T D_{12}^{\perp} C_1 = O$$
(2.28)

and an ARE in (2.15) have stabilizing solutions  $X \ge O$  and  $Y_r \ge O$  which satisfy  $\rho(XY) < 1$ , where

$$Y = \left[ \begin{array}{cc} O & O \\ O & Y_r \end{array} \right].$$

Under such conditions, the class of the  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty}^{2} = \left\{ K_{\infty}^{2}(s) : N(s) \in \mathcal{BH}_{\infty}, W(s) \in \mathcal{RH}_{\infty} \right\},\$$

where N(s) and W(s) are free parameters, and  $K^2_{\infty}(s)$  is represented as follows

$$K_{\infty}^{2}(s) = \mathcal{F}_{l} \left( \begin{bmatrix} A_{Y} + \hat{B}_{2}F_{\infty} & H_{\infty} & -\hat{B}_{2}\Sigma \\ \hline -\frac{-F_{\infty}}{D_{21}^{\dagger}\hat{C}_{2}Z} & O & -\Sigma \\ D_{21}^{\dagger}\hat{C}_{2}Z & D_{21}^{\dagger} & O \\ D_{21}^{\perp}C_{2} & D_{21}^{\perp} & O \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix} \right)$$
(2.29)

and

$$A_{Y} = A + YC_{1}^{T}C_{1} + H_{\infty}C_{2}$$

$$\hat{B}_{2} = B_{2} + YC_{1}^{T}D_{12}$$

$$\hat{C}_{2} = D_{21}B_{1}^{T}X + C_{2}$$

$$F_{\infty} = \left\{-D_{12}^{\dagger}C_{1} - D_{12}^{\dagger}\left(B_{2}D_{12}^{\dagger}\right)^{T}X\right\}Z$$

$$H_{\infty} = -B_{1}D_{21}^{\dagger} - Y\left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger} + L_{H}D_{21}^{\perp}$$

$$Z = (I - YX)^{-1}$$

$$\Sigma = \left(D_{12}^{T}D_{12}\right)^{-\frac{1}{2}}.$$

**Proof.** The  $H_{\infty}$  problem for  $\tilde{M}_2(s)$  in (2.27) is DF (Disturbance Feedforward) problem. In appendix A.7 the solution for the DF problem is listed. By using the solution, we derive the solution for the non-standard  $H_{\infty}$  problem of case 2.

(Solvability condition:) The solvability condition for the DF problem is such that ARE:

$$S\Psi + \Psi^T S + S\Upsilon S + \left(D_{12}^{\perp}C_1\right)^T D_{12}^{\perp}C_1 = O_1$$

where

$$\Psi \triangleq \hat{A} - \hat{B}_2 D_{12}^{\dagger} C_1 
\Upsilon \triangleq (B_1 - E_H) (B_1 - E_H)^T - \hat{B}_2 D_{12}^{\dagger} \left( \hat{B}_2 D_{12}^{\dagger} \right)^T$$

has a nonnegative definite stabilizing solution S. By using a definition in appendix A.8, we represent the solvability condition such that

$$\begin{cases} H_S \in \text{Dom}(\text{Ric})\\ S = \text{Ric}(H_S) \ge O \end{cases},$$
(2.30)

where

$$H_{S} \triangleq \begin{bmatrix} \Psi & \Upsilon \\ -\left(D_{12}^{\perp}C_{1}\right)^{T}D_{12}^{\perp}C_{1} & -\Psi^{T} \end{bmatrix}.$$

Since  $H_S$  satisfies

$$H_X \left[ \begin{array}{cc} I & Y \\ O & I \end{array} \right] = \left[ \begin{array}{cc} I & Y \\ O & I \end{array} \right] H_S,$$

## 2.4. THE NON-STANDARD $H_{\infty}$ PROBLEM OF CASE 2

where  $Y \ge O$  is a stabilizing solution of the ARE in (2.12), and  $H_X$  is defined as follows:

$$H_{X} \triangleq \begin{bmatrix} A - B_{2}D_{12}^{\dagger}C_{1} & B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger} \left(B_{2}D_{12}^{\dagger}\right)^{T} \\ - \left(D_{12}^{\perp}C_{1}\right)^{T}D_{12}^{\perp}C_{1} & - \left(A - B_{2}D_{12}^{\dagger}C_{1}\right)^{T} \end{bmatrix}$$

the solvability condition (2.30) is equivalent to

$$\begin{cases}
H_X \in \text{Dom}(\text{Ric}) \\
X = \text{Ric}(H_X) \ge O \\
\rho(XY) < 1
\end{cases}$$
(2.31)

where  $Y \ge O$  is a stabilizing solution of the ARE in (2.12). The solvability condition (2.31) implies that the ARE (2.28) has a nonnegative definite stabilizing solution X, and which satisfies  $\rho(XY) < 1$ , where  $Y = \begin{bmatrix} O & O \\ O & Y_r \end{bmatrix}$ , and  $Y_r$  is a nonnegative definite stabilizing solution of ARE (2.15).

Conversely if the solvability condition (2.31) is satisfied,  $H_S \in \text{Dom}(\text{Ric})$ . Then it can be verified that the matrix

$$\Psi - \hat{B}_2 D_{12}^{\dagger} \left( \hat{B}_2 D_{12}^{\dagger} \right)^T S = \hat{A} + \hat{B}_2 \left\{ -D_{12}^{\dagger} C_1 - D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T S \right\}$$

is stable, and  $(\hat{A}, \hat{B}_2)$  is stabilizable.

(Derivation of the controller:) Suppose that the solvability condition (2.31) is satisfied. In the beginning of this subsection, the feedback gain F can be selected from arbitrary matrix which stabilizes  $A_F = \hat{A} + \hat{B}_2 F$ . Since the condition (2.31) satisfies, we select F such that

$$F = F_{\infty} \triangleq -D_{12}^{\dagger}C_1 - D_{12}^{\dagger} \left(\hat{B}_2 D_{12}^{\dagger}\right)^T S,$$

where

$$S = XZ$$
$$Z = (I - YX)^{-1}.$$

By using the formula in Lemma A.7, we can derive the controller

$$-Q(s)D_{21} = \mathcal{F}_l \left( \begin{bmatrix} \hat{A} + \hat{B}_2 F_{\infty} & -(B_1 - E_H) & -\hat{B}_2 \Sigma \\ O & O & \Sigma \\ (B_1 - E_H)^T S & I & O \end{bmatrix}, N(s) \right),$$
(2.32)

where  $N(s) \in \mathcal{BH}_{\infty}$ , and  $\Sigma = (D_{12}^T D_{12})^{-\frac{1}{2}}$ . The equation (2.32) can be represented as follows:

$$-Q(s)D_{21} = \Sigma N(s) \left(I - M_{22}(s)N(s)\right)^{-1} M_{21}(s), \qquad (2.33)$$

where

$$M_{21}(s) = I - \Phi(s) (B_1 - E_H)$$
  

$$M_{22}(s) = -\Phi(s)\hat{B}_2\Sigma$$
  

$$\Phi(s) = (B_1 - E_H)^T S \left(sI - \hat{A} - \hat{B}_2F_\infty\right)^{-1}$$

In the equation (2.33)  $D_{21}$  is of full column rank, hence the solution of Q(s) is represented as follows:

$$Q(s) = -\Sigma N(s) \left( I - M_{22}(s) N(s) \right)^{-1} M_{21}(s) D_{21}^{\dagger} + \hat{W}(s) D_{21}^{\perp}$$

where  $\hat{W}(s) \in \mathcal{RH}_{\infty}^{m_2 \times (p_2 - m_1)}$ . Since  $(I - M_{22}(s)N(s))^{-1} \in \mathcal{RH}_{\infty}$ , we can replace  $\hat{W}(s)$  such that

$$\hat{W}(s) = -\Sigma \left( I - N(s) M_{22}(s) \right)^{-1} \left( W(s) + N(s) \Phi(s) L_H \right), \quad W(s) \in \mathcal{RH}_{\infty}^{m_2 \times (p_2 - m_1)}.$$

Then the solution Q(s) is given as

$$Q(s) = -\Sigma \left( I - N(s) M_{22}(s) \right)^{-1} \left\{ N(s) \left( D_{21}^{\dagger} + \Phi(s) H_{\infty} \right) + W(s) D_{21}^{\perp} \right\}$$
  
$$= \mathcal{F}_{l} \left( \begin{bmatrix} \hat{A} + \hat{B}_{2} F_{\infty} & H_{\infty} & -\hat{B}_{2} \Sigma \\ \hline 0 & 0 & -\Sigma \\ 0 & -$$

where  $N(s) \in \mathcal{BH}_{\infty}, W(s) \in \mathcal{RH}_{\infty}^{m_2 \times (p_2 - m_1)}$ . By applying this Q(s) into (2.26), we can derive the  $H_{\infty}$  controller in (2.29).

### Remark 2.4.1

- The class of H<sub>∞</sub> controllers is represented with two free parameters, where one of them is a free parameter in BH<sub>∞</sub> and the other one is a free parameter in RH<sub>∞</sub>. On the other hand, in the standard H<sub>∞</sub> problem, the class of H<sub>∞</sub> controller is represented with only a free parameter in BH<sub>∞</sub>. One of differences between the standard problem and the non-standard problem appears in this point.
- 2. The dynamical order of the central solution equals the dynamical order of the generalized plant, where we call the  $H_{\infty}$  controller, whose free parameters are fixed to zero, a central solution.
- 3. By using the free parameter, we can improve any performance of the central solution.

## 2.4.5 Derivation of reduced-order $H_{\infty}$ controller

This subsection derives reduced-order non-standard  $H_{\infty}$  controller. First, let us represent the full-order  $H_{\infty}$  controller with DHMT as follows:

$$K_{\infty}^{2}(s) = \mathcal{DHM}\left(\begin{bmatrix} A_{Y} & \hat{B}_{2} & H_{\infty} \\ \hline -\Sigma^{-1}F_{\infty} & \Sigma^{-1} & O \\ \hline D_{21}^{\dagger}\hat{C}_{2}\hat{Z} & O & D_{21}^{\dagger} \\ D_{21}^{\perp}\hat{C}_{2} & O & D_{21}^{\dagger} \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix}\right)$$
$$= \begin{bmatrix} A_{Y} & \hat{B}_{2} \\ \hline C_{K}(s) & \Sigma^{-1} \end{bmatrix}^{-1} \begin{bmatrix} A_{Y} & H_{\infty} \\ \hline C_{K}(s) & N(s)D_{21}^{\dagger} + W(s)D_{21}^{\perp} \end{bmatrix}, \qquad (2.35)$$

where  $C_K(s)$  is defined as follows.

$$C_K(s) \triangleq -\Sigma^{-1} F_{\infty} + N(s) D_{21}^{\dagger} \hat{C}_2 Z + W(s) D_{21}^{\perp} C_2.$$
(2.36)

The matrix  $A_Y$  is written in the equation (2.20), where  $L_{H_{33}} \in \mathbb{R}^{l \times (p_2 - m_1)}$  is an arbitrary matrix. We can choose  $L_{H_{33}}$  such that  $\tilde{A}_{33} = O$  holds. Hence, by letting  $L_{H_{33}}$  as

$$L_{H_{33}} = -A_{33} - Y_r \left( C_{1rr}^T C_{1ll} - C_{22rr}^T C_{22ll} \right), \qquad (2.37)$$

the matrix  $A_Y$  is represented as

$$A_Y = \begin{bmatrix} A_{11} + L_{H_1} & A_{12} & O & O \\ O & A_{14} & O & O \\ O & A_{32} & A_- & O \\ O & \tilde{A}_{34} & \tilde{A}_{43} & A_{Y_r} \end{bmatrix}$$

Then the matrix  $A_Y$  satisfies

$$A_Y V_Y = V_Y (A_{11} + L_{H_1}),$$

where  $V_Y = \begin{bmatrix} I_{p_2-m_1} \\ O \end{bmatrix} \in \mathbb{R}^{n \times (p_2-m_1)}$  is of full column rank. Hence, if  $C_K(s)$  satisfies

$$C_K(s)V_Y = O, (2.38)$$

the pair  $(A_Y, C_K(s))$  is not observable, hence the dynamical order of the controller  $K^2_{\infty}(s)$  is reduced by the number of  $p_2 - m_1$ .

From equation (2.36), the matrix  $C_K(s)$  is represented with the free parameter  $W(s) \in \mathcal{RH}_{\infty}^{m_2 \times (p_2 - m_1)}$ , and is spanned by the row-basis of  $D_{21}^{\perp}C_2$ . Since  $D_{21}^{\perp}C_2V_Y = I_{p_2-m_1}$  holds, if we choose the free parameter W(s) as

$$W(s) = -\left(-\Sigma^{-1}F_{\infty} + N(s)D_{21}^{\dagger}\hat{C}_{2}Z\right)V_{Y},$$
(2.39)

the matrix  $V_Y$  satisfies (2.38). Thus we can derive a reduced-order  $H_{\infty}$  controller.

**Theorem 2.4.3** Under the same solvability condition as in Theorem 2.4.2, the class of reduced-order  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty}^{r_2} = \left\{ K_{\infty}^{r_2}(s) : N(s) \in \mathcal{BH}_{\infty} \right\},\$$

where N(s) is a free parameter, and  $K^{r_2}_{\infty}(s)$  is represented as follows

$$K_{\infty}^{r_{2}}(s) = \mathcal{DHM}\left(\begin{bmatrix} \tilde{A}_{Y} & \tilde{B}_{2} & \tilde{H}_{\infty} \\ \hline -\Sigma^{-1}F_{\infty_{2}} & \Sigma^{-1} & \Sigma^{-1}F_{\infty_{1}}D_{21}^{\perp} \\ C_{D_{2}} & O & D_{21}^{\dagger} - C_{D_{1}}D_{21}^{\perp} \end{bmatrix}, N(s)\right),$$
(2.40)

and

$$\begin{split} A_{Y} &= \begin{bmatrix} -\frac{A_{11} + L_{H_{1}}}{O} & A_{12} & O & O \\ 0 & \tilde{A}_{Y} \\ 0 \end{bmatrix} \in \mathbb{R}^{n \times n}, \quad \tilde{A}_{Y} \in \mathbb{R}^{(n - (p_{2} - m_{1})) \times (n - (p_{2} - m_{1}))} \\ \hat{B}_{2} &= \begin{bmatrix} -\frac{B_{2u}}{B_{2}} \end{bmatrix} \in \mathbb{R}^{n \times m_{2}}, \quad \tilde{B}_{2} \in \mathbb{R}^{(n - (p_{2} - m_{1})) \times m_{2}} \\ H_{\infty} &= \begin{bmatrix} -\frac{B_{1u}}{B_{2}} \frac{D_{11}^{\dagger} + L_{H_{1}}}{\tilde{H}_{\infty}} \frac{D_{21}^{\dagger}}{-} \end{bmatrix} \in \mathbb{R}^{n \times p_{2}}, \quad \tilde{H}_{\infty} \in \mathbb{R}^{(n - (p_{2} - m_{1})) \times p_{2}} \\ F_{\infty} &= \begin{bmatrix} F_{\infty_{1}} & F_{\infty_{2}} \end{bmatrix}, \quad F_{\infty_{1}} \in \mathbb{R}^{m_{2} \times (p_{2} - m_{1})}, F_{\infty_{2}} \in \mathbb{R}^{m_{2} \times (n - (p_{2} - m_{1}))} \\ D_{21}^{\dagger} \hat{C}_{2} Z &= \begin{bmatrix} C_{D_{1}} & C_{D_{2}} \end{bmatrix}, \quad C_{D_{1}} \in \mathbb{R}^{m_{1} \times (p_{2} - m_{1})}, C_{D_{2}} \in \mathbb{R}^{m_{1} \times (n - (p_{2} - m_{1}))}. \end{split}$$

**Proof.** From the previous argument, the reduced order controller is derived by substituting  $L_{H_{33}}$  in (2.37) and W(s) in (2.39) into  $K^2_{\infty}(s)$  in (2.35). Since  $\mathcal{K}^{r_2}_{\infty}$  is a subset of  $\mathcal{K}^2_{\infty}$ , the reduced order controller is an  $H_{\infty}$  controller.

**Remark 2.4.2** The dynamical order of the central solution in  $\mathcal{K}_{\infty}^{r_2}$  is  $n - (p_2 - m_1)$ , which is lower than that of the central solution in  $\mathcal{K}_{\infty}^2$  by the number of  $p_2 - m_1$ . The controller order reduction is analogous to the order reduction in observer-based controllers, where the order of an observer is reduced by the number of independent outputs that are not corrupted by disturbances.

Remark 2.4.3 The controller parameters are also represented as follows:

$$\begin{split} \tilde{A}_{Y} &= \tilde{A}_{r} + \tilde{Y}_{r} \left( C_{1r}^{T} C_{1r} - C_{22r}^{T} C_{22r} \right) \\ \tilde{B}_{2} &= B_{2l} + \tilde{Y}_{r} C_{1r}^{T} D_{12} \\ \tilde{H}_{\infty} &= -B_{1l} D_{21}^{\dagger} - \tilde{Y}_{r} C_{22r}^{T} D_{21}^{\dagger} + L_{H_{2}} D_{21}^{\perp} \\ F_{\infty_{1}} &= -D_{12}^{\dagger} \left\{ C_{1ll} Z_{11} + C_{1r} Z_{21} + \left( D_{12}^{\dagger} \right)^{T} \left( B_{2u}^{T} S_{11} + B_{2l}^{T} S_{21} \right) \right\} \\ F_{\infty_{2}} &= -D_{12}^{\dagger} \left\{ C_{1ll} Z_{12} + C_{1r} Z_{22} + \left( D_{12}^{\dagger} \right)^{T} \left( B_{2u}^{T} S_{12} + B_{2l}^{T} S_{22} \right) \right\} \\ B_{C_{1}} &= B_{1u}^{T} S_{11} + B_{1l}^{T} S_{21} + C_{22ll} Z_{11} + C_{22r} Z_{21} \\ B_{C_{2}} &= B_{1u}^{T} S_{12} + B_{1l}^{T} S_{22} + C_{22ll} Z_{12} + C_{22r} Z_{22}, \end{split}$$

where

$$\begin{split} \tilde{A}_{r} &= \begin{bmatrix} A_{14} & O & O \\ A_{32} & A_{-} & O \\ A_{34} & O & A_{+} \end{bmatrix}, \quad \tilde{Y}_{r} = \begin{bmatrix} O & O & O \\ O & O & O \\ O & O & Y_{r} \end{bmatrix} \in \mathbb{R}^{(n - (p_{2} - m_{1})) \times (n - (p_{2} - m_{1}))} \\ S &= XZ = \begin{bmatrix} -S_{11} & S_{12} \\ -\overline{S_{21}} & S_{22} \end{bmatrix}, \quad Z = \begin{bmatrix} -Z_{11} & Z_{12} \\ -\overline{Z_{21}} & Z_{22} \end{bmatrix}, \quad L_{H} = \begin{bmatrix} -L_{H_{1}} \\ -\overline{L_{H_{2}}} \end{bmatrix}, \end{split}$$

and  $X \ge O$  is a stabilizing solution of ARE in (2.52) and  $Y_r \ge O$  is a stabilizing solution of ARE in (2.15).

## 2.4.6 The controller structure

This subsection reviews the full-order controller and the reduced-order controller from the viewpoint of the observer based output controller. The central solution in the class of full-order  $H_{\infty}$  controller in (2.29) is represented as

$$\begin{cases} \dot{\hat{x}} = A_Y \hat{x} + \hat{B}_2 u - H_\infty y\\ u = F_\infty \hat{x} \end{cases},$$
(2.41)

where  $\hat{x}$  is a state variable of the controller. If we assume an orthogonal condition:  $C_1^T D_{12} = O$ , these equations can be written as

$$\begin{cases} \dot{\hat{x}} = A\hat{x} + B_2 u + H_\infty \left( C_2 \hat{x} - y \right) + Y C_1^T C_1 \hat{x} \\ u = F_\infty \hat{x} \end{cases}$$
(2.42)

From these equations, it can be seen that the controller has the structure of an observer-based controller. In order to interpret the structure of the controller, let us consider the system:

$$\begin{cases} \dot{x} = Ax + B_1 w + B_2 u + Y \left\{ C_1^T C_1 x + \left( D_{21}^{\dagger} C_2 \right)^T w \right\} \\ y = C_2 x + D_{21} w \end{cases}$$
(2.43)

By letting the error between the states of the systems in (2.42) and (2.43) as

$$e \triangleq x - \hat{x}$$

the dynamical equation of the error can be given as follows.

$$\dot{e} = A_Y e \tag{2.44}$$

Since all of the eigenvalues in the matrix

$$A_Y = A + YC_1^T C_1 + H_\infty C_2$$

have negative real parts, as the time intends to  $\infty$ , from any initial values of e(0) the error e(t) converges to zero. Thus, the controller in (2.41) consists of an observer for the system in (2.43) and a state feedback controller. Figure 2.2 illustrates the structure of the closed loop system that is constructed with the central controller.



Figure 2.2: The structure of the closed loop system

By representing the class of the  $H_{\infty}$  controllers with the equation (2.35), the structure of the observer can also be seen explicitly in the  $H_{\infty}$  controller which is represented with free parameters. Figure 2.3 illustrates the  $H_{\infty}$  controller with the observer-based representation where the dynamical equation of the observer is expressed explicitly. The size of the matrix  $A_Y$  equals the dynamical order of a full-order observer. Since  $C_K(s)$  is a function of the free parameters, the selection in the free parameters reduces to yield the unobservable subspace in the pair  $(A_Y, C_K(s))$ , and the order of the controller can be reduced by the number of the degrees of the subspace. Hence, this type of the reduced-order  $H_{\infty}$  controller can be interpreted as a minimal-order-observer-type  $H_{\infty}$  controller.

## 2.5 The non-standard $H_{\infty}$ problem of case 1

This section derives full-order and reduced-order controllers of the non-standard  $H_{\infty}$  problem in case 1. Since the problem of case 1 is the dual problem of case 2, discussions of this section are dual of the previous section, hence details of the procedure in the derivation of the fullorder controller are omitted. In this case, we treat a generalized plant:

$$G(s) = \begin{bmatrix} A & B_1 & B_{21} & B_{22} \\ \hline C_1 & O & O & I_{p_1} \\ \hline C_2 & D_{21} & O & O \end{bmatrix},$$



Figure 2.3: The structure of  $K^2_{\infty}(s)$ 

where the assumptions A1, A2'-case1 and A3' are satisfied. We put variables as follows.

$$\begin{bmatrix} -\frac{B_2}{\overline{D}_{12}} \end{bmatrix} = \begin{bmatrix} -\frac{B_{21}}{\overline{O}} - \frac{B_{22}}{\overline{I}_{p_1}} \end{bmatrix}$$

# **2.5.1** Characterization of zeros in $G_{12}(s)$

Let us consider the non-standard problem of case 1. Since  $D_{12}$  is of full row rank, it is assumed that the generalized plant has been put into a basis corresponding to the canonical form in equation (2.3). Hence the following equations can be assumed without loss of generality

$$A - B_{2}D_{12}^{\dagger}C_{1} = \begin{bmatrix} -\frac{A_{11}}{A_{13}} & \frac{A_{12}}{A_{14}} & \frac{A_{21}}{A_{23}} & \frac{A_{22}}{A_{24}} \\ O & O & A_{-} & O \\ O & O & O & A_{+} \end{bmatrix}, A_{11} \in \mathbb{R}^{(m_{2}-p_{1})\times(m_{2}-p_{1})}$$

$$B_{2}D_{12}^{\dagger} = \begin{bmatrix} -\frac{I_{m_{2}-p_{1}}}{O} \\ O \\ O \end{bmatrix}, A_{11} \in \mathbb{R}^{(m_{2}-p_{1})\times(m_{2}-p_{1})}$$

$$B_{2}D_{12}^{\dagger} = \begin{bmatrix} -\frac{B_{22uu}}{B_{22ul}} \\ B_{22lu} \\ B_{22lu} \\ B_{22ll} \end{bmatrix} \triangleq \begin{bmatrix} -\frac{B_{22uu}}{B_{22lu}} \\ -\frac{B_{22uu}}{B_{22l}} \end{bmatrix}, B_{22uu} \in \mathbb{R}^{(m_{2}-p_{1})\times m_{2}}$$

$$(2.45)$$

where  $\lambda_i(A_{14}) < 0$ , both of the pairs  $(A_{11}, I_{m_2-p_1})$  and  $(A_+, B_{22ll})$  are controllable, and  $A_+$  has no  $j\omega$  eigenvalues. We call  $A_-$  the stable zero mode of  $G_{12}(s)$  and  $A_+$  unstable zero mode of  $G_{12}(s)$ .

For notational ease, let us partition the following matrices.

$$B_{1} = \begin{bmatrix} B_{1uu} \\ \overline{B}_{1ul} \\ B_{1lu} \\ B_{1ll} \end{bmatrix} \triangleq \begin{bmatrix} B_{1uu} \\ \overline{B}_{1l} \end{bmatrix}, B_{1uu} \in \mathbb{R}^{(m_{2}-p_{1})\times m_{1}}$$
$$C_{1} = \begin{bmatrix} C_{1l} & C_{1r} \end{bmatrix} \in \mathbb{R}^{p_{1}\times n}, C_{1l} \in \mathbb{R}^{p_{1}\times (m_{2}-p_{1})}$$
$$C_{2} = \begin{bmatrix} C_{2l} & C_{2r} \end{bmatrix} \in \mathbb{R}^{p_{2}\times n}, C_{2l} \in \mathbb{R}^{p_{2}\times (m_{2}-p_{1})}.$$

Under the above preparations, let us choose  $L_F$  such that the controllable subspace of the pair  $(A - B_2 D_{12}^{\dagger} C_1, B_2 D_{12}^{\perp})$  is stabilized, and put  $L_F$  as follows

$$L_F \triangleq \begin{bmatrix} L_{F_1} & -A_{12} & -A_{21} & L_{F_{22}} \end{bmatrix} \in \mathbb{R}^{(m_2 - p_1) \times n},$$
 (2.46)

where  $L_{F_1} \in \mathbb{R}^{(m_2-p_1)\times(m_2-p_1)}$  is chosen such that  $A_{11} + L_{F_1}$  is stable, and  $L_{F_{22}} \in \mathbb{R}^{(m_2-p_1)\times r}$  is an arbitrary matrix. Then  $A_{ZF}$  is defined as follows:

$$A_{ZF} \triangleq A - B_2 D_{12}^{\dagger} C_1 + B_2 D_{12}^{\perp} L_F = \begin{bmatrix} A_{11} + L_{F_1} & O & O & A_{22} + L_{F_{22}} \\ A_{13} & A_{14} & A_{23} & A_{24} \\ O & O & A_- & O \\ \hline O & O & O & - A_+ & O \\ \hline O & O & O & - A_+ & - \end{bmatrix}$$

where the (1,1)-block of  $A_{ZF}$  is a stable.

Next, let us choose  $E_F$  such that the unstable zero-mode of  $G_{12}(s)$  is stabilized without effecting the stable mode of  $A_{ZF}$ . This implies that  $E_F$  is chosen such that the following constraint:

$$E_F V = O, (2.47)$$

,

where V is a basis of the stable subspace in  $A_{ZF}$ , that is, V satisfies

$$A_{ZF}V = V\Lambda,$$

where  $\Lambda$  is a stable mode of  $A_{ZF}$ .

Hence, temporarily, we choose  $E_F$  as

$$E_F \triangleq \begin{bmatrix} O & O & O & E_{F_{22}} \end{bmatrix}. \tag{2.48}$$

Then  $A_F$  is represented as follows

$$A_{F} = A_{ZF} + B_{2}D_{12}^{\dagger}E_{F}$$

$$= \begin{bmatrix} A_{11} + L_{F_{1}} & O & O & A_{22} + L_{F_{22}} + B_{22uu}E_{F22} \\ A_{13} & A_{14} & A_{23} & A_{24} + B_{22ul}E_{F22} \\ O & O & A_{-} & B_{22lu}E_{F22} \\ O & O & O & A_{+} + B_{22ll}E_{F22} \end{bmatrix}, \quad (2.49)$$

where  $E_{F22}$  can be chosen such that  $A_+ + B_{22ll}E_{F22}$  is stable, because the pair  $(A_+, B_{22ll})$  is controllable. Then the matrix  $A_F$  is stable.

38

#### 2.5.2 A necessary condition for the solvability

First, we introduce Full Information problem, which will be used to derive a necessary condition for the solvability in the non-standard  $H_{\infty}$  problem of case 1.

**Definition 2.5.1** Let us consider the generalized plant (1.1), where the assumption A1 is satisfied. Suppose that the matrices  $C_2$  and  $D_{21}$  are defined as follows

$$C_2 = \begin{bmatrix} I_n \\ O \end{bmatrix}, \quad D_{21} = \begin{bmatrix} O \\ I_{m_1} \end{bmatrix}.$$
(2.50)

Then the  $H_{\infty}$  problem for a generalized plant

$$G_{FI}(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ \hline I_n & O & O \\ O & I_{m_1} & O \end{bmatrix},$$

is called FI (Full Information) problem.

The following lemma is useful for solving the non-standard  $H_{\infty}$  problems.

**Lemma 2.5.1** Suppose that the  $H_{\infty}$  problem with the assumption A1 is solvable. Then it is necessary that the FI problem is solvable.

#### **Proof.** Omitted.

From Lemma 2.5.1, it is necessary that the non-standard  $H_{\infty}$  problem where  $C_2$  and  $D_{21}$  are supposed to hold (2.50) is solvable. Hence we consider the non-standard problem for the generalized plant  $G_{FI}(s)$  where the assumptions A1, A2'-case 1 and A3' are satisfied.

If we choose a feedback gain F and an observer gain H such that

$$F \triangleq -D_{12}^{\dagger}C_1 + D_{12}^{\perp}L_F + D_{12}^{\dagger}E_F \qquad (2.51)$$
$$H \triangleq \begin{bmatrix} B_2F & -B_1 \end{bmatrix},$$

then  $A + B_2F = A + HC_2 = A_F$  is stable. From Lemma A.2.1, the stabilizing controllers K(s) for  $G_{22}(s)$  are parametrized with a free parameter  $Q(s) \in \mathcal{RH}_{\infty}$ , and the closed loop transfer function  $G_{zw} = \mathcal{F}_l(G_{FI}(s), K(s))$  is represented as follows.

$$G_{zw}(s) = \mathcal{F}_l \left( M_1(s), -D_{12}Q \right), \quad Q(s) \in \mathcal{RH}_{\infty}$$
$$M_1(s) = \begin{bmatrix} A_F & B_1 & B_2D_{12}^{\dagger} \\ \hline E_F & O & I \\ O & D_{21} & O \end{bmatrix}$$

From the above discussion, the  $H_{\infty}$  problem for  $G_{FI}(s)$  is solvable, if and only if the  $H_{\infty}$  problem for  $M_1(s)$  is solvable. Hence we consider the  $H_{\infty}$  problem for  $M_1(s)$ .

**Lemma 2.5.2** Suppose that the matrices  $L_F$  and  $E_F$  are selected as in (2.46) and (2.48) such that  $A_F$  is stable. Then, if the non-standard  $H_{\infty}$  problem of case 1 is solvable, it is necessary that the ARE

$$XA_{ZF} + A_{ZF}^{T}X + X\left\{B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger}\left(B_{2}D_{12}^{\dagger}\right)^{T}\right\}X = O,$$
(2.52)

where

$$A_{ZF} \triangleq A - B_2 D_{12}^{\dagger} C_1 + B_2 D_{12}^{\perp} L_F$$

has a stabilizing solution  $X \ge O$ , which stabilizes a matrix

$$A_X \triangleq A_{ZF} + \left\{ B_1 B_1^T - B_2 D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T \right\} X.$$

$$(2.53)$$

Proof. Omitted.

**Theorem 2.5.1** Suppose that the matrices  $L_F$  and  $E_F$  are selected as in (2.46) and (2.48) such that  $A_F$  is stable, and that there exists a stabilizing solution  $X \ge O$  in the ARE (2.52). Then the solution X can be represented as follows:

$$X = \begin{bmatrix} O & O \\ O & X_r \end{bmatrix} \in \mathbb{R}^{n \times n},$$
(2.54)

where  $X_r \in \mathbb{R}^{r \times r}$  is a nonnegative definite stabilizing solution of reduced-order ARE:

$$X_r A_+ + A_+^T X_r + X_r \left( B_{1ll} B_{1ll}^T - B_{22ll} B_{22ll}^T \right) X_r = O.$$
(2.55)

## **Proof.** Omitted.

By using X in (2.54), the matrix  $A_X$  is represented as follows:

$$A_{X} = A_{ZF} + \left\{ B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger} \left( B_{2}D_{12}^{\dagger} \right)^{T} \right\} X$$
  
$$= \begin{bmatrix} A_{11} + L_{F_{1}} & O & O & \tilde{A}_{22} \\ A_{13} & A_{14} & A_{23} & \tilde{A}_{24} \\ O & O & A_{-} & \tilde{A}_{42} \\ O & O & O & A_{X_{r}} \end{bmatrix}, \qquad (2.56)$$

where

$$\tilde{A}_{22} = A_{22} + L_{F_{22}} + (B_{1uu}B_{1ll}^T - B_{22uu}B_{22ll}^T) X_r$$

$$\tilde{A}_{24} = A_{24} + (B_{1ul}B_{1ll}^T - B_{22ul}B_{22ll}^T) X_r$$

$$\tilde{A}_{42} = (B_{1lu}B_{1ll}^T - B_{22lu}B_{22ll}^T)$$

$$A_{Xr} = A_+ + (B_{1ll}B_{1ll}^T - B_{22ll}B_{22ll}^T).$$

From Theorem 2.5.1, since the matrix  $A_{X_r}$  is stable, the matrix  $A_X$  is also stable.

## 2.5.3 Parametrization of full-order $H_{\infty}$ controller

**Theorem 2.5.2** The non-standard  $H_{\infty}$  problem for the generalized plant which satisfies assumptions A1, A2'-case 1 and A3' is solvable if and only if an ARE in (2.55) has a stabilizing solution  $X_+ \geq O$ , and an ARE:

$$Y\left(A - B_{1}D_{21}^{\dagger}C_{2}\right)^{T} + \left(A - B_{1}D_{21}^{\dagger}C_{2}\right)Y + Y\left\{C_{1}^{T}C_{1} - \left(C_{2}D_{21}^{\dagger}\right)^{T}C_{2}D_{21}^{\dagger}\right\}Y + B_{1}D_{21}^{\perp}\left(B_{1}D_{21}^{\perp}\right)^{T} = O$$
(2.57)

has a stabilizing solution  $Y \ge O$  which satisfy  $\rho(XY) < 1$ , where

$$X = \left[ \begin{array}{cc} O & O \\ O & X_r \end{array} \right].$$

Under such conditions, the class of the  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}^{1}_{\infty} = \left\{ K^{1}_{\infty}(s) : N(s) \in \mathcal{BH}_{\infty}, W(s) \in \mathcal{RH}_{\infty} \right\},\$$

where N(s) and W(s) are free parameters, and  $K^1_{\infty}(s)$  is represented as follows

$$K_{\infty}^{1}(s) = \mathcal{F}_{l}\left(\begin{bmatrix} A_{X} + H_{\infty}\hat{C}_{2} & -H_{\infty} & Z\hat{B}_{2}D_{12}^{\dagger} & B_{2}D_{12}^{\dagger} \\ F_{\infty} & O & D_{12}^{\dagger} & D_{12}^{\dagger} \\ -\Sigma\hat{C}_{2} & \Sigma & O & O \\ -\Sigma\hat{C}_{2} & \Sigma & O & O \\ \end{bmatrix}, \begin{bmatrix} N(s) \\ W(s) \end{bmatrix}\right)$$
(2.58)

and

$$A_{X} = A + B_{1}B_{1}^{T}X + B_{2}F_{\infty}$$

$$\hat{B}_{2} = B_{2} + YC_{1}^{T}D_{12}$$

$$\hat{C}_{2} = D_{21}B_{1}^{T}X + C_{2}$$

$$F_{\infty} = -D_{12}^{\dagger}C_{1} + D_{12}^{\perp}L_{F} - D_{12}^{\dagger} \left(B_{2}D_{12}^{\dagger}\right)^{T}X$$

$$H_{\infty} = Z \left\{-B_{1}D_{21}^{\dagger} - Y \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}\right\}$$

$$Z = (I - YX)^{-1}$$

$$\Sigma = \left(D_{21}D_{21}^{T}\right)^{-\frac{1}{2}}.$$

Proof. Omitted.

#### Remark 2.5.1

- 1. By comparing the controllers of case 1 and case 2, it can be verified that each controller has dual structure of another controller.
- 2. The dynamical order of the central solution equals the dynamical order of the generalized plant, where we call the  $H_{\infty}$  controller, whose free parameters are fixed to zero, a central solution.
- 3. By using the free parameter, we can improve any performance of the central solution.

## 2.5.4 Derivation of reduced-order $H_{\infty}$ controller

By using the free parameter in  $\mathcal{K}^1_{\infty}$ , we derive the reduced-order  $H_{\infty}$  controller. First, let us represent the full-order  $H_{\infty}$  controller with a homogeneous transformation as follows:

$$K_{\infty}^{1}(s) = \mathcal{H}\mathcal{M}\left(\left[\begin{array}{c|c} A_{X} & Z\hat{B}_{2}D_{12}^{\dagger} & B_{2}D_{12}^{\bot} & -H_{\infty}\Sigma^{-1} \\ \hline F_{\infty} & D_{12}^{\dagger} & D_{12}^{\dagger} & -D_{12}^{\bot} \\ \hline C_{2} & -D_{12}^{\dagger} & D_{12}^{\bot} & -D_{\Sigma^{-1}}^{\bot} \\ \hline O & -D_{\Sigma^{-1}}^{\bot} & -D_{\Sigma^{-1}}^{\bot} \\ \hline \end{array}\right], \begin{bmatrix} N(s) \\ W(s) \end{bmatrix}\right) \\ = \left[\begin{array}{c|c} A_{X} & B_{K}(s) \\ \hline F_{\infty} & D_{12}^{\dagger}N(s) + D_{12}^{\bot}W(s) \end{array}\right] \left[\begin{array}{c|c} A_{X} & B_{K}(s) \\ \hline C_{2} & \Sigma^{-1} \end{array}\right]^{-1}, \quad (2.59)$$

where  $B_K(s)$  is defined as

$$B_K(s) \triangleq Z\hat{B}_2 D_{12}^{\dagger} N(s) + B_2 D_{12}^{\perp} W(s) - H_{\infty} \Sigma^{-1}.$$
 (2.60)

In the equation (2.56), since the matrix  $L_{F_{22}} \in \mathbb{R}^{(m_2-p_1)\times r}$  is arbitrary, we can choose  $L_{F_{22}}$  such that  $\tilde{A}_{22} = O$  holds. Hence let us choose  $L_{F_{22}}$ 

$$L_{F_{22}} = -A_{22} - \left(B_{1uu}B_{1ll}^T - B_{22uu}B_{22ll}^T\right)X_r$$
(2.61)

then the matrix  $A_X$  is represented as

$$A_X = \begin{bmatrix} A_{11} + L_{F_1} & O & O & O \\ A_{13} & A_{14} & A_{23} & \tilde{A}_{24} \\ O & O & A_- & \tilde{A}_{42} \\ O & O & O & A_{X_r} \end{bmatrix}$$

The matrix  $A_X$  satisfies

$$U_X A_X = (A_{11} + L_{F_1}) U_X,$$

where  $U_X = \begin{bmatrix} I_{m_2-p_1} & O \end{bmatrix} \in \mathbb{R}^{(m_2-p_1)\times n}$ . Hence, if  $B_K(s)$  satisfies

$$U_X B_K(s) = O, (2.62)$$

the pair  $(A_X, B_K(s))$  has an uncontrollable subspace, hence the order of the controller in  $\mathcal{K}^1_{\infty}$  is reduced by the number of  $m_2 - p_1$ .

From equation (2.60), the matrix  $B_K(s)$  is represented with the free parameter  $W(s) \in \mathcal{RH}_{\infty}$ , and is spanned by the basis of  $B_2 D_{12}^{\perp}$ . Since  $U_X B_2 D_{12}^{\perp} = I_{m_2-p_1}$  holds, if we choose the free parameter W(s) as

$$W(s) = -U_X \left( Z \hat{B}_2 D_{12}^{\dagger} N(s) - H_{\infty} \Sigma^{-1} \right), \qquad (2.63)$$

the matrix  $U_X$  satisfies (2.62). Thus we can derive a reduced-order  $H_{\infty}$  controller.

**Theorem 2.5.3** Under the same solvability condition as in Theorem 2.5.2, the class of reduced-order  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty}^{r_1} = \left\{ K_{\infty}^{r_1}(s) : N(s) \in \mathcal{BH}_{\infty} \right\},\,$$

where N(s) is a free parameter, and  $K_{\infty}^{r_1}(s)$  is represented as follows

$$K_{\infty}^{r_{1}}(s) = \mathcal{H}\mathcal{M}\left(\begin{bmatrix} \tilde{A}_{X} & B_{D_{2}} & -H_{\infty_{2}}\Sigma^{-1} \\ \tilde{F}_{\infty} & D_{12}^{\dagger} - D_{12}^{\bot}B_{D_{1}} & D_{12}^{\bot}H_{\infty_{1}}\Sigma^{-1} \\ \tilde{C}_{2} & O & \Sigma^{-1} \end{bmatrix}, N(s)\right),$$
(2.64)

and

$$\begin{split} A_X &= \begin{bmatrix} -A_{11} + L_{F_1} & 0 & 0 & 0 \\ A_{13} & & \\ 0 & & A_X \\ 0 & & \\ \end{bmatrix} \in \mathbb{R}^{n \times n}, \quad \tilde{A}_X \in \mathbb{R}^{(n - (m_2 - p_1)) \times (n - (m_2 - p_1))} \\ Z \hat{B}_2 D_{12}^{\dagger} &= \begin{bmatrix} -B_{D_1} \\ -\overline{B}_{D_2} \end{bmatrix}, \quad B_{D_1} \in \mathbb{R}^{(m_2 - p_1) \times p_1}, B_{D_2} \in \mathbb{R}^{(n - (m_2 - p_1)) \times p_1} \\ H_{\infty} &= \begin{bmatrix} -H_{\infty_1} \\ -\overline{H}_{\infty_2} \end{bmatrix}, \quad H_{\infty_1} \in \mathbb{R}^{(m_2 - p_1) \times p_2}, H_{\infty_2} \in \mathbb{R}^{(n - (m_2 - p_1)) \times p_2} \\ F_{\infty} &= \begin{bmatrix} -D_{12}^{\dagger} C_{1l} + D_{12}^{\perp} L_{F_1} & \tilde{F}_{\infty} \end{bmatrix} \in \mathbb{R}^{m_2 \times n}, \quad \tilde{F}_{\infty} \in \mathbb{R}^{m_2 \times (n - (m_2 - p_1))} \\ \hat{C}_2 &= \begin{bmatrix} C_{2l} & \tilde{C}_2 \end{bmatrix} \in \mathbb{R}^{p_2 \times n}, \quad \tilde{C}_2 \in \mathbb{R}^{p_2 \times (n - (m_2 - p_1))}. \end{split}$$

**Proof.** From the previous argument, the reduced order controller is derived by substituting  $L_{F_{22}}$  in (2.61) and W(s) in (2.63) into  $K_{\infty}^1(s)$  in (2.59). Since  $\mathcal{K}_{\infty}^{r_1}$  is a subset of  $\mathcal{K}_{\infty}^1$ , the reduced order controller is an  $H_{\infty}$  controller.

**Remark 2.5.2** The dynamical order of the central solution in  $\mathcal{K}_{\infty}^{r_1}$  is  $n - (m_2 - p_1)$ , which is lower than that of the central solution in  $\mathcal{K}_{\infty}^1$  by the number of  $m_2 - p_1$ . The controller order reduction is analogous to the order reduction in dual-observer-based controllers, where the order of the controller is reduced by the number of redundant inputs.

**Remark 2.5.3** The controller parameters are also represented as follows:

$$\begin{split} \tilde{A}_{X} &= \tilde{A}_{r} + \left(B_{1l}B_{1l}^{T} - B_{22l}B_{22l}^{T}\right)\tilde{X}_{r} \\ \tilde{C}_{2} &= C_{2r} + D_{21}B_{1l}^{T}\tilde{X}_{r} \\ \tilde{F}_{\infty} &= -D_{12}^{\dagger}C_{1r} + D_{12}^{\perp}L_{F_{2}} - D_{12}^{\dagger}B_{22l}^{T}\tilde{X}_{r} \\ H_{\infty_{1}} &= -\left\{Z_{11}B_{1uu} + Z_{12}B_{1l} + \left(S_{11}C_{2l}^{T} + S_{12}C_{2r}^{T}\right)\left(D_{21}^{\dagger}\right)^{T}\right\}D_{21}^{\dagger} \\ H_{\infty_{2}} &= -\left\{Z_{21}B_{1uu} + Z_{22}B_{1l} + \left(S_{21}C_{2l}^{T} + S_{22}C_{2r}^{T}\right)\left(D_{21}^{\dagger}\right)^{T}\right\}D_{21}^{\dagger} \\ B_{D_{1}} &= Z_{11}B_{22uu} + Z_{12}B_{22l} + S_{11}C_{1l}^{T} + S_{12}C_{1r}^{T} \\ B_{D_{2}} &= Z_{21}B_{22uu} + Z_{22}B_{22l} + S_{21}C_{1l}^{T} + S_{22}C_{1r}^{T}, \end{split}$$

where

$$\begin{split} \tilde{A}_r &= \begin{bmatrix} A_{14} & A_{23} & \tilde{A}_{24} \\ O & A_- & \tilde{A}_{42} \\ O & O & A_{X_r} \end{bmatrix}, \quad \tilde{X}_r = \begin{bmatrix} O & O & O \\ O & O & O \\ O & O & X_r \end{bmatrix} \in \mathbb{R}^{(n - (m_2 - p_1)) \times (n - (m_2 - p_1))} \\ S &= ZY = \begin{bmatrix} -S_{11} & S_{12} \\ -S_{21} & S_{22} \end{bmatrix}, \quad Z = \begin{bmatrix} -Z_{11} & Z_{12} \\ -Z_{21} & Z_{22} \end{bmatrix}, \quad L_F = \begin{bmatrix} -L_{F_1} \\ -L_{F_2} \end{bmatrix}, \end{split}$$

and  $X_r \ge O$  is a stabilizing solution of ARE in (2.55) and  $Y \ge O$  is a stabilizing solution of ARE in (2.57).

## 2.5.5 The controller structure

This subsection reviews the full-order controller and the reduced-order controller from the viewpoint of the dual-observer based output controller. The central solution in the class of full-order  $H_{\infty}$  controller in (2.58) is represented as

$$\begin{cases} \dot{\hat{x}} = \left(A_X + H_\infty \hat{C}_2\right) \hat{x} + H_\infty y\\ u = -F_\infty \hat{x} \end{cases},$$

where  $\hat{x}$  is a state variable of the controller. If we assume a orthogonal condition:  $D_{21}B_1^T = O$ , these equations can be written as

$$\begin{cases} \dot{\hat{x}} = A_X \hat{x} + \nu \\ u = -F_\infty \hat{x} \\ \nu = H_\infty \nu \\ \nu = y + C_2 \hat{x} \end{cases}$$
(2.65)

From these equations, it can be seen that the controller has the structure of a dual-observerbased controller. In order to interpret the structure of the controller, let us consider the system:

$$\begin{cases} \dot{x} = Ax + ZB_1w + B_2u + \left\{ B_1B_1^TXx + ZY \left( D_{21}^{\dagger}C_2 \right)^T w \right\} \\ y = C_2x + D_{21}w \end{cases}$$
(2.66)

Then let us construct a composite system from the systems (2.66) and (2.65). By letting a new state variable of the composite system as

$$\eta = x + \hat{x},$$

the dynamical equation of the composite system is given as follows:

$$\begin{cases} \begin{bmatrix} \dot{\eta} \\ \dot{\hat{x}} \end{bmatrix} = \begin{bmatrix} A + B_1 B_1^T X & O \\ O & A_X \end{bmatrix} \begin{bmatrix} \eta \\ \hat{x} \end{bmatrix} + \begin{bmatrix} Z \left\{ B_1 + Y \left( D_{21}^{\dagger} C_2 \right)^T \right\} \end{bmatrix} w + \begin{bmatrix} I \\ I \end{bmatrix} v \\ \nu = \begin{bmatrix} C_2 & O \end{bmatrix} \begin{bmatrix} \eta \\ \hat{x} \end{bmatrix} + D_{21} w \end{cases},$$



Figure 2.4: The composite system

where it should be noted that  $v \in \mathbb{R}^n$ , and that the subsystem of the state variable  $\hat{x}$  is unobservable. The composite system is illustrated in Figure 2.4. Hence, the composite system of the state variable  $\eta$  is stabilizable with an output feedback controller, if the pair  $(A + B_1 B_1^T X, C_2)$  is observable. In fact, the output feedback controller:

$$v = H_{\infty}\nu$$

stabilizes the system as follows:

$$\dot{\eta} = A_X \eta,$$

where  $A_X$  is a stable matrix. The size of the matrix  $A_X$  equals the dynamical order of a full-order observer.

By representing the class of the  $H_{\infty}$  controllers with the equation (2.64), the structure of the dual-observer can also be seen explicitly in the  $H_{\infty}$  controller which is represented with free parameters. Figure 2.5 illustrates the  $H_{\infty}$  controller with the dual-observer-based representation where the dynamical equation of the dual-observer is expressed explicitly. The size of the matrix  $A_X$  equals the dynamical order of a full-order dual-observer. Since  $B_K(s)$ is a function of the free parameters, the selection in the free parameters reduces to yield the uncontrollable subspace in the pair  $(A_X, B_K(s))$ , and the order of the controller can be



Figure 2.5: The structure of  $K^1_{\infty}(s)$ 

reduced by the number of the degrees of the subspace. Hence, this type of the reduced-order  $H_{\infty}$  controller can be interpreted as a minimal-order dual-observer-type  $H_{\infty}$  controller.

# **2.6** The non-standard $H_{\infty}$ problem of case 3

This section derives full-order and reduced-order controllers in the non-standard  $H_{\infty}$  problem of case 3. Since the problem of case 3 combines problems of cases 1 and 2, the procedure in the derivation of full-order controller is similar to the derivations in the problems of cases 1 and 2. However the reduced-order controller is slightly different from the reduced-order controllers in the problems of cases 1 and 2. Hence, details of the procedure in the derivation of the full-order controller are omitted, and almost all of discussions are devoted for the derivation of the reduced-order controller.

We treat a generalized plant:

$$G(s) = \begin{bmatrix} A & B_1 & B_{21} & B_{22} \\ \hline C_1 & O & O & I_{p_1} \\ \hline C_{21} & O & O & O \\ \hline C_{22} & I_{m_1} & O & O \end{bmatrix},$$

where the assumptions A1, A2'-case 3 and A3' are satisfied. We put variables as follows.

$$\begin{bmatrix} C_2 & D_{21} \end{bmatrix} = \begin{bmatrix} C_{21} & O \\ C_{22} & I_{m_1} \end{bmatrix}, \begin{bmatrix} -B_2 \\ -\overline{D}_{12} \end{bmatrix} = \begin{bmatrix} -B_{21} & -B_{22} \\ -\overline{O} & -\overline{I}_{p_1} \end{bmatrix}$$

## 2.6.1 Parametrization of full-order $H_{\infty}$ controller

Since the matrix  $D_{21}$  is of full column rank, we can derive a necessary condition for the solvability in the problem of case 3 in the same manner that is used in the problem of case 2. Thus we deduce the necessary condition by using the result of the FC problem.

As is seen in the problem of case 2, we can choose a matrix  $L_H$  such that the matrix

$$A_{ZH} = A - B_1 D_{21}^{\dagger} C_2 + L_H D_{21}^{\perp} C_2$$

satisfies

$$\begin{cases} UA_{ZH} = \Lambda U, \quad \lambda_i \left( \Lambda \right) < O, \forall i \\ \ker U \subset \ker U_o \end{cases}$$

where U is a row-basis of the stable subspace in  $A_{ZH}$ , and  $U_o$  is a row-basis of observable subspace of the pair  $\left(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2\right)$ . Also, we can choose a matrix  $E_H$  such that the matrices  $A_H = A_{ZH} + E_H D_{21}^{\dagger} C_2$  and  $E_H$  satisfy

$$\lambda_i (A_H) < 0, \forall i$$
$$UE_H = O. \tag{2.67}$$

,

**Lemma 2.6.1** Suppose that the matrices  $L_H$  and  $E_H$  are chosen such that the matrix  $A_H$  is stable. Then, if the non-standard  $H_{\infty}$  problem of case 3 is solvable, it is necessary that the ARE in (2.12) has a stabilizing solution  $Y \ge O$ .

**Proof.** From Lemma 2.4.1, in this case, it is also necessary that the FC problem for G(s) where  $B_{21} = I_n, B_{22} = O$  is solvable. The solvability condition is equivalent to the condition such that the  $H_{\infty}$  problem for  $M_2(s)$  is solvable. Hence, as is in the problem of case 2, it is necessary that the ARE in (2.12) has a stabilizing solution  $Y \ge O$ .

Suppose that the ARE in (2.12) has a stabilizing solution  $Y \ge O$ . Then we can choose a matrix  $E_H$  as

$$E_{H} = -Y \left( D_{21}^{\dagger} C_{2} \right)^{T}.$$
 (2.68)

The matrix  $E_H$  satisfies (2.67), and stabilizes the matrix  $A_H$ . Hence, let us choose  $E_H$  as in (2.68). Then, as is in the problem of case 2, Lemma 2.4.3 holds. Under the necessary condition such that the ARE in (2.12) has a nonnegative definite stabilizing solution Y, the  $H_{\infty}$  problem for G(s) is reduced to the  $H_{\infty}$  problem for  $G_{tmp}(s)$ :

$$G_{tmp}(s) = \begin{bmatrix} \hat{A} & B_1 - E_H & \hat{B}_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix},$$

where  $D_{12}$  is of full row rank, and

$$\hat{A} = A + YC_1^T C_1$$
$$\hat{B}_2 = B_2 + YC_1^T D_{12}$$

Hence, we consider the  $H_{\infty}$  problem for  $G_{tmp}(s)$ . As is in the problem of case 2, let us parametrize the stabilizing controllers for  $G_{tmp}(s)$ .

Suppose that  $(\hat{A}, \hat{B}_2)$  is stabilizable. Then we can choose a feedback gain F such that

$$A_F \triangleq \hat{A} + \hat{B}_2 F$$

is stable. Here, since  $D_{12}$  is of full row rank, as is in the problem of case 1, we choose F as

$$F = -D_{12}^{\dagger}C_1 + D_{12}^{\perp}L_F + D_{12}^{\dagger}E_F.$$

Then the matrix  $A_F$  is represented as

$$A_F = A_{ZF} + \hat{B}_2 D_{12}^{\dagger} E_F,$$
  
$$A_{ZF} = A - B_2 D_{12}^{\dagger} C_1 + B_2 D_{12}^{\perp} L_F,$$

where  $E_F$  is an arbitrary matrix which satisfies a constraint:

$$E_F V = O,$$

where V is a basis of the stable subspace in  $A_{ZF}$ , and  $L_F$  is chosen such that the controllable subspace of the pair  $\left(A - B_2 D_{12}^{\dagger} C_1, B_2 D_{12}^{\perp}\right)$  is stabilized.

Next, let us choose an observer gain  $H_{\infty}$  as in (2.23), then

$$A_H \triangleq \hat{A} + H_\infty C_2 = A_Y$$

is stable. From Lemma A.2.1, the class of stabilizing controllers is represented as follows.

$$K(s) = \mathcal{F}_l \left( \begin{bmatrix} A_Y + \hat{B}_2 F & -H_\infty & -\hat{B}_2 \\ F & O & -I \\ -C_2 & I & O \end{bmatrix}, Q(s) \right), Q(s) \in \mathcal{RH}_\infty$$
(2.69)

Then the closed loop transfer function is represented as follows

$$\mathcal{F}_{l}(G_{tmp}(s), K(s)) = \mathcal{F}_{l}(M_{3}(s), -D_{12}Q(s)D_{21}), \qquad (2.70)$$

where

$$M_{3}(s) = \begin{bmatrix} \hat{A} + \hat{B}_{2}F & B_{1} - E_{H} & \hat{B}_{2}D_{12}^{\dagger} \\ E_{F} & O & I \\ O & I & O \end{bmatrix}.$$
 (2.71)

From (2.70), the solvability conditions of the  $H_{\infty}$  problems for  $G_{tmp}(s)$  and for  $M_3(s)$  are equivalent, hence we consider the  $H_{\infty}$  problem for  $M_3(s)$ .

**Theorem 2.6.1** The non-standard  $H_{\infty}$  problem for the generalized plant which satisfies assumptions A1, A2'-case 3 and A3' is solvable if and only if an AREs in (2.52) and (2.12) have stabilizing solutions  $X \ge 0$  and  $Y \ge O$  which satisfy  $\rho(XY) < 1$ . Under the conditions, the class of the  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty} = \{K_{\infty}(s) : N(s) \in \mathcal{BH}_{\infty}, W_i(s) \in \mathcal{RH}_{\infty}, i = 1, 2, 3\}$$

where N(s) and  $W_i(s)$ ,  $\forall i$  are free parameters, and  $K_{\infty}(s)$  is represented as follows

$$K_{\infty}(s) = \mathcal{F}_{l} \left( \begin{bmatrix} \hat{A} + \hat{B}_{2}F_{\infty} + H_{\infty}C_{2} & H_{\infty} & -\hat{B}_{2}D_{12}^{\dagger} & -B_{2}D_{12}^{\bot} \\ - & -F_{\infty} & 0 & D_{12}^{\dagger} & D_{12}^{\dagger} & D_{12}^{\bot} \\ - & D_{21}^{\dagger}\hat{C}_{2}Z & D_{21}^{\dagger} & 0 & 0 \\ D_{21}^{\pm}\hat{C}_{2} & D_{21}^{\pm} & 0 & 0 \end{bmatrix}, \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix} \right)$$

$$(2.72)$$

and

$$\begin{split} \hat{A} &= A + Y C_1^T C_1 \\ \hat{B}_2 &= B_2 + Y C_1^T D_{12} \\ \hat{C}_2 &= D_{21} B_1^T X + C_2 \\ F_\infty &= \left\{ -D_{12}^{\dagger} C_1 - D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T X \right\} Z + D_{12}^{\perp} L_F \\ H_\infty &= -B_1 D_{21}^{\dagger} - Y \left( D_{21}^{\dagger} C_2 \right)^T D_{21}^{\dagger} + L_H D_{21}^{\perp} \\ Z &= (I - YX)^{-1} \,. \end{split}$$

**Proof.** The  $H_{\infty}$  problem for  $M_3(s)$  in (2.71) is DF (Disturbance Feedforward) problem. In appendix A.7 the solution for the DF problem is listed. By using the result from Lemma A.7.1, we derive the solution for the non-standard  $H_{\infty}$  problem of case 3.

(Solvability condition:) The solvability condition for the DF problem is such that ARE:

$$\tilde{S}A_{ZF} + A_{ZF}^T \tilde{S} + \tilde{S}\Upsilon \tilde{S} = O, \qquad (2.73)$$

where

$$\Upsilon \triangleq (B_1 - E_H) (B_1 - E_H)^T - \hat{B}_2 D_{12}^{\dagger} \left( \hat{B}_2 D_{12}^{\dagger} \right)^T$$

has a nonnegative definite stabilizing solution  $\tilde{S}$ . By using a definition in appendix A.8, we represent the solvability condition such that

$$\begin{cases} H_{\tilde{S}} \in \text{Dom}\left(\text{Ric}\right)\\ \tilde{S} = \text{Ric}\left(H_{\tilde{S}}\right) \ge O \end{cases},$$
(2.74)

where

$$H_{\tilde{S}} \triangleq \left[ \begin{array}{cc} A_{ZF} & \Upsilon \\ O & -A_{ZF}^T \end{array} \right]$$

 $H_{\tilde{S}}$  satisfies

$$\begin{bmatrix} I & Y \\ O & I \end{bmatrix} H_{\tilde{S}} = \begin{bmatrix} A_{ZF} & B_1 B_1^T - B_2 D_{12}^{\dagger} \begin{pmatrix} B_2 D_{12}^{\dagger} \end{pmatrix}^T + \delta \\ O & -A_{ZF}^T \end{bmatrix} \begin{bmatrix} I & Y \\ O & I \end{bmatrix},$$

where  $Y \ge O$  is a stabilizing solution of the ARE in (2.12), and  $\delta$  is defined as

$$\delta = \left( L_H D_{21}^{\perp} C_2 - B_2 D_{12}^{\dagger} L_F \right) Y + Y \left( L_H D_{21}^{\perp} C_2 - B_2 D_{12}^{\dagger} L_F \right)^T$$

Since a matrix  $U_o$ , which is a row-basis of observable subspace of the pair  $\left(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2\right)$ , satisfies  $U_o Y = O$ , the matrix  $D_{21}^{\perp} C_2$  satisfies

$$D_{21}^{\perp}C_2Y = O.$$

Hence, the solvability condition (2.74) is equivalent to

$$\begin{cases}
H_X \in \text{Dom}(\text{Ric}) \\
X = \text{Ric}\left(\tilde{H}_X\right) \ge O , \\
\rho(XY) < 1
\end{cases}$$
(2.75)

where  $\tilde{H}_X$  is defined as follows:

$$H_{X} \triangleq \begin{bmatrix} A_{ZF} & B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger} \left( B_{2}D_{12}^{\dagger} \right)^{T} - \left\{ B_{2}D_{12}^{\dagger}L_{F}Y + Y \left( B_{2}D_{12}^{\dagger}L_{F} \right)^{T} \right\} \\ O & -A_{ZF}^{T} \end{bmatrix},$$

and  $Y \ge O$  is a stabilizing solution of the ARE in (2.12). The solvability condition (2.75) implies that the ARE:

$$XA_{ZF} + A_{ZF}^{T}X + X \left\{ B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger} \left( B_{2}D_{12}^{\dagger} \right)^{T} \right\} X - X \left\{ B_{2}D_{12}^{\perp}L_{F}Y + Y \left( B_{2}D_{12}^{\perp}L_{F} \right)^{T} \right\} X = O \quad (2.76)$$

has a nonnegative definite stabilizing solution X, and which satisfies  $\rho(XY) < 1$ .

On the other hand, from the solvability condition in (2.74), it can be verified that the matrix  $\tilde{S} = \text{Ric}(H_{\tilde{S}})$  satisfies

$$SB_2D_{12}^\perp = O$$

hence

$$X = \operatorname{Ric}(H_X)$$
$$= \left(I + \tilde{S}Y\right)^{-1} \tilde{S}$$

satisfies

$$XB_2D_{12}^{\perp} = O. (2.77)$$

From the equation in (2.77), the ARE in (2.76) is reduced to the ARE in (2.52). Thus, the solvability condition (2.74) is equivalent to the condition:

$$\begin{cases} H_X \in \text{Dom} (\text{Ric}) \\ X = \text{Ric} (H_X) \ge O \\ \rho(XY) < 1 \end{cases},$$

where  $H_X$  is given as follows

$$H_X \triangleq \begin{bmatrix} A_{ZF} & B_1 B_1^T - B_2 D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T \\ O & -A_{ZF}^T \end{bmatrix}$$

This implies that the ARE in (2.52) has a stabilizing solution  $X \ge O$  where X satisfies  $\rho(XY) < 1$ .

Conversely if the solvability condition (2.75) is satisfied, it can be verified that  $H_{\tilde{S}} \in$ Dom (Ric). Hence, the matrix

$$A_{ZF} - \hat{B}_2 D_{12}^{\dagger} \left( \hat{B}_2 D_{12}^{\dagger} \right)^T \tilde{S} = \hat{A} + \hat{B}_2 \left\{ -D_{12}^{\dagger} C_1 + D_{12}^{\perp} L_F - D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T \tilde{S} \right\}$$

is stable, hence  $(\hat{A}, \hat{B}_2)$  is stabilizable.

(Derivation of the controller:) In the beginning of this subsection, the feedback gain F is selected such that  $A_F = \hat{A} + \hat{B}_2 F$  is stable. Here, we choose F based on the solution of the ARE in (2.52). Suppose that the solvability condition (2.31) is satisfied. Now, we select F such that

$$F = F_{\infty} \triangleq -D_{12}^{\dagger}C_{1} + D_{12}^{\perp}L_{F} - D_{12}^{\dagger} \left(\hat{B}_{2}D_{12}^{\dagger}\right)^{T} \tilde{S},$$

where

$$\tilde{S} = XZ$$
$$Z = (I - YX)^{-1}$$

Then  $A_F = \hat{A} + \hat{B}_2 F_{\infty}$  is stable. By using the formula in Lemma A.7 we can derive the controller

$$-D_{12}Q(s)D_{21} = \mathcal{F}_l \left( \begin{bmatrix} \hat{A} + \hat{B}_2 F_\infty & -(B_1 - E_H) & -\hat{B}_2 D_{12}^{\dagger} \\ \hline O & O & I \\ (B_1 - E_H)^T \tilde{S} & I & O \end{bmatrix}, N(s) \right), \quad (2.78)$$

where  $N(s) \in \mathcal{BH}_{\infty}$ . The equation (2.78) can be represented as follows:

$$-D_{12}Q(s)D_{21} = (I - N(s)M_{22}(s))^{-1}N(s)M_{21}(s), \qquad (2.79)$$

where

$$M_{21}(s) = I - \Phi(s) (B_1 - E_H)$$
  

$$M_{22}(s) = -\Phi(s)\hat{B}_2 D_{12}^{\dagger}$$
  

$$\Phi(s) = (B_1 - E_H)^T \tilde{S} \left(sI - \hat{A} - \hat{B}_2 F_\infty\right)^{-1}$$

In the equation (2.79),  $D_{12}$  is of full row rank and  $D_{21}$  is of full column rank, hence the solution of Q(s) is derived as follows:

$$Q(s) = -\begin{bmatrix} D_{12}^{\dagger} & D_{12}^{\perp} \end{bmatrix} \begin{bmatrix} (I - N(s)M_{22}(s))^{-1}N(s)M_{21}(s) & \hat{W}_{1}(s) \\ \hat{W}_{2}(s) & \hat{W}_{3}(s) \end{bmatrix} \begin{bmatrix} D_{21}^{\dagger} \\ D_{21}^{\perp} \end{bmatrix},$$

where  $\hat{W}_i(s) \in \mathcal{RH}_{\infty}$ . Since  $(I - M_{22}(s)N(s))^{-1} \in \mathcal{RH}_{\infty}$ , we can replace  $\hat{W}_i(s)$  as

$$\hat{W}_{1}(s) = (I - N(s)M_{22}(s))^{-1} (W_{1}(s) + N(s)\Phi(s)L_{H}) 
\hat{W}_{2}(s) = W_{2}(s) \left\{ I + M_{22}(s) (I - N(s)M_{22}(s))^{-1} N(s) \right\} M_{21}(s) 
\hat{W}_{3}(s) = W_{1}(s)M_{22}(s) (I - N(s)M_{22}(s))^{-1} (W_{1}(s) + N(s)\Phi(s)L_{H}) 
+ W_{3}(s) + W_{1}(s)\Phi(s)L_{H},$$

where  $W_i(s) \in \mathcal{RH}_{\infty}$  are free parameters. Then the solution Q(s) is represented as follows:

$$Q(s) = -\begin{bmatrix} D_{12}^{\dagger} & D_{12}^{\perp} \end{bmatrix} \left\{ I - \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix} \begin{bmatrix} M_{22}(s) & -\Phi(s)\hat{B}_{2}D_{12}^{\perp} \\ O & O \end{bmatrix} \right\}^{-1} \\ \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix} \begin{bmatrix} D_{21}^{\dagger} + \Phi(s)H_{\infty} \\ D_{21}^{\perp} \end{bmatrix} \\ = \mathcal{F}_{l} \left( \begin{bmatrix} \hat{A} + \hat{B}_{2}F_{\infty} & -H_{\infty} & -\hat{B}_{2}D_{12}^{\dagger} & -\hat{B}_{2}D_{12}^{\dagger} \\ -\frac{O}{(B_{1} - E_{H})^{T}} \hat{S} & -\frac{O}{-D_{21}^{\dagger}} & -\frac{D_{12}^{\dagger}}{O} & O \\ -D_{21}^{\perp} & O & O \end{bmatrix} , \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix} \right),$$
(2.80)

where  $N(s) \in \mathcal{BH}_{\infty}, W_i(s) \in \mathcal{RH}_{\infty}$ . By applying this Q(s) into (2.69), we can derive the  $H_{\infty}$  controller in (2.72).

# 2.6.2 Derivation of reduced-order $H_{\infty}$ controllers based on characterization of zeros in $G_{21}(s)$

This subsection derives the reduced-order non-standard  $H_{\infty}$  controller. Since  $D_{21}$  is of full column rank, it is assumed that the generalized plant has been put into a basis corresponding to the canonical form in equation (2.4). Hence the following equations can be assumed without loss of generality

$$A - B_1 D_{21}^{\dagger} C_2 = \begin{bmatrix} -A_{11} & A_{12} & O & O \\ -A_{13} & A_{14} & O & -O \\ A_{31} & A_{32} & A_- & O \\ A_{33} & A_{34} & O & A_+ \end{bmatrix}, A_{11} \in \mathbb{R}^{(p_2 - m_1) \times (p_2 - m_1)}$$
$$D_{21}^{\perp} C_2 = \begin{bmatrix} I_{p_2 - m_1} & O & O & O \\ C_{22lr} & C_{22rl} & C_{22rr} \end{bmatrix} \triangleq \begin{bmatrix} C_{22ll} & C_{22r} \end{bmatrix}, C_{22ll} \in \mathbb{R}^{p_2 \times (p_2 - m_1)}$$

where  $\lambda_i(A_{14}) < 0$ , both of the pairs  $(A_{11}, I_{p_2-m_1})$  and  $(A_+, C_{22rr})$  are observable, and  $A_+ \in \mathbb{R}^{l \times l}$  has no  $j\omega$  eigenvalues. We call  $A_-$  the stable zero mode of  $G_{21}(s)$  and  $A_+$  the unstable zero mode of  $G_{21}(s)$ .

,

For notational ease, let us partition the following matrices.

$$B_{1} = \begin{bmatrix} B_{1u} \\ B_{1l} \end{bmatrix}, B_{1u} \in \mathbb{R}^{(p_{2}-m_{1})\times m_{1}}$$

$$B_{2} = \begin{bmatrix} B_{2u} \\ B_{2l} \end{bmatrix}, B_{2u} \in \mathbb{R}^{(p_{2}-m_{1})\times m_{2}}$$

$$C_{1} = \begin{bmatrix} C_{1ll} & C_{1lr} & C_{1rl} & C_{1rr} \end{bmatrix} \triangleq \begin{bmatrix} C_{1ll} & C_{1r} \end{bmatrix}, C_{1ll} \in \mathbb{R}^{p_{1}\times(p_{2}-m_{1})}$$

## 2.6. THE NON-STANDARD $H_{\infty}$ PROBLEM OF CASE 3

Under the above preparation, let us represent the full-order  $H_{\infty}$  controller in (2.72) with the dual homogeneous transformation as follows:

$$K_{\infty}(s) = \mathcal{DHM}\left(\begin{bmatrix} A_{Y} & \hat{B}_{2} & H_{\infty} \\ \hline D_{12} & D_{12} & O \\ (D_{12}^{\perp})^{T} & F_{\infty} & (D_{12}^{\perp})^{T} & O \\ (D_{12}^{\perp})^{T} & O & D_{21}^{\dagger} \\ \hline D_{21}^{\dagger} \tilde{C}_{2Z} & O & D_{21}^{\dagger} \\ D_{21}^{\dagger} C_{2} & O & D_{21}^{\dagger} \end{bmatrix}, \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix}\right)$$
$$= \begin{bmatrix} A_{Y} & \hat{B}_{2} \\ \hline C_{K}(s) & D_{12} \\ (D_{12}^{\perp})^{T} \end{bmatrix}^{-1} \begin{bmatrix} A_{Y} & H_{\infty} \\ \hline C_{K}(s) & [N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix} \begin{bmatrix} D_{21}^{\dagger} \\ D_{21}^{\dagger} \end{bmatrix}, \quad (2.81)$$

where  $C_K(s)$  is defined as follows.

$$C_{K}(s) \triangleq - \begin{bmatrix} D_{12} \\ (D_{12}^{\perp})^{T} \end{bmatrix} F_{\infty} + \begin{bmatrix} N(s) \\ W_{2}(s) \end{bmatrix} D_{21}^{\dagger} \hat{C}_{2} Z + \begin{bmatrix} W_{1}(s) \\ W_{3}(s) \end{bmatrix} D_{21}^{\perp} C_{2}.$$
(2.82)

In the equation (2.20), since the matrix  $L_{H_{33}} \in \mathbb{R}^{l \times (p_2 - m_1)}$  is arbitrary, we can choose  $L_{H_{33}}$  such that  $\tilde{A}_{33} = O$  holds. Hence let us choose  $L_{H_{33}}$ 

$$L_{H_{33}} = -A_{33} - Y_r \left( C_{1rr}^T C_{1ll} - C_{22rr}^T C_{22ll} \right), \qquad (2.83)$$

then the matrix  $A_Y$  is represented as

$$A_Y = \begin{bmatrix} A_{11} + L_{H_1} & A_{12} & O & O \\ O & A_{14} & O & O \\ O & A_{32} & A_- & O \\ O & \tilde{A}_{34} & \tilde{A}_{43} & A_{Y_r} \end{bmatrix}.$$

The matrix  $A_Y$  satisfies

$$A_Y V_Y = V_Y \left( A_{11} + L_{H_1} \right),$$

where 
$$V_Y = \begin{bmatrix} I_{p_2-m_1} \\ O \end{bmatrix} \in \mathbb{R}^{n \times (p_2-m_1)}$$
. Hence, if  $C_K(s)$  satisfies  
 $C_K(s)V_Y = O,$  (2.84)

the pair  $(A_Y, C_K(s))$  is not observable, hence the order of the controller  $K_{\infty}(s)$  is reduced by the number of  $p_2 - m_1$ .

From equation (2.82), the matrix  $C_K(s)$  is represented with the free parameter  $W_i(s) \in \mathcal{RH}_{\infty}$ , and is spanned by the row-basis of  $D_{21}^{\perp}C_2$ . Since  $D_{21}^{\perp}C_2V = I_{p_2-m_1}$  holds, if we choose the free parameters  $W_1(s)$  and  $W_3(s)$  as

$$\begin{bmatrix} W_1(s) \\ W_3(s) \end{bmatrix} = -\left(-\begin{bmatrix} D_{12} \\ (D_{12}^{\perp})^T \end{bmatrix} F_{\infty} + \begin{bmatrix} N(s) \\ W_2(s) \end{bmatrix} D_{21}^{\dagger} \hat{C}_2 Z\right) V_Y,$$
(2.85)

the matrix  $V_Y$  satisfies (2.84). Thus we can derive a reduced-order  $H_{\infty}$  controller.

**Theorem 2.6.2** Under the same solvability condition as in Theorem 2.6.1, the class of reduced-order  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty}^{r_{32}} = \left\{ K_{\infty}^{r_{32}}(s) : N(s) \in \mathcal{BH}_{\infty}, W_2(s) \in \mathcal{RH}_{\infty} \right\},\$$

where N(s) is a free parameter, and  $K^{r_{32}}_{\infty}(s)$  is represented as follows

$$K_{\infty}^{r_{32}}(s) = \mathcal{DHM}\left(M_{\infty}^{r_{32}}(s), \begin{bmatrix} N(s) \\ W_2(s) \end{bmatrix}\right),$$
(2.86)

and

$$\begin{split} M_{\infty}^{r_{32}}(s) &= \begin{bmatrix} \tilde{A}_{Y} & \tilde{B}_{2} & \tilde{H}_{\infty} \\ \hline D_{12} & D_{12} & D_{12} \\ (D_{12}^{\perp})^{T} & F_{\infty_{2}} & (D_{12}^{\perp})^{T} & F_{\infty_{1}}D_{21}^{\perp} \\ (D_{12}^{\perp})^{T} & D_{21}^{\perp} & F_{\infty_{1}}D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{21}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{22}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{22}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{22}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{21}^{\perp} & D_{21}^{\perp} \\ \hline D_{21}^{\perp} & D_{21}^{\perp} &$$

**Proof.** From the previous argument, the reduced order controller is derived by substituting  $L_{H_{33}}$  in (2.83) and  $W_1(s)$  and  $W_3(s)$  in (2.85) into  $K_{\infty}(s)$  in (2.81). Since  $\mathcal{K}_{\infty}^{r_{32}}$  is a subset of  $\mathcal{K}_{\infty}$ , the reduced order controller is an  $H_{\infty}$  controller.

**Remark 2.6.1** The dynamical order of the central solution in  $\mathcal{K}_{\infty}^{r_{32}}$  is  $n - (p_2 - m_1)$ , which is lower than that of the central solution in  $\mathcal{K}_{\infty}$  by the number of  $p_2 - m_1$ . The controller order reduction is analogous to the order reduction in observer-based controllers, where the order of the controller is reduced by the number of independent outputs that are not corrupted by disturbances.

# 2.6.3 Derivation of reduced-order $H_{\infty}$ controllers based on characterization of zeros in $G_{12}(s)$

This subsection derives the reduced-order non-standard  $H_{\infty}$  controller.

**Corollary 2.6.1** Under the same condition in Theorem 2.6.1, the class of the  $H_{\infty}$  controllers is also represented as follows:

$$\mathcal{K}_{\infty} = \{ K_{\infty}(s) : N(s) \in \mathcal{BH}_{\infty}, W_i(s) \in \mathcal{RH}_{\infty}, i = 1, 2, 3 \}$$

## 2.6. THE NON-STANDARD $H_{\infty}$ PROBLEM OF CASE 3

where N(s) and  $W_i(s)$ ,  $\forall i$  are free parameters, and  $K_{\infty}(s)$  is represented as follows

$$K_{\infty}(s) = \mathcal{F}_l \left( M_{\infty}^1(s), \begin{bmatrix} N(s) & W_1(s) \\ W_2(s) & W_3(s) \end{bmatrix} \right)$$
(2.87)

and

$$M_{\infty}^{1}(s) = \begin{bmatrix} \frac{\hat{A} + B_{2}F_{\infty} + H_{\infty}\hat{C}_{2} & -H_{\infty} & Z\hat{B}_{2}D_{12}^{\dagger} & B_{2}D_{12}^{\dagger} \\ F_{\infty} & O & D_{12}^{\dagger} & D_{12}^{\dagger} \\ -D_{21}^{\dagger}\hat{C}_{2} & D_{21}^{\dagger} & O & O \\ -D_{21}^{\dagger}\hat{C}_{2} & D_{21}^{\dagger} & O & O \\ D_{21}^{\dagger} & O & O \\ \end{bmatrix}$$
$$\hat{A} = A + B_{1}B_{1}^{T}X$$
$$\hat{B}_{2} = B_{2} + YC_{1}^{T}D_{12}$$
$$\hat{C}_{2} = D_{21}B_{1}^{T}X + C_{2}$$
$$F_{\infty} = -D_{12}^{\dagger}C_{1} + D_{12}^{\dagger}L_{F} - D_{12}^{\dagger}\left(B_{2}D_{12}^{\dagger}\right)^{T}X$$
$$H_{\infty} = Z\left\{-B_{1}D_{21}^{\dagger} - Y\left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}\right\} + L_{H}D_{21}^{\dagger}$$
$$Z = (I - YX)^{-1}.$$

Proof. Omitted.

Since  $D_{12}$  is of full row rank, it is assumed that the generalized plant has been put into a basis corresponding to the canonical form in equation (2.3). Hence the following equations can be assumed without loss of generality

$$A - B_2 D_{12}^{\dagger} C_1 = \begin{bmatrix} A_{11} & A_{12} & A_{21} & A_{22} \\ A_{13} & A_{14} & A_{23} & A_{24} \\ O & O & A_- & O \\ O & O & O & A_+ \end{bmatrix}, A_{11} \in \mathbb{R}^{(m_2 - p_1) \times (m_2 - p_1)}$$
$$B_2 D_{12}^{\perp} = \begin{bmatrix} I_{m_2 - p_1} \\ O \\ O \\ O \end{bmatrix}$$
$$B_2 D_{12}^{\dagger} = \begin{bmatrix} I_{m_2 - p_1} \\ O \\ O \\ O \end{bmatrix}$$
$$B_2 D_{12}^{\dagger} = \begin{bmatrix} I_{m_2 - p_1} \\ O \\ O \\ B_{22lu} \\ B_{22lu} \\ B_{22ll} \end{bmatrix} \triangleq \begin{bmatrix} B_{22uu} \\ B_{22lu} \\ B_{22l} \end{bmatrix}, B_{22uu} \in \mathbb{R}^{(m_2 - p_1) \times m_2}$$

where  $\lambda_i(A_{14}) < 0$ , both of the pairs  $(A_{11}, I_{m_2-p_1})$  and  $(A_+, B_{22ll})$  are controllable, and  $A_+$ has no  $j\omega$  eigenvalues. We call  $A_-$  the stable zero mode of  $G_{12}(s)$  and  $A_+$  unstable zero mode of  $G_{12}(s)$ .

For notational ease, let us partition the following matrices.

$$B_{1} = \begin{bmatrix} B_{1uu} \\ \overline{B}_{1ul} \\ B_{1lu} \\ B_{1lu} \\ B_{1ll} \end{bmatrix} \triangleq \begin{bmatrix} B_{1uu} \\ B_{1l} \end{bmatrix}, B_{1uu} \in \mathbb{R}^{(m_{2}-p_{1})\times m_{1}}$$
$$C_{1} = \begin{bmatrix} C_{1l} & C_{1r} \end{bmatrix} \in \mathbb{R}^{p_{1}\times n}, C_{1l} \in \mathbb{R}^{p_{1}\times (m_{2}-p_{1})}$$
$$C_{2} = \begin{bmatrix} C_{2l} & C_{2r} \end{bmatrix} \in \mathbb{R}^{p_{2}\times n}, C_{2} \in \mathbb{R}^{p_{2}\times (m_{2}-p_{1})}.$$

Under the above preparations, let us represent the full-order  $H_{\infty}$  controller in (2.87) with the homogeneous transformation as follows:

$$K_{\infty}(s) = \mathcal{H}\mathcal{M}\left(\tilde{M}_{\infty}^{1}(s), \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix}\right)$$
$$= \begin{bmatrix} A_{X} & B_{K}(s) \\ F_{\infty} & \begin{bmatrix} D_{12}^{\dagger} & D_{12}^{\perp} \end{bmatrix} \begin{bmatrix} N(s) & W_{1}(s) \\ W_{2}(s) & W_{3}(s) \end{bmatrix}} \begin{bmatrix} A_{X} & B_{K}(s) \\ \hline \hat{C}_{2} & \begin{bmatrix} D_{21} & (D_{21}^{\perp})^{T} \end{bmatrix} \end{bmatrix}^{-1},$$

$$(2.88)$$

where  $\tilde{M}^1_{\infty}(s)$  and  $B_K(s)$  are defined as

$$\tilde{M}_{\infty}^{1}(s) \triangleq \begin{bmatrix}
\frac{A_{X} \quad Z\hat{B}_{2}D_{12}^{\dagger} \quad B_{2}D_{12}^{\perp} & -H_{\infty}D_{21} & -H_{\infty}\left(D_{21}^{\perp}\right)^{T} \\
\frac{F_{\infty}}{\hat{C}_{2}} & -\frac{D_{12}^{\dagger}}{O} & -\frac{D_{12}}{O} & -\frac{O}{D_{21}} & 0 \\
B_{K}(s) \triangleq Z\hat{B}_{2}D_{12}^{\dagger} \left[ N(s) \quad W_{1}(s) \right] \\
+B_{2}D_{12}^{\perp} \left[ W_{2}(s) \quad W_{3}(s) \right] - H_{\infty} \left[ D_{21} \quad \left(D_{21}^{\perp}\right)^{T} \right].$$
(2.89)

In the equation (2.56), since the matrix  $L_{F_{22}} \in \mathbb{R}^{(m_2-p_1)\times r}$  is arbitrary, we can choose  $L_{F_{22}}$  such that  $\tilde{A}_{22} = O$  holds. Hence let us choose  $L_{F_{22}}$ 

$$L_{F_{22}} = -A_{22} - \left(B_{1uu}B_{1ll}^T - B_{22uu}B_{22ll}^T\right)X_r$$
(2.90)

then the matrix  $A_X$  is represented as

$$A_X = \begin{bmatrix} A_{11} + L_{F_1} & O & O & O \\ A_{13} & A_{14} & A_{23} & \tilde{A}_{24} \\ O & O & A_- & \tilde{A}_{42} \\ O & O & O & A_{X_r} \end{bmatrix}.$$

The matrix  $A_X$  satisfies

$$U_X A_X = (A_{11} + L_{F_1}) \, U_X,$$

where  $U_X = \begin{bmatrix} I_{m_2-p_1} & O \end{bmatrix} \in \mathbb{R}^{(m_2-p_1) \times n}$ . Hence, if  $B_K(s)$  satisfies

$$U_X B_K(s) = O, (2.91)$$

56

the pair  $(A_X, B_K(s))$  is not controllable, hence the order of the controller in  $\mathcal{K}_{\infty}$  is reduced by the number of  $m_2 - p_1$ .

From equation (2.89), the matrix  $B_K(s)$  is represented with the free parameter  $W_i(s) \in \mathcal{RH}_{\infty}$ , and is spanned by the basis of  $B_2 D_{12}^{\perp}$ . Since  $U_X B_2 D_{12}^{\perp} = I_{m_2-p_1}$  holds, if we choose the free parameters  $W_2(s)$  and  $W_3(s)$  as

$$\begin{bmatrix} W_2(s) & W_3(s) \end{bmatrix} = -U_X \left( Z \hat{B}_2 D_{12}^{\dagger} \begin{bmatrix} N(s) & W_1(s) \end{bmatrix} - H_{\infty} \begin{bmatrix} D_{21} & (D_{21}^{\perp})^T \end{bmatrix} \right), \quad (2.92)$$

the matrix  $U_X$  satisfies (2.91). Thus we can derive a reduced-order  $H_{\infty}$  controller.

**Theorem 2.6.3** Under the same solvability condition as in Theorem 2.6.1, the class of reduced-order  $H_{\infty}$  controllers is represented as:

$$\mathcal{K}_{\infty}^{r_{31}} = \left\{ K_{\infty}^{r_{31}}(s) : N(s) \in \mathcal{BH}_{\infty}, W_1(s) \in \mathcal{RH}_{\infty} \right\},\$$

where N(s) is a free parameter, and  $K_{\infty}^{r_{31}}(s)$  is represented as follows

$$K_{\infty}^{r_{31}}(s) = \mathcal{H}\mathcal{M}\left(M_{\infty}^{r_{31}}(s), \begin{bmatrix} N(s) & W_1(s) \end{bmatrix}\right),$$
(2.93)

and

$$\begin{split} M_{\infty}^{r_{31}}(s) &= \begin{bmatrix} \tilde{A}_{X} & B_{D_{2}} & -H_{\infty_{2}} \begin{bmatrix} D_{21} & (D_{21}^{\perp})^{T} \\ D_{21} & (D_{21}^{\perp})^{T} \end{bmatrix} \\ \tilde{F}_{\infty} & D_{12}^{\dagger} - D_{12}^{\perp} B_{D_{1}} & D_{12}^{\perp} H_{\infty_{1}} \begin{bmatrix} D_{21} & (D_{21}^{\perp})^{T} \\ D_{21} & (D_{21}^{\perp})^{T} \end{bmatrix} \end{bmatrix} \\ A_{X} &= \begin{bmatrix} -A_{11} + L_{F_{1}} & 0 & 0 & 0 \\ -\tilde{A}_{13} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \end{bmatrix} \in \mathbb{R}^{n \times n}, \quad \tilde{A}_{X} \in \mathbb{R}^{(n - (m_{2} - p_{1})) \times (n - (m_{2} - p_{1}))} \\ Z\hat{B}_{2}D_{12}^{\dagger} &= \begin{bmatrix} -B_{D_{1}} \\ -\bar{B}_{D_{2}} - \end{bmatrix}, \quad B_{D_{1}} \in \mathbb{R}^{(m_{2} - p_{1}) \times p_{1}, B_{D_{2}} \in \mathbb{R}^{(n - (m_{2} - p_{1})) \times p_{1}} \\ H_{\infty} &= \begin{bmatrix} -H_{\infty_{1}} \\ -\bar{H}_{\infty_{2}} - \end{bmatrix}, \quad H_{\infty_{1}} \in \mathbb{R}^{(m_{2} - p_{1}) \times p_{2}, H_{\infty_{2}} \in \mathbb{R}^{(n - (m_{2} - p_{1})) \times p_{2}} \\ F_{\infty} &= \begin{bmatrix} -D_{12}^{\dagger}C_{1l} + D_{12}^{\perp}L_{F_{1}} & \tilde{F}_{\infty} \end{bmatrix} \in \mathbb{R}^{m_{2} \times n}, \quad \tilde{F}_{\infty} \in \mathbb{R}^{m_{2} \times (n - (m_{2} - p_{1}))} \\ \hat{C}_{2} &= \begin{bmatrix} C_{2l} & \tilde{C}_{2} \end{bmatrix} \in \mathbb{R}^{p_{2} \times n}, \quad \tilde{C}_{2} \in \mathbb{R}^{p_{2} \times (n - (m_{2} - p_{1}))}. \end{split}$$

**Proof.** From the previous argument, the reduced order controller is derived by substituting  $L_{F_{22}}$  in (2.90) and  $W_2(s)$  and  $W_3(s)$  in (2.92) into  $K_{\infty}(s)$  in (2.88). Since  $\mathcal{K}_{\infty}^{r_{31}}$  is a subset of  $\mathcal{K}_{\infty}$ , the reduced order controller is an  $H_{\infty}$  controller.

**Remark 2.6.2** The dynamical order of the central solution in  $\mathcal{K}_{\infty}^{r_{31}}$  is  $n - (m_2 - p_1)$ , which is lower than that of the central solution in  $\mathcal{K}_{\infty}$  by the number of  $m_2 - p_1$ . The controller order reduction is analogous to the order reduction in dual-observer-based controllers, where the order of the controller is reduced by the number of redundant inputs.



Figure 2.6: The magnetic levitation system

# 2.7 A numerical example and discussions

## 2.7.1 Magnetic levitation system

In this thesis, we use a model of a magnetic levitation system to examine the effectiveness of proposed methods by numerical examples. Let us consider the magnetic levitation system[34] shown in the Figure 2.6, where the vertical position  $x_m[m]$  of a steel ball is controlled by operating the terminal voltage of an electromagnet e[V].  $R[\Omega]$  is a resistance of the excitation circuit and L[H] is an inductance of the coil and i[A] is an excitation current of the electromagnet. M[kg] is mass of the steel ball. Let us assume that the movement of the ball is restricted to the vertical axis, then we can describe the equation of the motion

$$\begin{cases} M\frac{d^2x_m}{dt^2} = Mg + \frac{1}{2}i^2\frac{\partial L}{\partial x_m} \\ e = Ri + \frac{d}{dt}(Li) \end{cases}.$$

Also, let us assume that the inductance L satisfies the equation

$$L(x_m) = \frac{Q}{X + x_m} + L_0,$$

where  $Q, X, L_0$  are some parameters.

The state variables are defined as follows

$$x \triangleq \begin{bmatrix} x_1 & x_2 & x_3 \end{bmatrix}^T = \begin{bmatrix} i & \dot{x}_m & x_m \end{bmatrix}^T$$

and the input variable u and the output variable y are defined as follows.

$$u = e, \quad y = x_m$$

### 2.7. A NUMERICAL EXAMPLE AND DISCUSSIONS

Then the state space equation of the magnetic levitation system is written as follows.

$$\begin{cases} \dot{x} = \begin{bmatrix} \frac{\{Qx_2 - (X+x_3)^2 R\}x_1}{Q(X+x_3) + L_0(X+x_3)^2} \\ g - \frac{Qx_1^2}{2M(X+x_3)^2} \\ x_2 \end{bmatrix} + \begin{bmatrix} \frac{X+x_3}{L_0(X+x_3) + Q} \\ 0 \\ 0 \end{bmatrix} u \\ y = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix} x \end{cases}$$

Thus, the system is represented as a nonlinear state space equation. By expanding the nonlinear equation into Taylor series around an equilibrium point

$$x_{eq} = \begin{bmatrix} x_1^0 & 0 & x_3^0 \end{bmatrix}^T, \quad u_{eq},$$

and by approximating high-order terms, a linear approximation form can be given as follows

$$\begin{cases} \dot{\bar{x}} = A\bar{x} + B\bar{u} \\ \bar{y} = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix} \bar{x} \end{cases},$$
(2.94)

where A and B are defined as

$$A \triangleq \begin{bmatrix} \frac{-X_c R}{Q + L_0 X_c} & \frac{x_3^0 Q}{(Q + L_0 X_c) X_c} & 0\\ \frac{-Q x_3^0}{M X_c^2} & 0 & \frac{Q x_3^{0^2}}{M X_c^3}\\ 0 & 1 & 0 \end{bmatrix} = \begin{bmatrix} a_{11} & a_{12} & 0\\ a_{21} & 0 & a_{23}\\ 0 & 1 & 0 \end{bmatrix}$$
$$B \triangleq \begin{bmatrix} \frac{X_c}{Q + L_0 X_c}\\ 0\\ 0 \end{bmatrix} = \begin{bmatrix} b_1\\ 0\\ 0 \end{bmatrix},$$

and  $X_c, \bar{x}, \bar{u}, \bar{y}$  are defined as follows

$$\begin{aligned} X_c &\triangleq X + x_1^0 \\ \bar{x} &\triangleq x - x_{eq} \\ \bar{u} &\triangleq u - u_{eq} \\ \bar{y} &\triangleq y - y_{eq}. \end{aligned}$$

By using the Laplace transformation for the equation (2.94), we can obtain a set of transfer functions from the input  $\bar{u}$  to the output  $\bar{y}$ :

$$\mathcal{P}(s; M, R, Q, L_0, X) \triangleq \begin{bmatrix} 0 & 0 & 1 \end{bmatrix} (sI - A)^{-1}B.$$

By substituting the values of the parameters in Table 2.1 into the above equation, we can obtain a transfer function, which we call a nominal model, as follows

$$P(s) = \frac{-67.03}{(s - 47.8)(s + 46.3)(s + 14.9)}.$$
(2.95)
М	R	Q	$L_0$	Х	$x_{1}^{0}$	$x_{3}^{0}$
0.54	11.6	$8.513\times10^{-4}$	0.789	0.0043	1.144	0.006

Table 2.1: The values of parameters

### 2.7.2 An uncertain plant with partial state measurement

For the linear model of the magnetic levitation system, in this chapter we assume that some of the sates are measured without disturbances, and the other states are measured with being effected by disturbances. Thus the model  $\tilde{P}(s)$  which we assume here can be written as follows

$$\tilde{P}(s): \begin{cases} \dot{x} = Ax + Bu \\ y_1 = x_1 \\ y_2 = x_2 \\ y_3 = x_3 + w \end{cases}$$

where, w indicates an input of the disturbance that is caused by the approximation error or measurement noise. The variables  $x = \begin{bmatrix} x_1 & x_2 & x_3 \end{bmatrix}^T$  and u correspond to each variable with upper ver in the equation (2.94).



Figure 2.7: The plant model

We assume that the dynamical equation of the disturbance w can be written as follows

$$\Delta(s)P(s): \begin{cases} \dot{x} = Ax + Bu\\ w = C_w x + d_w u \end{cases}$$

where,  $\Delta(s)$  indicates a model of a perturbation that causes the disturbance. Here, it should be noted that both states in the plant and the perturbation are common. Then the whole plant model is illustrated in Figure 2.7. In this chapter, we assume that a transfer function of the error system between the nominal model P(s) and the perturbed model:

$$\begin{cases} \dot{x} = Ax + Bu\\ y_3 = x_3 + w\\ w = C_w x + d_w u \end{cases}$$



Figure 2.8: Gain plots of the perturbation and its boundary

has already been identified, and is plotted in Figure 2.8 with red line.

For the error system, we set a boundary of the perturbation such that a partial system from the input u to the output  $y_3$  varies in a set

$$\mathcal{P} = \{ (1 + \Delta(s)) P(s) : |\Delta(j\omega)| \le |W_T(j\omega)|, \forall \omega \}, \qquad (2.96)$$

where  $W_T(s)$  is a function of the relative error bound between the perturbed model and P(s). In this thesis, by trial and error, we select  $W_T(s)$  as

$$W_T(s) = 0.2 \times 10^{-8} (5.0s^3 + 2.0 \times 10^4 s^2 + 1.0 \times 10^6 s + 10),$$

such that the plant set  $\mathcal{P}$  includes the perturbed plant. Figure 2.8 plots the gain of the weighting function  $W_T(s)P(s)$  with green line.

#### 2.7.3 Low order robust controller design

We construct a generalized plant G(s) for the robust controller design as shown in Figure 2.9. Then, by solving the problem of finding a stabilizing controller K(s) such that  $\|\mathcal{F}_l(G(s), K(s))\|_{\infty} < 1$ , we can obtain a robust controller. Thus we have reduced the robust controller design problem as an  $H_{\infty}$  problem. By setting the realization of  $W_S(s)$  as

$$W_S(s) \triangleq \begin{bmatrix} A_S & B_S \\ \hline C_S & 0 \end{bmatrix}$$



Figure 2.9: The generalized plant

the generalized plant G(s) can be written as follows

$$G(s) = \begin{bmatrix} a_{11} & a_{12} & 0 & O & 0 & -b_1 \\ a_{21} & 0 & a_{23} & O & 0 & 0 \\ 0 & 1 & 0 & O & 0 & 0 \\ O & O & B_S & A_S & B_S & O \\ \hline c_{w_1} & c_{w_2} & c_{w_3} & O & 0 & -d_w \\ 0 & 0 & 0 & C_S & 0 & 0 \\ \hline 1 & 0 & 0 & O & 0 & 0 \\ 0 & 1 & 0 & O & 0 & 0 \\ 0 & 0 & 1 & O & 1 & 0 \end{bmatrix},$$

where,  $C_w$  is partitioned as

$$C_w = \left[\begin{array}{ccc} c_{w_1} & c_{w_2} & c_{w_3} \end{array}\right]$$

By substituting each parameter into G(s) and applying an equivalent transformation by

 $T = \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}, \text{ we can derive the generalized plant as}$   $G(s) = \begin{bmatrix} -13 & 12 & 0 & 0 & 0 & -3.4 \\ 2400 & 2500 & 2500 & 0 & 0 & 0 \\ -2400 & -2500 & -2500 & 0 & 0 & 3.4 \\ 1.0 & 1.0 & 1.0 & -1.0 \times 10^{-5} & 1.0 & 0 \\ 0.098 & 0.10 & 0.099 & 0 & 0 & 6.7 \times 10^{-7} \\ 0 & 1.0 & 0 & 0 & 0 & 0 \\ 1.0 & 1.0 & 1.0 & 0 & 0 & 1.0 & 0 \end{bmatrix},$  (2.97)

#### 2.7. A NUMERICAL EXAMPLE AND DISCUSSIONS

where the weighting function  $W_S(s)$  is put as

$$W_S(s) = \frac{0.5}{s + 0.00001}.$$

It can easily be verified that G(s) satisfies the assumptions A1, A2'-Case2 and A3', hence the  $H_{\infty}$  problem is the non-standard  $H_{\infty}$  problem of case 2. Thus, by following the results given in section 2.4 we can derive the low-order  $H_{\infty}$  controller whose order is two. Here, it should be also noted that G(s) in (2.97) satisfies the assumption in (2.5). Thus, by using the result of Theorem 2.4.3, a reduced order  $H_{\infty}$  controller can be given as

$$K(s) = \begin{bmatrix} -7.9 \times 10^5 & -2.6 \times 10^6 & -7.8 \times 10^5 & -8.0 \times 10^5 & 0\\ 0 & -1.0 \times 10^{-5} & 0 & 0 & 1.0\\ \hline -2.3 \times 10^5 & -7.5 \times 10^5 & -2.3 \times 10^5 & -2.3 \times 10^5 & 0 \end{bmatrix}.$$

The order of the controller is two, and is lower than that of G(s) by second order, which is equal to the dimension of the state measurable without noise.



Figure 2.10: The closed loop system In this figure,  $P_i(s)$  indicates a perturbed plant:  $(1 + \Delta(s))P(s)$ .

In order to examine the performance of the controller, free responses from an initial state  $x_0 = \begin{bmatrix} 0 & 0 & 1.0 \times 10^{-3} \end{bmatrix}^T$  are simulated in the closed loop system of Figure 2.10. Figure 2.7.3 illustrates the plots of the free responses, where Figure 2.7.3-(a) is the response of the nominal closed loop system, and Figure 2.7.3-(b) is the response of the perturbed closed loop system. It can be verified that both of the closed loop systems are stable, and that the controller is certainly the robust controller. Thus, we have designed a low-order robust controller by using the method proposed in this chapter.

## 2.7.4 Discussion

From the example it is verified that the low-order  $H_{\infty}$  controller is certainly derived for the partial state feedback system, where the partial states are measured without disturbances.



Figure 2.11: The closed loop responses

(a) is an initial response in the closed loop system where the nominal plant P(s) is used. (b) is an initial response in the closed loop system where a perturbed plant  $(1 + \Delta(s))P(s)$  is used.

Correctness of the theoretical result that is derived in this chapter is also verified. Although, for the system of the partial state feedback, it is known that a low-order controller which bases its structure on a reduced-order observer can stabilize the system, the method of designing the low-order  $H_{\infty}$  controller, which is derived with so-called ARE approach [11, 47], has not been known except the study of Zhang and Hosoe [71]. On the other hand, as compared with the study of them, our result is more general in the point that we can treat all of the cases of the non-standard problems.

The above numerical example is merely one of instances in which the theoretical result given in this chapter is applicable. There must exist other examples in which the result is also applicable. However, it is hard to find out a non-trivial example that is applicable the result. Hence, it is useful to clarify the class of the applicable problems. The following chapters will make it clear to utilize the result given in this chapter.

## 2.8 Summary

By using the free parameters in the class of the full order controllers, this chapter has given the reduced-order controllers of the non-standard  $H_{\infty}$  control problems where  $D_{12}$  is of full row rank and  $D_{21}$  is of full column rank. The controllers are also represented with free parameters; hence, the classes of the reduced-order controllers are derived. The classes are divided into two types: reduced-order observer type and reduced-order dual observer type. The orders of the controllers are listed in Table 2.2. Since the classes of the reduced-order controllers are subclass of the non-standard controllers, the reduced-order controllers satisfy the same  $H_{\infty}$  norm specifications that the full order controllers satisfy. Thus, we have given a fundamental result for designing a low order  $H_{\infty}$  controller, and have clarified one of merits formulating as the non-standard problems. The effectiveness of this method for designing the reduced-order  $H_{\infty}$  controller will be shown with practical problems in later chapters.

Table 2.2: Order of each controller

In this table, n is the order of the generalized plant,  $p_1$  is the number of the controlled outputs,  $p_2$  is the number of the measurement outputs,  $m_1$  is the number of the disturbance inputs and  $m_2$  is the number of the control inputs.

Cases of problems	$\deg\left(K ight)$
Case 1	$n - (m_2 - p_1)$
Case 2	$n - (p_2 - m_1)$
Case 3	$ \left\{ \begin{array}{c} n - (m_2 - p_1) \\ n - (p_2 - m_1) \end{array} \right. $

## Chapter 3

# A synthesis of low-order integral-type controller

## 3.1 Introduction

This chapter considers a synthesis problem of a robust servo control system [14] where it is required that controlled outputs track step-formed references in the presence of uncertainties in a plant. It is known that in the case where the references are assumed to be of the stepformed signals, the robust servo controller requires to include integrators in its structure. Hence the problem is formulated as an integral-type  $H_{\infty}$  controller design. This problem naturally reduces to some of the non-standard  $H_{\infty}$  problems in which the non-standard  $H_{\infty}$ problem of case 2 is included.

Up to now, there are many studies concerned about the integral-type  $H_{\infty}$  controller design [23, 27, 26, 25, 49, 47, 39, 40, 73]. In these studies the problem is often solved under a transformation to the standard  $H_{\infty}$  problem. One well known approach is based on an approximation of the integrator with a stable first-order transfer function [73]. This approach enables the  $H_{\infty}$  controller to be designed easily but the controller becomes an approximated integral-type controller. Mita et al. [49, 47] and Liu et al. [39, 40] proposed the concept of pseudo-stabilization of the ARE and solved the problem based on the standard  $H_{\infty}$  control. Hara et al. [23, 27, 26, 25] proposed a method based on the transformation of the generalized plant and the re-construction of the controller. Mita, et al. also proposed a method [48] to solve the problem where the plant has poles on the  $j\omega$ -axis, and that method requires to factorize the plant into a part having the  $j\omega$ -poles and another part. Other interesting results for the robust servo controller design include the results by Zhang et al. [71] and Hozumi et al. [26] where the problem is treated as some of the non-standard  $H_{\infty}$  problems.

In this study, we treat the integral-type  $H_{\infty}$  controller design as the non-standard  $H_{\infty}$ problem where a direct feed-through term of the subsystem from the external input to the measurement output (i.e.,  $D_{21}$ ) is column full rank. Although, solving the non-standard  $H_{\infty}$ problem has some advantages, a defect such that the resultant controller becomes high-order



Figure 3.1: A closed loop system

arises. Furthermore, in the case where a plant has the  $j\omega$ -poles, a solution to the problem becomes more complex. This chapter treats the problem as the non-standard  $H_{\infty}$  problem in both of the cases where the plant has no integrators and the plant has the  $j\omega$ -poles. Thus the main topic of this study is the derivation of the low-order controller for the problem of the integral-type  $H_{\infty}$  controller design. Another topic is the treatment of the both of the cases where the plant has no integrators and the plant has integrators in a less complex manner.

## 3.2 Robust servo controller design

## 3.2.1 Specifications

Consider a feedback system shown in Figure 3.1, where K(s) is a controller and P(s) is a plant which satisfies the following assumptions:

- $P(s) \in \mathbb{C}^{p \times m}$  has no  $j\omega$ -invariant zeros,
- $m \ge p$ .

These assumptions are fundamental for the robust servo system design[14]. In Figure 3.1,  $r \in \mathbb{R}^p$  represents the reference signal,  $u \in \mathbb{R}^m$  is the signal of control input,  $y_p \in \mathbb{R}^p$  is the output from the plant and  $y \in \mathbb{R}^p$  which represents the error is the input to the controller. We denote the closed loop system in Figure 3.1 as (P(s), K(s)). Specifications of the robust servo system design are

- **S1** stabilize the closed loop system (P(s), K(s)),
- **S2** let the  $H_{\infty}$  norm of closed loop transfer functions be less than  $\gamma$ , where  $\gamma \in \mathbb{R}$  is a positive number fixed *a priori*,
- S3 in the presence of the modeling error in P(s) and step-shaped disturbances, let the control outputs asymptotically track the reference inputs which are restricted to the step-shaped signals.

The specification S1 is necessary for the closed loop system to be internally stable. Besides improving the transient responses of the outputs in the closed loop system, the specification

S2 assures the robust stability of the closed loop system in the presence of additive or multiplicative uncertainty in the plant. The specification S3 is the requirement for the system to hold a performance of the robust tracking.

## 3.2.2 Formulation with the mixed sensitivity problem

In the early many studies, for the above specifications the  $H_{\infty}$  control problem with the generalized plant described in Figure 3.2 is solved. In this figure  $W_T(s)$  and  $W_S(s)$  represent the weighting functions. This is the generalized plant of the well-known mixed sensitivity problem. Thus, by solving the  $H_{\infty}$  control problem which is formulated as

$$\left\|\begin{array}{c}T_{z_1w}(s)\\T_{z_2w}(s)\end{array}\right\|_{\infty} < \gamma,$$

where  $T_{z_iw}(s)$  denotes a transfer function from the signal w to the signal  $z_i$ , the specifications S1 and S2 are satisfied. In order to satisfy the specification S3, integrators are introduced into the weighting function  $W_S(s)$ . This is because if a condition

$$T_{z_1w}(s) \in \mathcal{RH}_{\infty} \tag{3.1}$$

is satisfied for the weight  $W_S(s)$  such that

$$W_S(s) = \frac{\tilde{W}_S(s)}{s}, \ \tilde{W}_S(s) \in \mathcal{RH}_{\infty}, \\ \tilde{W}_S(0) \neq 0,$$
(3.2)

then the  $H_{\infty}$  controller includes an integrator in the case where P(s) has no integrators, hence in this case the specification S3 is satisfied. Thus, by solving the mixed sensitivity problem with the weighting function in (3.2), the robust servo controller can be designed in the case where P(s) has no integrators. If we give minimal state space representations of P(s),  $W_S(s)$  and  $W_T(s)$  as

$$P(s) \triangleq \begin{bmatrix} A_P & B_P \\ \hline C_P & O \end{bmatrix}, A_P \in \mathbb{R}^{n_p \times n_p}$$
(3.3)

$$W_S(s) \triangleq \begin{bmatrix} A_W & B_W \\ \hline C_W & O \end{bmatrix}, A_W \in \mathbb{R}^{n_w \times n_w}$$
(3.4)

$$W_T(s)P(s) \triangleq \begin{bmatrix} A_P & B_P \\ \hline C_T & I_m \end{bmatrix},$$
(3.5)

the state space representation of the generalized plant in Figure 3.2 is written as follows:

$$G_{ms}(s) = \begin{bmatrix} A_W & -B_W C_P & B_W & O \\ O & A_P & O & B_P \\ \hline C_W & O & O & O \\ -\overline{O} & -\overline{-C_P} & -\overline{D_P} & \overline{D} \end{bmatrix}$$
$$= \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ \hline C_2 & D_{21} & O \end{bmatrix} = \begin{bmatrix} G_{11}(s) & G_{12}(s) \\ G_{21}(s) & G_{22}(s) \end{bmatrix}.$$
(3.6)



Figure 3.2: The generalized plant for mixed sensitivity problem

However, it should be noted that there are difficulties in solving the mixed sensitivity problem form two points. The first point is that the mixed sensitivity problem with the weight  $W_S(s)$  in (3.2) is the non-standard problem where the assumptions A1-2) and A3-2) are not satisfied. This is easily verified from the generalized plant in (3.6). Especially a deviation from the assumption A1-2) makes the problem hard to solve. The other point is that the above formulation is not complete in the case where P(s) has integrators. In this case, the condition (3.1) is satisfied regardless whether the controller has integrators or not, hence there is a possibility such that the solution derives an  $H_{\infty}$  controller which has no integrators. The control system, where the controller has no model of the reference input or the disturbance input, is not robust in the presence of the uncertainty of the plant. This leads to a new formulation for the integral-type  $H_{\infty}$  controller design, where the problem is considered in the two cases where 1) P(s) has no  $j\omega$ -poles, and 2) P(s) has  $j\omega$ -poles.

## **3.3** In the case P(s) has no $j\omega$ -poles

### 3.3.1 Formulation

In this section, it is assumed that the original plant of P(s) has no poles on the  $j\omega$ -axis. In the first place, let us review the generalized plant of Figure 3.2. In the generalized plant, the state of the weighting function  $W_S(s)$ , which is unstable transfer function, is not measurable from the output y, hence the system is not detectable and the assumption A1-2) is not satisfied. Also in this generalized plant, since the dimensions of the signals w and y are equal, the invariant zeros of the system from w to y include the mode of  $W_S(s)$  (See appendix A.1.), hence the assumption A3-2) is also not satisfied. Therefore, the  $H_{\infty}$  control problem for the generalized plant of Figure 3.2 is the non-standard  $H_{\infty}$  control problem where the assumptions A1-2) and A3-2) are not satisfied.

On the other hand, let us consider a generalized plant in Figure 3.3. In the generalized plant the weighting function  $W_S(s)$  is also selected such that (3.2) is satisfied. A difference in the generalized plants of Figure 3.2 and Figure 3.3 is that the observed outputs have increased in the case of Figure 3.3, where we assume that the state of  $W_S(s)$  is observed from an output



Figure 3.3: The generalized plant for mixed sensitivity problem formulated as the non-standard problem

 $y_1$ . Assume that  $P(s), W_S(s)$  and  $W_T(s)$  have the same state space representations in (3.3) to (3.5), and that  $W_I(s)$  has the following stabilizable and detectable state space realization

$$W_I(s) \triangleq \begin{bmatrix} A_W & B_W \\ I_{n_w} & O \end{bmatrix}, \qquad (3.7)$$

where  $W_I(s)$  is a strictly proper rational transfer function such that the poles are located only on the domain  $\{s : \operatorname{Re}(s) \leq 0\}$ . For simplicity  $\tilde{W}_s(s) \in \mathcal{RH}_\infty$  is selected as a constant  $C_W$ . Then the state space realization of the generalized plant G(s) is represented as follows

$$G(s) = \begin{bmatrix} A_W & -B_W C_P & B_W & O \\ O & A_P & O & B_P \\ \hline C_W & O & O & O \\ O & C_T & O & I_m \\ \hline I_{n_w} & O & O & O \\ O & -C_P & I_p & O \end{bmatrix}$$
$$= \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix} = \begin{bmatrix} G_{11}(s) & G_{12}(s) \\ G_{21}(s) & G_{22}(s) \end{bmatrix}.$$
(3.8)

In this generalized plant, since  $D_{21}$  is a matrix of full column rank, the assumption **A2-2**) is not satisfied. On the other hand, the assumption **A1-2**), i.e., the detectability condition of the pair  $(A, C_2)$  is satisfied since the sates of  $W_S(s)$  are measurable through the output  $y_1$ . Furthermore, since  $D_{21}$  is of full column rank, the invariant zeros of  $G_{21}(s)$  are unobservable poles of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$ , which include the unobservable poles of the pair  $(A, C_2)$ .\* All the modes of  $W_S(s)$  are included in the observable subspace of the pair  $(A, C_2)$ , hence the unstable modes of  $W_S(s)$  are not included in the invariant zeros of  $G_{21}(s)$ . In fact, from the expression such that

$$\begin{bmatrix} -A - B_1 D_{21}^{\dagger} C_2 \\ -\overline{D}_{21}^{-} \overline{C}_2^{-} \end{bmatrix} = \begin{bmatrix} A_W & O \\ O \\ -\overline{I}_{n_w} - \overline{O} \end{bmatrix}$$
(3.9)

<sup>\*</sup>If  $\lambda_i$  is an unobservable pole of the pair  $(A, C_2)$ ,  $\exists v_i \neq O$ ;  $Av_i = \lambda_i v_i$ ,  $C_2 v_i = O$ . Then  $\lambda_i$  and  $v_i$  satisfy  $\left(A - B_1 D_{21}^{\dagger} C_2\right) v_i = \lambda_i v_i$ ,  $D_{21}^{\perp} C_2 v_i = O$ . Thus  $\lambda_i$  is an unobservable pole of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$ .



Figure 3.4: The generalized plant and the integral-type  $H_{\infty}$  controller

it is apparently verified that the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is the mode of  $W_S(s)$ . On the other hand, it is also verified that the unobservable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is the mode of P(s). From the assumption such that P(s) has no  $j\omega$ poles, the assumption **A3-2**) is satisfied. Thus, the  $H_{\infty}$  control problem for the generalized plant of Figure 3.4 is the non-standard  $H_{\infty}$  problem where the assumption **A2-2**) is not satisfied.

## 3.3.2 A high-order controller design

Thus the problem of the robust servo system design is reduced to the non-standard  $H_{\infty}$  problem of case 2. Hence, by following its solution which has already been given in chapter 2, we can obtain an  $H_{\infty}$  controller  $K_{\infty}(s)$ . Then as shown in Figure 3.4, by combining a part of the weight  $W_S(s)$  and  $K_{\infty}(s)$ , an integral-type  $H_{\infty}$  controller  $\hat{K}_I(s)$  is constructed as follows.

$$\hat{K}_{I}(s) = K_{\infty}(s) \begin{bmatrix} W_{I}(s) \\ I_{p} \end{bmatrix}$$
(3.10)

Certainly,  $\hat{K}_I(s)$  is a controller which satisfies specifications S1, S2 and S3. However, it should be noted that the order of the controller is higher than that of the generalized plant G(s) by the order of  $W_I(s)$ . Thus, this approach yields a high-order controller, and this is a defect of this approach.

#### 3.3.3 A low-order controller design

This section aims at deriving the integral-type  $H_{\infty}$  controller of low order. This section first derives the controller  $K_{\infty}(s)$  which is the  $H_{\infty}$  controller of the non-standard  $H_{\infty}$  control problem for G(s) in (3.8). Since G(s) fits the generalized plant of the non-standard  $H_{\infty}$ problem of case 2, whose solution has already been discussed in the previous section 2.4. By using the solution we can solve the problem as follows.

Let us define  $L_H$  such that the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is stabilized with  $L_H$ , that is,

$$L_{H} \triangleq \begin{bmatrix} L_{H_{1}} \\ O \end{bmatrix} \in \mathbb{R}^{(n_{w}+n_{p})\times n_{w}}, \ \forall L_{H_{1}} \in \{L_{H_{1}} : \operatorname{Re}(\lambda_{i}(A_{W}+L_{H_{1}})) < 0, \forall i\}$$

Also, let us define  $A_{ZH}$  as follows

$$A_{ZH} = A - B_1 D_{21}^{\dagger} C_2 + L_H D_{21}^{\perp} C_2.$$
(3.11)

Lemma 3.3.1 If the AREs:

$$X\left(A - B_2 D_{12}^{\dagger} C_1\right) + \left(A - B_2 D_{12}^{\dagger} C_1\right)^T X + X\left\{B_1 B_1^T - B_2 D_{12}^{\dagger} \left(B_2 D_{12}^{\dagger}\right)^T\right\} X + \left(D_{12}^{\perp} C_1\right)^T D_{12}^{\perp} C_1 = O$$
(3.12)

and

$$YA_{ZH}^{T} + A_{ZH}Y + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}Y = O$$
(3.13)

have stabilizing solutions  $X \ge O$  and  $Y \ge O$  which satisfy  $\rho(XY) < 1$ , the class of the  $H_{\infty}$  controllers for G(s) in (3.8) is represented as:

$$\mathcal{K}_{\infty} = \{ K_{\infty}(s) : N(s) \in \mathcal{BH}_{\infty}, W(s) \in \mathcal{RH}_{\infty} \}$$

where N(s) and W(s) are free parameters, and  $K_{\infty}(s)$  is represented as follows

$$K_{\infty}(s) = \mathcal{F}_{l} \left( \begin{bmatrix} A_{Y} + \hat{B}_{2}F_{\infty} & H_{\infty} & -\hat{B}_{2} \\ \hline -\frac{-F_{\infty}}{D_{21}^{\dagger}\hat{C}_{2}Z} & O & I_{m} \\ \hline D_{21}^{\dagger}\hat{C}_{2}Z & D_{21}^{\dagger} & O \\ D_{21}^{\perp}C_{2} & D_{21}^{\perp} & O \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix} \right),$$
(3.14)

where

$$\begin{split} A_Y &= A + Y C_1^T C_1 + H_{\infty} C_2 \\ \hat{B}_2 &= B_2 + Y C_1^T D_{12} \\ \hat{C}_2 &= D_{21} B_1^T X + C_2 \\ F_{\infty} &= \left\{ -D_{12}^{\dagger} C_1 - D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T X \right\} Z \\ H_{\infty} &= -B_1 D_{21}^{\dagger} - Y \left( D_{21}^{\dagger} C_2 \right)^T D_{21}^{\dagger} + L_H D_{21}^{\perp} \\ Z &= (I - YX)^{-1} \,. \end{split}$$

**Proof.** From Theorem 2.4.2, the result is immediately derived.

As we have considered in chapter 2, it is possible to reduce the order of the  $H_{\infty}$  controller derived above. Thus we derive the reduced-order  $H_{\infty}$  controller and obtain the integral-type, reduced-order  $H_{\infty}$  controller. The following lemma is useful for the derivation.

**Lemma 3.3.2** Assume that the ARE in (3.13) has a positive semi-definite stabilizing solution. Then the stabilizing solution of the ARE can be represented as follows:

$$Y = \begin{bmatrix} O & O \\ O & Y_r \end{bmatrix} \in \mathbb{R}^{(n_p + n_w) \times (n_p + n_w)}, \tag{3.15}$$

where  $Y_r \ge O$  is a stabilizing solution of the reduced-order ARE:

$$Y_r A_P^T + A_P Y_r + Y_r \left( C_T^T C_T - C_P^T C_P \right) Y_r = O.$$
(3.16)

**Proof.** The ARE in (3.13) can be represented as follows

$$YA_Y^T + A_{ZH}Y = O, (3.17)$$

where  $A_Y$  is denoted as

$$A_Y = A_{ZH} + Y \left\{ C_1^T C_1 - \left( D_{21}^{\dagger} C_2 \right)^T D_{21}^{\dagger} C_2 \right\}$$

and is a stable matrix. Let U be a row-basis of the stable subspace of  $A_{ZH}$ . Then the matrix U can be selected as  $U = \begin{bmatrix} I_{n_w} & O \end{bmatrix}$ . Then the matrix U satisfies the following equation:

$$UA_{ZH} = (A_W + L_{H_1}) U.$$

Pre-multiplying the equation (3.17) by U, the following equation is obtained

$$(UY)A_Y^T + (A_W + L_{H_1})(UY) = O.$$

Since  $\operatorname{Re}(\lambda_i(A_Y)) < 0$  and  $\operatorname{Re}(\lambda_i(A_W + L_{H_1})) < 0$  hold, the solution Y satisfies UY = O(See appendix A.3.1) and Y must be in the form (3.15). By substituting  $Y \ge O$  in (3.15) into the ARE in (3.13),  $Y_r \ge O$  must be a solution of the reduced-order ARE in (3.16).

In the second place, let us represent the full-order  $H_{\infty}$  controller given in (3.14) with DHMT as follows:

$$K_{\infty}(s) = \mathcal{DHM}\left(\begin{bmatrix}\frac{A_{Y}}{-F_{\infty}} & \hat{B}_{2} & H_{\infty}\\ \hline -F_{\infty} & I_{m} & O\\ \hline D_{21}^{\dagger} \hat{C}_{2} \bar{Z} & O & D_{21}^{\dagger}\\ D_{21}^{\dagger} C_{2} & O & D_{21}^{\dagger} \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix}\right)$$
$$= \begin{bmatrix}\frac{A_{Y}}{C_{K}(s)} & \hat{B}_{2}\\ \hline -C_{K}(s) & I_{m}\end{bmatrix}^{-1} \begin{bmatrix}\frac{A_{Y}}{C_{K}(s)} & H_{\infty}\\ \hline -C_{K}(s) & N(s)D_{21}^{\dagger} + W(s)D_{21}^{\dagger}\end{bmatrix},$$

where  $C_K(s)$  is defined as follows.

$$C_K(s) \triangleq -F_{\infty} + N(s)D_{21}^{\dagger}\hat{C}_2 Z + W(s)D_{21}^{\perp}C_2.$$
(3.18)

#### 3.3. IN THE CASE P(S) HAS NO $J\omega$ -POLES

By using the result of Lemma 3.3.2 the matrix  $A_Y$  can be represented as follows

$$A_Y = \begin{bmatrix} A_W + L_{H_1} & O \\ O & A_P + Y_r \left( C_T^T C_T - C_P^T C_P \right) \end{bmatrix}$$

where  $Y_r \geq O$  is a stabilizing solution of the reduced-order ARE in (3.16). In the equation (3.18),  $C_K(s)$  is explicitly represented with the free parameter  $W(s) \in \mathcal{RH}_{\infty}$ . Hence an adequate selection of the free-parameter W(s) yields pole-zero cancellations in the controller, and the order of the controller can be reduced.

**Theorem 3.3.1** Under the same solvability condition as in Lemma 3.3.1, the class of reduced order  $H_{\infty}$  controllers is parametrized as follows:

$$K_{\infty}^{r}(s) = \mathcal{DHM}\left(\begin{bmatrix} \tilde{A}_{Y} & \hat{B}_{P} & H_{\infty_{2}} \\ \hline -F_{\infty_{2}} & I_{m} & \hat{F}_{\infty_{1}}D_{21}^{\perp} \\ C_{D_{2}} & O & D_{21}^{\dagger} - \hat{C}_{D_{1}}D_{21}^{\perp} \end{bmatrix}, N(s)\right),$$
(3.19)

where  $N(s) \in \mathcal{BH}_{\infty}$  is a free parameter, and  $\tilde{A}_Y, \hat{B}_P, H_{\infty_2}, \hat{F}_{\infty_1}, F_{\infty_2}, \hat{C}_{D_1}, C_{D_2}$  are defined as follows:

$$\begin{split} \tilde{A}_Y &= A_P + Y_r \left( C_T^T C_T - C_P^T C_P \right) \\ \hat{B}_P &= B_P + Y_r C_T^T \\ H_{\infty_2} &= \begin{bmatrix} O & Y_r C_P^T \end{bmatrix} \\ F_{\infty} &= \begin{bmatrix} F_{\infty_1} & F_{\infty_2} \end{bmatrix}, \quad F_{\infty_1} \in \mathbb{R}^{m \times n_w}, F_{\infty_2} \in \mathbb{R}^{m \times n_p} \\ D_{21}^{\dagger} \hat{C}_2 Z &= \begin{bmatrix} C_{D_1} & C_{D_2} \end{bmatrix}, \quad C_{D_1} \in \mathbb{R}^{n_w \times n_w}, C_{D_2} \in \mathbb{R}^{n_w \times n_p}. \end{split}$$

**Proof.** By setting the free parameter as

$$W(s) = -\left(F_{\infty} + N(s)D_{21}^{\dagger}\hat{C}_{2}Z\right)V,$$

where V is a full column rank matrix defined as

$$V \triangleq \left[ \begin{array}{c} I_{n_w} \\ O \end{array} \right] \in \mathbb{R}^{(n_w + n_p) \times n_w},$$

the matrices  $A_Y$  and  $C_K(s)$  satisfy

$$\begin{cases} A_Y V = V \left( A_W + L_{H_1} \right) \\ C_K(s) V = O \end{cases}$$

Hence the pair  $(A_Y, C_K(s))$  is unobservable, and the order of the controller is reduced by the dimension of rank  $(V) = n_w$ . Then the reduced order  $H_\infty$  controller is derived.

**Remark 3.3.1** The McMillan degree of the central solution of the  $H_{\infty}$  controller in (3.14) is reduced by the dimension of rank  $(V) = n_w$ , and is equal to  $n_p$ .



Figure 3.5: The generalized plant and the integral-type, reduced-order  $H_{\infty}$  controller

**Remark 3.3.2** It should be also noted that the class of the reduced-order  $H_{\infty}$  controllers is represented with free parameter  $N(s) \in \mathcal{BH}_{\infty}$ .

**Remark 3.3.3** In this theorem the reduced order TDF controller is represented with DHMT, and this implies that the controller is in a form of coprime factorization over  $\mathcal{RH}_{\infty}$ . This form is useful for reducing the order of the controller by approximation, because it is hard to apply approximation methods for unstable systems.

From the above theorem, a reduced-order controller  $K_{\infty}^{r}(s)$  is derived. Then as shown in Figure 3.5, by combining a part of the weight  $W_{S}(s)$  and  $K_{\infty}^{r}(s)$ , an integral-type, reduced-order  $H_{\infty}$  controller is constructed as follows.

$$K_{\infty}^{int,r}(s) = K_{\infty}^{r}(s) \begin{bmatrix} W_{I}(s) \\ I_{p} \end{bmatrix}$$
(3.20)

From Figure 3.5, it can be verified that the controller  $K_{\infty}^{int,r}(s)$  satisfies the specifications S1, S2 and S3. The controller is parametrized with a free parameter  $N(s) \in \mathcal{BH}_{\infty}$ . The order of the central solution is  $n_w + n_p$ , which is lower than that of the controller  $\hat{K}_I(s)$  in (3.10). Thus we can derive an integral-type, reduced-order  $H_{\infty}$  controller by way of reconstruction of the controller.

## 3.3.4 A direct derivation of an integral-type $H_{\infty}$ controller

In the previous section, the integral-type, reduced-order  $H_{\infty}$  controller is derived indirectly. This section aims at deriving the integral-type, reduced-order  $H_{\infty}$  controller directly. Firstly, we investigate modes of the full-order  $H_{\infty}$  controller in (3.14). The following definition is useful for the investigation.

**Definition 3.3.1 (Reduction mode of ARE)** Assume that the ARE in (3.13) has a positive semi-definite stabilizing solution Y. If there exists a full row rank matrix U which satisfies the equations:

$$UA_{ZH} = \Lambda U \tag{3.21}$$

$$UY = O, (3.22)$$

then we call the matrix  $\Lambda$  reduction mode of the ARE and U eigen-subspace of the mode.

**Remark 3.3.4** In the standard  $H_{\infty}$  problem, where  $D_{21}$  is of full row rank, the solution is given by solving an ARE:

$$Y\left(A - B_{1}D_{21}^{\dagger}C_{2}\right)^{T} + \left(A - B_{1}D_{21}^{\dagger}C_{2}\right)Y + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}Y + B_{1}D_{21}^{\perp}\left(B_{1}D_{21}^{\perp}\right)^{T} = O.$$
 (3.23)

In this case, the reduction mode of the ARE in (3.23) is restricted to the mode of the stable invariant zeros of  $G_{21}(s)$ , that is, it is restricted to the stable uncontrollable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, B_1 D_{21}^{\perp})$ . This is stated in Lemma A.10.1. On the other hand, in the non-standard  $H_{\infty}$  problem of case 2, not only the mode of the stable invariant zeros of  $G_{21}(s)$ but also the stable mode which is obtained by stabilizing the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  are included in the reduction mode of the ARE. This can be verified from the discussion in section 2.4: for a full row rank matrix U the equations (2.17) and (2.19) hold, where the stable matrix  $\Lambda$  in (2.18) includes the stabilized mode and the stable zero mode. In the case of the non-standard  $H_{\infty}$  problem for the generalized plant in (3.8), the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is stabilized with  $L_{H_1}$  such that  $\Lambda_1 \triangleq A_W + L_{H_1}$  is stable, and  $\Lambda_1$  is included in the reduction mode of the ARE in addition to the mode of the stable invariant zeros of  $G_{21}(s)$ .

The following lemma clarifies the relationship between the reduction mode of the ARE and the structure of the non-standard  $H_{\infty}$  controller.

**Lemma 3.3.3** Assume that the non-standard  $H_{\infty}$  controller  $K_{\infty}(s) \in \mathcal{K}_{\infty}$  in (3.14) is minimal realized. Then if the full row rank matrix U, which is an eigen-subspace of the reduction mode of the ARE in (3.13), satisfies

$$UB_2 = O, (3.24)$$

the eigenvalues of  $\Lambda_1 = A_W + L_{H_1}$  are included in  $K_{\infty}(s) \in \mathcal{K}_{\infty}2$  as its real modes.

**Proof.** Since a full row rank matrix  $U \triangleq \begin{bmatrix} I_{n_w} & O \end{bmatrix}$  satisfies the equations (3.21) and (3.22) for  $\Lambda = \Lambda_1$ , the matrix U holds the equation:

$$UA_Y = \Lambda_1 U,$$

where  $A_Y$  is represented as

$$A_Y = A_{ZH} + Y \left\{ C_1^T C_1 - \left( D_{21}^{\dagger} C_2 \right)^T D_{21}^{\dagger} C_2 \right\}, \qquad (3.25)$$

hence  $\Lambda_1$  is a mode of  $A_Y$ . If the matrix U satisfies (3.24), the so-called A-matrix of the  $H_{\infty}$  controller in (3.14)

$$A_{K_{\infty}} \triangleq A_{Y} + \left(B_{2} + YC_{1}^{T}D_{12}\right)F_{\infty}$$

7

satisfies

$$UA_{K_{\infty}} = \Lambda_1 U,$$

where U is a matrix of full row rank. Hence, if  $K_{\infty}(s)$  is a minimal form,  $\Lambda$  is a real mode of  $K_{\infty}(s)$ . This implies that the eigenvalues of  $\Lambda_1$  are included as the poles of the  $H_{\infty}$  controller.

**Remark 3.3.5** Since in the generalized plant G(s) in (3.8) the matrix U satisfies (3.24), from the Lemma 3.3.3 it follows that if  $A_W$  is a stable matrix, that is,  $W_S(s)$  is a stable transfer function,  $L_{H_1}$  can be selected as zero and the mode of  $W_S(s)$  is included in the modes of the controller. Hence the eigenvalues of  $W_S(s)$  are included in  $K_{\infty}(s)$  as its eigenvalues if we choose  $W_S(s)$  as a stable transfer function, or  $W_S(s)$  is regarded as a stable transfer function.

Based on the above idea, we intend to introduce the  $j\omega$ -eigenvalues of  $W_S(s)$  into the controller. For this purpose the concept of pseudo-stabilization is adopted. The concept is firstly proposed [49] in the problem of the standard  $H_{\infty}$  control.

This study extends the pseudo-stabilizing solution of the ARE in the case of non-standard  $H_{\infty}$  problem of case 2.

**Definition 3.3.2** A pseudo-stabilizing solution of the ARE in (3.13) is defined as a solution that satisfies both of the following items.

1. for a full row rank matrix U, the following equations are satisfied

$$UA_{ZH} = \Lambda U, \quad \operatorname{Re}\left(\lambda_i(\Lambda)\right) \le 0, \forall i$$
  
 $UY = O,$ 

where  $A_{ZH}$  is the matrix which is defined in (3.11) and Y is a positive semi-definite solution of the ARE in (3.13).

#### 3.3. IN THE CASE P(S) HAS NO $J\omega$ -POLES

2. real parts of all the eigen values of  $A_Y$  in (3.25) are negative except the eigenvalues of  $\Lambda$ 

**Remark 3.3.6** In the standard  $H_{\infty}$  problem, the pseudo-stabilizing solution of ARE is interpreted as the stabilizing solution of the ARE where  $j\omega$ -invariant zeros of  $G_{21}(s)$  are regarded as stable zeros. On the other hand, in the non-standard  $H_{\infty}$  problem of case 2, the pseudostabilizing solution of ARE is interpreted as the stabilizing solution of the ARE where not only the  $j\omega$ -invariant zeros of  $G_{21}(s)$  but also  $j\omega$ -mode which is included in the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  are regarded as stable modes.

The following theorem enables us to give a pseudo-stabilizing solution of the ARE in (3.13) with solving the stabilizing solution of the ARE.

**Theorem 3.3.2** Let us consider the ARE in (3.13). The stabilizing solution  $Y \ge O$ , which is obtained with the observable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is stabilized, equals the pseudo-stabilizing solution  $\tilde{Y} \ge O$  which is obtained by letting  $L_H = O$ .

**Proof.** From (3.9), the full row rank matrix U can be chosen as

$$U = \begin{bmatrix} I_{n_w} & O \end{bmatrix}. \tag{3.26}$$

The stabilizing solution  $Y \ge O$  and the pseudo stabilizing solution  $\tilde{Y} \ge O$  of the ARE in (3.13) are the symmetric matrices, and they satisfy UY = O and  $U\tilde{Y} = O$ , hence both of the solutions must be of the form:

$$Y = \begin{bmatrix} O & O \\ O & Y_r \end{bmatrix}, \tag{3.27}$$

$$\tilde{Y} = \begin{bmatrix} O & O \\ O & \tilde{Y}_r \end{bmatrix}.$$
(3.28)

Substituting the candidate of the stabilizing solution Y in (3.27) into the ARE in (3.13), a reduced-order ARE :

$$Y_r A_P^T + A_P Y_r + Y_r \left( C_T^T C_T - C_P^T C_P \right) Y_r = 0, (3.29)$$

is derived where  $Y_r \ge O$  needs to be a stabilizing solution of the reduced-order ARE. On the other hand, substituting the candidate of the pseudo-stabilizing solution  $\tilde{Y}$  in (3.28) into the ARE in (3.13), a reduced-order ARE :

$$\tilde{Y}_r A_P^T + A_P \tilde{Y}_r + \tilde{Y}_r \left( C_T^T C_T - C_P^T C_P \right) \tilde{Y}_r = O$$
(3.30)

is derived. From the statement in Definition 3.3.2, it is verified that  $\tilde{Y}_r \ge O$  needs to be a stabilizing solution of the reduced-order ARE. Thus, it is shown that  $Y_r = \tilde{Y}_r$  and that the stabilizing solution Y in (3.27) equals the pseudo-stabilizing solution  $\tilde{Y}$  in (3.28).

**Remark 3.3.7** The above theorem also states that the pseudo-stabilizing solution of the ARE in (3.13) can be given by solving the stabilizing solution of the reduced-order ARE in (3.30).

The next theorem shows that by using the pseudo-stabilizing solution, a solution to the non-standard  $H_{\infty}$  problem is given.

**Theorem 3.3.3** Assume that the ARE in (3.12) has a stabilizing solution  $X \ge O$  and that the ARE in (3.13) has a pseudo-stabilizing solution  $Y \ge O$  and they satisfy an inequality

$$\rho(XY) < 1. \tag{3.31}$$

Then, the class of the controllers  $\{K_{\infty}(s) : N(s) \in \mathcal{BH}_{\infty}\}$ :

$$K_{\infty}(s) = \mathcal{F}_{l} \left( \begin{bmatrix} A_{K_{\infty}} & H_{\infty} & -\hat{B}_{2} \\ -F_{\infty} & O & I_{m} \\ D_{21}^{\dagger}\hat{C}_{2}Z & D_{21}^{\dagger} & O \\ D_{21}^{\dagger}\hat{C}_{2} & D_{21}^{\dagger} & O \end{bmatrix}, \begin{bmatrix} N(s) & 0 \end{bmatrix} \right),$$
(3.32)

where  $N(s) \in \mathcal{BH}_{\infty}$  is a free parameter and the other parameters are defined as

$$\begin{aligned} A_{K_{\infty}} &= A + YC_{1}^{T}C_{1} + \hat{B}_{2}F_{\infty} + H_{\infty}C_{2} \\ \hat{B}_{2} &= YC_{1}^{T}D_{12} + B_{2} \\ \hat{C}_{2} &= C_{2} + D_{21}B_{1}^{T}X \\ F_{\infty} &= \left(-D_{12}^{\dagger}C_{1} - D_{12}^{\dagger}\left(B_{2}D_{12}^{\dagger}\right)^{T}X\right)Z \\ H_{\infty} &= -B_{1}D_{21}^{\dagger} + L_{H}D_{21}^{\perp} - Y\left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger} \\ Z &= (I - YX)^{-1}, \end{aligned}$$

give a subclass of the  $H_{\infty}$  controllers for G(s) in (3.8).

**Proof.** We show that the controller given by the equation (3.14), where  $X \ge O$  is the stabilizing solution for ARE in (3.12) and  $Y \ge O$  is the pseudo-stabilizing solution for ARE (3.13), is the  $H_{\infty}$  controller when W(s) = 0. As shown in Figure 3.6 the controller in (3.14) has a double-input/single-output structure and this can be represented as follows

$$K_{\infty}(s) = \begin{bmatrix} K_1(s) & K_2(s) \end{bmatrix}, K_1(s) \in \mathbb{C}^{m \times n_w}, K_2(s) \in \mathbb{C}^{m \times p}.$$
(3.33)

Then the control input u is represented as

$$u = \begin{bmatrix} K_1(s) & K_2(s) \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \end{bmatrix},$$

where,  $y_1$  is an input signal to the controller and this is yielded from  $W_S(s)$ , which is a mode of the observable subspace of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$ . By letting  $L_{H_1} = O$ , the following equation

$$H_{\infty} \begin{bmatrix} I_{n_w} \\ O \end{bmatrix} = O \tag{3.34}$$



Figure 3.6: Closed loop system

holds. Let us denote the controller in (3.14) as

$$K_{\infty}(s) = \mathcal{F}_l\left(\left[\begin{array}{cc} K_{11}(s) & K_{12}(s) \\ K_{21}(s) & K_{22}(s) \end{array}\right], \left[\begin{array}{cc} N(s) & W(s) \end{array}\right]\right),$$

where  $N(s) \in \mathcal{BH}_{\infty}$  and  $W(s) \in \mathcal{RH}_{\infty}$  are free parameters. Then the equations

$$K_{11}(s) \begin{bmatrix} I_{n_w} \\ O \end{bmatrix} = 0, \quad K_{21}(s) \begin{bmatrix} I_{n_w} \\ O \end{bmatrix} = \begin{bmatrix} O \\ I_{n_w} \end{bmatrix}$$
(3.35)

are satisfied. Therefore, the controller  $K_1(s)$  can be represented as follows

$$\begin{split} K_{1}(s) &= K_{\infty}(s) \begin{bmatrix} I_{n_{w}} \\ O \end{bmatrix} \\ &= \mathcal{F}_{l} \left( \begin{bmatrix} A_{K_{\infty}} & O & -\hat{B}_{2} \\ \hline -F_{\infty} & O & -I & I_{m} \\ \hline D_{21}^{\dagger} \hat{C}_{2} Z & O & -I & O \\ D_{21}^{\dagger} \hat{C}_{2} & I_{n_{w}} & O \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix} \right) \end{split}$$

Hence by letting W(s) = 0,  $K_1(s)$  becomes  $K_1(s) = 0$ . Then,  $K_2(s)$  becomes

$$K_{2}(s) = K_{\infty}(s) \begin{bmatrix} O \\ I_{p} \end{bmatrix}$$

$$= \mathcal{F}_{l} \left( \begin{bmatrix} \frac{A_{K_{\infty}}}{-F_{\infty}} & H_{\infty}^{O} & -\hat{B}_{2} \\ \hline -F_{\infty} & O & I_{m} \\ \tilde{D}_{21}^{\dagger}\tilde{C}_{2}Z & I_{p} & O \end{bmatrix}, N(s) \right)$$

$$= \mathcal{F}_{l} \left( \begin{bmatrix} \frac{\hat{A}_{K_{\infty}}}{-F_{\infty}^{0}} & O & I_{m} \\ \hline D_{21}^{\dagger}\tilde{C}_{2} & I_{p} & O \end{bmatrix}, N(s) \right), \qquad (3.36)$$

where

$$\hat{A}_{K_{\infty}} = A + B_1 B_1^T X + B_2 F_{\infty}^0 + Z H_{\infty}^0 \tilde{C}_2$$
$$F_{\infty}^0 = -D_{12}^{\dagger} C_1 - B_2^T X$$
$$H_{\infty}^0 = -B_1 - Y \left( \tilde{D}_{21}^{\dagger} \tilde{C}_2 \right)^T$$

On the other hand, by using Lemma A.11.2, it can be verified that the controller in (3.36) equals the  $H_{\infty}$  controller of the standard  $H_{\infty}$  problem for  $G_{ms}(s)$  in (3.6).

**Remark 3.3.8** The order of the  $H_{\infty}$  controller in (3.36) equals the order of the generalized plant. It is lower than the order of the  $H_{\infty}$  controller in (3.10).

Thus, a design step of the robust servo controller has been given.

**Proposition 3.3.1** Assume that P(s) has no integrators. The following Design Procedure 1 yields an integral-type  $H_{\infty}$  controller which includes the same number of integrators which are included in  $W_S(s)$  and the controller satisfies the specifications S1, S2 and S3.

#### Design Procedure 1

**STEP 1** Introduce integrators into  $W_S(s)$  as follows:

$$W_S(s) = \frac{W_S(s)}{s^n}, \ \tilde{W}_S(s) \in \mathcal{RH}_{\infty}, \tilde{W}_S(0) \neq 0,$$

where n is the number of integrators which are needed to be included in the controller.

- **STEP 2** Select an  $L_H$  such that  $L_{H_1} \in \{L_{H_1} : \operatorname{Re}(\lambda_i(A_W + L_{H_1})) < 0\}$  and obtain the positive semi-definite stabilizing solutions of the AREs in (3.12) and (3.13). Then, verify the inequality (3.31).
- **STEP 3** If the inequality is satisfied, set  $L_{H_1} = O$  and the controller (3.32) is a solution.

**Proof.** Since it is apparent that the resultant controller is the  $H_{\infty}$  controller from the above discussion, it suffice to show that the controller has integrators. The full row rank matrix U, which satisfies (3.21) and (3.22), can be given as (3.26), hence, from the generalized plant G(s) represented in (3.8), it can be verified that U satisfies (3.24). Using the result of Lemma 3.3.3, it is shown that the controller (3.32) has the integrators, the number of which is same as the number of  $j\omega$ -eigenvalues of  $\Lambda = A_W$ .

**Remark 3.3.9** It should be noted that, in Design Procedure 1, it is not needed to solve the pseudo-stabilizing solution of the ARE in (3.13).

**Remark 3.3.10** The order of the resultant controller is equal to that of the generalized plant. Thus, as compared with the controller in (3.10), the order of the controller is reduced by the order of  $W_S(s)$ .

## **3.4** In the case P(s) has $j\omega$ -poles

## **3.4.1** The $j\omega$ invariant zeros of $G_{21}(s)$

Let us consider the problem of the robust servo controller design where P(s) has poles on the  $j\omega$ -axis. We first show that the  $j\omega$ -poles of P(s) appear as the invariant zeros of the subsystem from the external input to the measurement output in the generalized plants of Figures 3.2 and 3.3.

First, let us review the generalized plant  $G_{ms}(s)$  in Figure 3.2. Since the invariant zeros of  $\tilde{G}_{21}(s)$  are all the poles of the matrix

$$A - B_1 \tilde{D}_{21}^{-1} \tilde{C}_2 = \left[ \begin{array}{cc} A_W & O\\ O & A_P \end{array} \right]$$

, uncontrollable poles of the pair  $(A, B_1)$ , i. e., all the modes of P(s) in which the mode on the  $j\omega$ -axis is included, are included in the zeros of  $\tilde{G}_{21}(s)$ , hence, in this case, the generalized plant  $G_{ms}(s)$  doesn't satisfy the assumption **A3-2**).

In the case of the generalized plant G(s) in Figure 3.3, since the invariant zeros of  $G_{21}(s)$  are included in the unobservable mode of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$ , from the equation (3.9), it is apparently verified that the  $j\omega$ -poles of P(s) are included in the zeros of  $G_{21}(s)$ . Hence, in the case of the generalized plant of G(s), the assumption **A3-2**) is not satisfied.

#### 3.4.2 Design

Thus, in the case where P(s) has  $j\omega$ -poles, the  $H_{\infty}$  control problems for both of the generalized plants  $G_{ms}(s)$  and G(s) are the non-standard  $H_{\infty}$  problems in which the assumption **A3-2**) is not satisfied. For the generalized plant of  $G_{ms}(s)$ , a solution to the  $H_{\infty}$  problem is proposed in a paper [48], where the non-standard  $H_{\infty}$  problem for the generalized plant  $G_{ms}(s)$  is transformed to a standard  $H_{\infty}$  problem by separating a  $j\omega$  part from P(s). Hence the solution needs a transformation in the generalized plant.

The present study proposes a direct solution to the non-standard  $H_{\infty}$  problem for the generalized plant of  $G_{ms}(s)$ , and generalizes the solution to the non-standard  $H_{\infty}$  problem for the generalized plant of G(s). Firstly, the following lemma gives a direct solution to the non-standard  $H_{\infty}$  problem for the generalized plant of  $G_{ms}(s)$  where P(s) has  $j\omega$ -poles.

**Lemma 3.4.1** It is assumed that the strictly proper transfer function P(s), which has  $j\omega$ poles can be factorized as follows

$$P(s) = \hat{P}(s)a(s), \tag{3.37}$$

where

$$\hat{P}(j\omega) < \infty, \forall \omega \in \mathbb{R}, \quad a^{-1}(s) \in \mathcal{RH}_{\infty}.$$



Figure 3.7: A transformation in the closed loop

In the generalized plant of Figure 3.2, let us select  $W_S(s)$  as follows

$$W_S(s) = a(s)W_s(s),$$
 (3.38)

where

$$\hat{W}(\infty) = 0, \quad \hat{W}(s) < \infty, \forall s \in \{s : \operatorname{Re}(s) > 0\}.$$

Then, if the ARE in (3.12) has a stabilizing solution  $X \ge O$ , the ARE in (3.13), where  $\tilde{C}_2$  and  $\tilde{D}_{21}^{\dagger}$  are substituted for  $C_2$  and  $D_{21}$ , has a pseudo-stabilizing solution  $Y \ge O$  and they satisfy the inequality in (3.31), then by using the solutions X and Y the  $H_{\infty}$  controller for the generalized plant of  $G_{ms}(s)$  is given with the expression (A.11) which is a solution of the standard  $H_{\infty}$  problem.

**Proof.** In the early paper [48], the  $H_{\infty}$  controller for  $G_{ms}(s)$  is indirectly given by converting an  $H_{\infty}$  controller for  $\hat{G}_{ms}(s)$  which is obtained through a factorization in the closed loop of  $G_{ms}(s)$ . (See Figure 3.7.) Here, it is shown that the  $H_{\infty}$  controller given by the method coincides with the controller which is directly given by using this theorem. Outline of the proof is as follows.

- 1. Construct the generalized plant  $\hat{G}_{ms}(s)$ .
- 2. It is shown that the solutions of the AREs for  $\hat{G}_{ms}(s)$  coincide with the solutions of the AREs for  $G_{ms}(s)$ .
- 3. It is shown that the controller  $K_{ms}(s)$  which is given by using the theorem equals an  $H_{\infty}$  controller  $a^{-1}(s)\hat{K}_{ms}(s)$  which is derived through  $\hat{K}_{ms}(s)$  which is an  $H_{\infty}$  controller for  $\hat{G}_{ms}(s)$ .

<sup>&</sup>lt;sup>†</sup>Since  $\tilde{D}_{21} = I$ ,  $\tilde{D}_{21}^{\dagger} = I$  and  $\tilde{D}_{21}^{\perp} = O$ .

## 3.4. IN THE CASE P(S) HAS $J\omega$ -POLES

Concrete calculations are shown as follows.

1. Let us represent P(s) as follows:

$$P(s) = P_1(s)P_2(s) = \begin{bmatrix} A_{P_2} & O & B_{P_2} \\ B_{P_1}C_{P_2} & A_{P_1} & O \\ \hline D_{P_1}C_{P_2} & C_{P_1} & O \end{bmatrix} \triangleq \begin{bmatrix} A_P & B_P \\ \hline C_P & O \end{bmatrix},$$

where  $P_1(s)$  and  $P_2(s)$  are defined as

$$P_1(s) = \begin{bmatrix} A_{P_1} & B_{P_1} \\ \hline C_{P_1} & D_{P_1} \end{bmatrix}, \quad \operatorname{Re}\left(\lambda_i(A_{P_1})\right) \neq 0, \forall i,$$
$$P_2(s) = \begin{bmatrix} A_{P_2} & B_{P_2} \\ \hline C_{P_2} & O \end{bmatrix}, \quad \operatorname{Re}\left(\lambda_i(A_{P_2})\right) = 0, \forall i.$$

Then by letting the factor a(s) as

$$a(s) = \left[ \begin{array}{c|c} A_{P_2} & B_{P_2} \\ \hline C_a & I \end{array} \right], \quad \operatorname{Re}\left(\lambda_i (A_{P_2} - B_{P_2} C_a)\right) < 0, \forall i \in \mathbb{N}$$

 $\hat{P}(s)$  can be represented as follows

$$\hat{P}(s) = P(s)a^{-1}(s) = \begin{bmatrix} A_{P_2} - B_{P_2}C_a & O & B_{P_2} \\ B_{P_1}C_{P_2} & A_{P_1} & O \\ \hline D_{P_1}C_{P_2} & C_{P_1} & O \end{bmatrix} = \begin{bmatrix} A_{\hat{P}} & B_{P} \\ \hline C_{P} & O \end{bmatrix}.$$
(3.39)

Also, by letting  $C_T = \begin{bmatrix} C_{T_1} & C_{T_2} \end{bmatrix}$ ,  $W_T(s)\hat{P}(s)$  is expressed as follows

$$W_T(s)\hat{P}(s) = \begin{bmatrix} \underline{A_{\hat{P}}} & B_P \\ \hline C_{\hat{T}} & I \end{bmatrix}, C_{\hat{T}} = \begin{bmatrix} C_{T_1} - C_a & C_{T_2} \end{bmatrix}.$$

For the above plant  $\hat{P}(s)$ , the generalized plant for the mixed sensitivity problem is represented as follows

$$\hat{G}_{ms}(s) = \begin{bmatrix} A_W & -B_W C_P & B_W & O \\ O & A_{\hat{P}} & O & B_P \\ \hline C_W & O & O & O \\ O & -C_{\hat{T}} & O & O \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & O & O \\ \hline O & -C_{\hat{T}} & O & I \\ \hline O & O & O \\ \hline O \\ \hline O & O \\ \hline O \\ \hline O & O \\ \hline O \\ \hline$$

2. For the generalized plant  $\hat{G}_{ms}(s)$ , an ARE

$$X\hat{A}_{ZF} + \hat{A}_{ZF}^{T}X + X\hat{R}X + \left(D_{12}^{\perp}\tilde{C}_{1}\right)^{T}D_{12}^{\perp}\tilde{C}_{1} = O,$$

where

$$\hat{A}_{ZF} \triangleq \tilde{A} - B_2 D_{12}^{\dagger} \tilde{C}_1 = A_{ZF} \\ \hat{R} \triangleq B_1 B_1^T - B_2 D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T = R \\ \left( D_{12}^{\perp} \tilde{C}_1 \right)^T D_{12}^{\perp} \tilde{C}_1 = \left( D_{12}^{\perp} C_1 \right)^T D_{12}^{\perp} C_1,$$

coincides with the ARE in (3.12) which is needed to be solved in the mixed sensitivity problem for  $G_{ms}(s)$ . Hence, the solutions for each AREs coincide with each other if they are solvable.

On the other hand, if an ARE for  $\hat{G}_{ms}(s)$ 

$$Y\hat{A}_{ZH}^T + \hat{A}_{ZH}Y + Y\hat{Q}Y = O, \qquad (3.41)$$

where

$$\hat{A}_{ZH} \triangleq \tilde{A} - B_1 \tilde{D}_{21}^{\dagger} \tilde{C}_2$$
$$\hat{Q} \triangleq \tilde{C}_1^T \tilde{C}_1 - \left(\tilde{D}_{21}^{\dagger} \tilde{C}_2\right)^T \tilde{D}_{21}^{\dagger} \tilde{C}_2,$$

has a pseudo-stabilizing solution Y, there exists a full row rank matrix U such that

$$U\hat{A}_{ZH} = \Lambda U, \quad \Lambda = \begin{bmatrix} A_W & O\\ O & A_{P_2} - B_{P_2}C_a \end{bmatrix}$$
$$UY = O.$$

Hence, by applying the equivalent transformation for (3.41)

$$\begin{bmatrix} U \\ (U^{\perp})^T \end{bmatrix} \left( Y \hat{A}_{ZH}^T + \hat{A}_{ZH} Y + Y \hat{Q} Y \right) \begin{bmatrix} U \\ (U^{\perp})^T \end{bmatrix}^T = O$$
(3.42)

It appears that the pseudo-stabilizing solution of the ARE in (3.41) is expressed as

$$Y = \begin{bmatrix} O & O & O \\ O & O & O \\ O & O & Y_l \end{bmatrix},$$
 (3.43)

where  $Y_l \ge O$  is a stabilizing solution of a reduced-order ARE

$$Y_l A_{P_1}^T + A_{P_1} Y_l + Y_l \left( C_{T_2}^T C_{T_2} - C_{P_1}^T C_{P_1} \right) Y_l = O.$$

For  $G_{ms}(s)$ , in a similar way if an ARE

$$YA_{ZH}^T + A_{ZH}Y + YQY = O, (3.44)$$

where

$$A_{ZH} \triangleq A - B_1 \tilde{D}_{21}^{\dagger} \tilde{C}_2$$
$$Q \triangleq C_1^T C_1 - \left(\tilde{D}_{21}^{\dagger} \tilde{C}_2\right)^T \tilde{D}_{21}^{\dagger} \tilde{C}_2,$$

has a pseudo-stabilizing solution Y, for Y there exists a full row rank matrix U such that

$$UA_{ZH} = \Lambda U, \quad \Lambda = \begin{bmatrix} A_W & O \\ O & A_{P_2} \end{bmatrix}$$
$$UY = O.$$

Hence, by applying the same equivalent transformation as in (3.42) for (3.44), it appears that the pseudo-stabilizing solution of the ARE in (3.44) is also represented as (3.43).

#### 3.4. IN THE CASE P(S) HAS $J\omega$ -POLES

3. By using the parameters of the  $H_{\infty}$  controller for  $G_{ms}(s)$  (See appendix A.11), the  $H_{\infty}$  controller for  $\hat{G}_{ms}(s)$  can be represented as follows

$$\hat{K}_{ms}(s) = \mathcal{F}_l \left( \begin{bmatrix} \hat{A} & -ZH_{\infty} & Z\hat{B}_2 \\ \overline{\tilde{F}_{\infty}} & O & I \\ -\hat{C}_2 & I & O \end{bmatrix}, N(s) \right),$$
(3.45)

where

$$\tilde{F}_{\infty} \triangleq F_{\infty} + \begin{bmatrix} O & C_a & O \end{bmatrix}.$$

By calculating  $a^{-1}(s)\hat{K}_{ms}(s)$  and transforming its state with a non-singular matrix

$$T = \begin{bmatrix} I & O & -I & O \\ O & I & O & O \\ O & O & I & O \\ O & O & O & I \end{bmatrix},$$

it is shown the following equation holds.

$$a^{-1}(s)\hat{K}_{ms}(s) = K_{ms}(s) \tag{3.46}$$

**Remark 3.4.1** From the above proof, it is seen that in the mixed sensitivity problem where P(s) has  $j\omega$ -poles, the poles appear as  $j\omega$ -invariant zeros of the controller. Since the invariant zeros of the controller are canceled out by the  $j\omega$ -poles of P(s), the internal stability of the system is not satisfied. In order to avoid the cancellation, the weighting function  $W_S(s)$  is selected as in (3.38).

The following theorem generalizes the result of Lemma 3.4.1, and gives a direct solution to the non-standard  $H_{\infty}$  problem for the generalized plant of G(s) where P(s) has  $j\omega$ -poles.

**Theorem 3.4.1** It is assumed that the strictly proper transfer function P(s), which has  $j\omega$ poles, can be factorized as (3.37). In the generalized plant of Figure 3.3, let us select  $W_S(s)$ as (3.38). Then, if the ARE in (3.12) has a stabilizing solution  $X \ge O$ , the ARE in (3.13) has a pseudo-stabilizing solution  $Y \ge O$  and they satisfy the inequality in (3.31), then using the solutions X and Y, the controller in (3.14) with W(s) = 0 is the  $H_{\infty}$  controller for the generalized plant of G(s).

**Proof.** In the AREs of the non-standard  $H_{\infty}$  problem for G(s), (3.12) coincides with one of the AREs of the standard  $H_{\infty}$  problem for  $G_{ms}(s)$ , and by letting  $L_{H_1} = O$ , (3.13) coincides with the other ARE of the standard  $H_{\infty}$  problem for  $G_{ms}(s)$ . Hence, from the result of Lemma 3.4.1, using the stabilizing solution for the ARE (3.12) and the pseudo-stabilizing



Figure 3.8: A sketch in the proof

solution for the ARE (3.13) with  $L_{H_1} = O$ , the  $H_{\infty}$  controller for  $G_{ms}(s)$ , where P(s) has  $j\omega$ -poles, is given by (A.11).

On the other hand, from the proof of Theorem 3.3.3, it is shown that by letting W(s) = 0, the (1,2)-entry of the controller  $K_{\infty}(s)$  in (3.14) coincides with the controller given by (A.11) and the (1,1)-entry of the controller  $K_{\infty}(s)$  is zero. From these, it is apparent that the controller  $K_{\infty}(s)$  in (3.14) with W(s) = 0 is the  $H_{\infty}$  controller for the generalized plant of G(s). This is sketched in Figure 3.8.

Thus, in the case where P(s) has  $j\omega$ -poles, a design procedure of the robust servo controller is given.

**Proposition 3.4.1** Let  $N_P$  be the number of integrators included in P(s), and let  $N_W$  be the number of integrators included in  $W_S(s)$ . The following procedure yields an integral-type  $H_{\infty}$  controller which includes  $N_W - N_P$  integrators and the controller satisfies the specifications S1, S2 and S3.

#### Design Procedure 2

**STEP 1** Introduce integrators into  $W_S(s)$  as follows:

$$W_S(s) = rac{ ilde{W}_S(s)}{s^n}, \ ilde{W}_S(s) \in \mathcal{RH}_{\infty}, ilde{W}_S(0) \neq 0,$$

where  $n = N_W$  is the sum of the number of integrators which are needed to be included in the controller and the number of integrators which are included in P(s).

**STEP 2** Select an  $L_H$  such that  $L_{H_1} \in \{L_{H_1} : \operatorname{Re}(\lambda_i(A_W + L_{H_1})) < 0\}$  holds. Then, obtain the positive semi-definite stabilizing solution of the ARE in (3.12) and the

#### 3.5. NUMERICAL EXAMPLES

positive semi-definite pseudo-stabilizing solution of the ARE in (3.13). Then, verify the inequality (3.31).

**STEP 3** If the inequality (3.31) is satisfied, set  $L_{H_1} = O$  and W(s) = 0, then the controller (3.32) is a solution.

**Proof.** Since it is apparent that the resultant controller is the  $H_{\infty}$  controller from the above discussion, it suffice to show that the controller has integrators. The full row rank matrix U, which satisfies (3.21) and (3.22), can be given as (3.26), hence, from the generalized plant G(s) represented in (3.8), it can be verified that U satisfies (3.24). Using the result of Lemma 3.3.3, it is shown that the controller (3.32) has integrators, the number of which is same as the number of 0-eigenvalues of  $\Lambda = A_W$ . However, form the relation in (3.46), some of the integrators in the controller are canceled out by zeros on the origin of  $a^{-1}(s)$ .

**Remark 3.4.2** In the above procedure, although, in the case P(s) has  $j\omega$ -poles, it is still necessary to obtain the pseudo-stabilizing solution of the ARE, it is not necessary to separate the  $j\omega$  part from P(s).

**Remark 3.4.3** The order of the resultant controller is equal to that of the generalized plant. Thus, as compared with the controller in (3.10), the order of the controller is reduced by the order of  $W_S(s)$ .

## 3.5 Numerical examples

This subsection illustrates numerical examples of the robust servo control with simple models.

### **3.5.1** P(s) has no $j\omega$ -poles

Let us consider the following system

$$P(s) = \frac{1}{s+1} = \begin{bmatrix} -1 & 1\\ 1 & 0 \end{bmatrix}$$

In order to introduce an integrator into the controller,  $W_S(s)$  is selected as follows

$$W_S(s) = \frac{1/\gamma}{s} = \left[ \frac{0 | 1}{1/\gamma | 0} \right].$$

Let us adopt  $W_T(s)$  as

$$W_T(s)P(s) = \frac{s+0.5}{s+1} = \left[\frac{-1}{-0.5} \mid 1\right]$$

Then the problem is solvable if  $\gamma$  satisfies  $\gamma \ge 1.2$ . Hence, we set  $\gamma = 1.2$ . In this case, a stabilizing solution of the ARE in (3.12) is given as

$$X = \left[ \begin{array}{cc} 25.5 & -25.5 \\ -25.5 & 26.0 \end{array} \right].$$

By setting  $L_{H_1}$  as  $L_{H_1} = -\alpha, \alpha > 0$ , the ARE in (3.13) is represented as follows

$$YA_{ZH}^T + A_{ZH}Y + YQY = O,$$

where

$$A_{ZH} = \begin{bmatrix} -\alpha & 0\\ 0 & -1 \end{bmatrix}, \quad Q = \begin{bmatrix} 1/\gamma^2 & 0\\ 0 & -0.75 \end{bmatrix}.$$

Apparently the stabilizing solution of the ARE is Y = O. By letting  $L_{H_1} = O$ , a central controller is derived from (3.32) as

$$K_{\infty}(s) = \begin{bmatrix} 0 & 25.5 \frac{s+1}{s(s+26.5)} \end{bmatrix}.$$

It can be verified that the controller includes an integrator.

## **3.5.2** P(s) has $j\omega$ -poles

Let us consider the following system

$$P(s) = \frac{1}{s} = \begin{bmatrix} 0 & 1\\ 1 & 0 \end{bmatrix},$$

which includes an  $j\omega$ -pole in  $\omega = 0$ .

In order to introduce an integrator into the controller  $W_S(s)$  is selected as follows

$$W_S(s) = \frac{1/\gamma}{s^2} = \begin{bmatrix} 0 & 0 & 1\\ 1 & 0 & 0\\ 0 & 1/\gamma & 0 \end{bmatrix}.$$

Let us adopt  $W_T(s)$  as

$$W_T(s)P(s) = \frac{s+0.5}{s} = \left[\frac{0 | 1}{0.5 | 1}\right].$$

Then the problem is solvable if  $\gamma$  satisfies  $\gamma \geq 3.6$ . Hence,  $\gamma = 3.6$  is adopted. By obtaining a stabilizing solution of an ARE in (3.12) and obtaining a pseudo stabilizing solution of the ARE in (3.13), an  $H_{\infty}$  controller (a central solution) is derived from (3.32).

$$K_{\infty}(s) = \left[ \begin{array}{cc} 0 & 265.3 \frac{s+0.322}{s(s+266.3)} \end{array} \right]$$

It can also be verified that the controller includes an integrator.

## 3.6 Summary

In this chapter, we have considered a synthesis of the low-order integral-type  $H_{\infty}$  control system, where two types of plants: integral-type and non-integral-type, are treated. The synthesis problem is formulated as the non-standard  $H_{\infty}$  problem, where a direct feed-through term of the subsystem from the external input to the measurement output is of full column rank. Although the formulation resembles the results by Zhang et al. [71] and Hozumi et al. [26], the solutions are distinctive. This approach to the integral-type robust controller design is based on the non-standard  $H_{\infty}$  control problem. The controller is given with low order by using the solutions of the AREs. In the case where the plant has no integrators, it is not necessary to solve the so-called pseudo stabilizing solution of the ARE. We have extended the result to the case where the plant has the integrators. In this case, although it is still necessary to solve the pseudo stabilizing solution of the ARE, there is no necessity for transforming the non-standard problem to the standard problem and separating the integrators from the plant. Moreover, the results given in this chapter can be extended to the twodegree-of-freedom controller design problem which can also be reduced to the non-standard  $H_{\infty}$  control problem. This problem will be treated in the following chapter.

## Chapter 4

# Trade-off analysis of a low-order TDF control system

## 4.1 Introduction

Closed loop stabilization, disturbance elimination, and reference tracking in the presence of uncertainties of a model – these are the main goals of a robust control system design, and TDF control is one of the most effective ways to reach all these goals simultaneously [44, 22, 13, 29]. In recent years, there has been many studies concerned about application of  $H_{\infty}$  and  $H_2$ control theory in the designing of TDF control systems [17, 38, 7, 4, 27]. Most of these studies focused on transformation of the TDF control problems to the standard  $H_{\infty}$  or  $H_2$ control problems. This has resulted in high-performance TDF controllers, but excessive transformation sometimes results in controllers of unacceptably high order. That is, the McMillan degree of the  $H_{\infty}$  controller is no less than that of the generalized plant, especially in the TDF case, it becomes no less than three times as high as that of the plant. Therefore, it is required to study the low-order TDF controller design.

On the basis of such a background, some earlier studies have tackled the problem of designing low-order controllers [5, 6, 24, 7, 38, 17]. These studies are based on the idea of using the same dynamics in both of the feedback controller and the feedforward controller, where some of these studies [5, 6, 24, 17] have adopted a sequential design in which the feedback controller and the feedforward controller are designed separately. In this approach, a low-order TDF controller can be designed. A key feature of the approach is that the zeros and the gain of the feedforward controller are optimized in the second step, which is preceded by designing of the feedback controller. Therefore, it should be noted that, in this approach, the resultant feedforward controller has the same pole position as the feedback controller has, and it is not necessarily optimal for tracking performance. In general, it is well known that the feedback performance and the tracking performance are independently specified in the TDF control system [66, 44], but it is not known whether this is true in the case of the same-dynamic TDF control system. Since the answer to this question affects the optimality



Figure 4.1: Basic structure of TDF control system

in the low-order TDF controller design, the question must be investigated.

Thus, prior to considering the synthesis of the low-order TDF controller, this chapter is devoted to clarifying the relationship between the feedback performance and the feedforward performance in the case where the TDF controller is low order, especially, the feedback controller and the feedforward controller share the common dynamics. Thus, this chapter analyzes the trade-off of the low-order TDF control system. It is shown that the independence of the feedback performance and the tracking performance is not maintained in this case, and that the low-order TDF controller should therefore be constructed with a simultaneous design of the feedback controller and the feedforward controller.

## 4.2 Basic analysis and design of the TDF control system

In this section, we analyze the feedback properties and the feedforward properties in the TDF control system by representing the class of transfer functions of the TDF control system. Based on the analysis we introduce a basic design of the TDF control system by using the  $H_{\infty}$  control. Then, we evaluate the order of the resultant TDF controller.

#### 4.2.1 Basic analysis of the TDF control system

Consider the TDF control system shown in Figure 4.1. P(s) represents a plant model. K(s) is the feedback controller and F(s) is the feedforward controller. Where,  $r \in \mathbb{R}^p$  is the reference input,  $u \in \mathbb{R}^m$  is the control input,  $y \in \mathbb{R}^p$  is the measurement output and  $d, n \in \mathbb{R}^p$ respectively represent disturbance input and observation noise.

Then the feedback performance of the closed loop system is characterized with the transfer functions from d to y and n to y, denoted respectively  $T_{yd}(s)$  and  $T_{yn}(s)$ .

$$T_{yd}(s) = (I + P(s)K(s))^{-1} = S(s)$$
(4.1)

$$T_{yn}(s) = -(I + P(s)K(s))^{-1}P(s)K(s) = -T(s)$$
(4.2)

Where, the transfer functions S(s) and T(s) are known as the sensitivity function and the

complementary sensitivity function, which are defined as follows

$$S(s) \triangleq (I + P(s)K(s))^{-1}$$
$$T(s) \triangleq (I + P(s)K(s))^{-1}P(s)K(s).$$

The tracking performance is characterized with the transfer function

$$T_{yr}(s) = (I + P(s)K(s))^{-1}P(s)F(s).$$
(4.3)

Now let us represent P(s), K(s) and F(s) as the coprime factorization [67, 44] over  $\mathcal{RH}_{\infty}$ , i.e.,

$$P(s) = N(s)M^{-1}(s) = \tilde{M}^{-1}(s)\tilde{N}(s)$$
(4.4)

$$K(s) = N_K(s)M_K^{-1}(s) = \tilde{M}_K^{-1}(s)\tilde{N}_K(s)$$
(4.5)

$$F(s) = N_F(s)M_F^{-1}(s) = \tilde{M}_F^{-1}(s)\tilde{N}_F(s)$$
(4.6)

where all of the factors are the elements of  $\mathcal{RH}_{\infty}$  and coprime. Then it is known that the internal stability condition of the system in the Figure 4.1 is given as follows.

**Lemma 4.2.1** Let (4.4), (4.5) and (4.6) be coprime factorization over  $\mathcal{RH}_{\infty}$ , then the TDF control system in figure 4.1 is internal stable iff

$$\begin{cases} K(s) \in \mathcal{K} \\ \tilde{M}_K(s)F(s) \in \mathcal{RH}_\infty \end{cases}$$

Here,  $\mathcal{K}$  is the class of the feedback controllers achieving the internal stability, i.e.,

$$\mathcal{K} = \left\{ \left( Y(s) - Q(s)\tilde{N}(s) \right)^{-1} \left( X(s) + Q(s)\tilde{M}(s) \right) \mid \forall Q(s) \in \mathcal{RH}_{\infty} \right\} \\ = \left\{ \left( \tilde{X}(s) + M(s)Q(s) \right) \left( \tilde{Y}(s) - N(s)Q(s) \right)^{-1} \mid \forall Q(s) \in \mathcal{RH}_{\infty} \right\},$$

where,  $X(s), Y(s), \tilde{X}(s), \tilde{Y}(s)$  are the solutions of the Bezout identity

$$\begin{bmatrix} X(s) & Y(s) \\ -\tilde{M}(s) & \tilde{N}(s) \end{bmatrix} \begin{bmatrix} N(s) & -\tilde{Y}(s) \\ M(s) & \tilde{X}(s) \end{bmatrix} = I.$$

**Proof.** See reference [66].

From Lemma 4.2.1, the class of internally stabilizing controllers K(s), F(s) are represented with free-parameters Q(s),  $R(s) \in \mathcal{RH}_{\infty}$ .

$$\mathcal{K} = \left\{ \left( Y(s) - Q(s)\tilde{N}(s) \right)^{-1} \left( X(s) + Q(s)\tilde{M}(s) \right) \mid \forall Q(s) \in \mathcal{RH}_{\infty} \right\}$$
$$\mathcal{F} = \left\{ \left( Y(s) - Q(s)\tilde{N}(s) \right)^{-1} R(s) \mid \forall R(s) \in \mathcal{RH}_{\infty} \right\}$$


Figure 4.2: Feedback controller design

By substituting controllers  $K(s) \in \mathcal{K}$  and  $F(s) \in \mathcal{F}$  into (4.1), (4.2) and (4.3), the class of transfer functions  $T_{yd}(s), T_{yn}(s)$  and  $T_{yr}(s)$  which are achievable in the TDF system are represented as

$$\begin{aligned}
\mathcal{T}_{yd} &= \left\{ \left( \tilde{Y}(s) - N(s)Q(s) \right) \tilde{M}(s) \mid \forall Q(s) \in \mathcal{RH}_{\infty} \right\} \\
\mathcal{T}_{yn} &= \left\{ N(s) \left( X(s) + Q(s)\tilde{M}(s) \right) \mid \forall Q(s) \in \mathcal{RH}_{\infty} \right\} \\
\mathcal{T}_{yr} &= \left\{ N(s)R(s) \mid \forall R(s) \in \mathcal{RH}_{\infty} \right\}.
\end{aligned} \tag{4.7}$$

Note that Q(s) and R(s) are the independent free-parameters of  $\mathcal{K}$  and  $\mathcal{F}$ . It can be seen that there doesn't exist any trade-off between the feedback performances and the tracking performance, hence these performances are independently specified with those free-parameters. This property affects the way of TDF controller construction, that is, since the TDF performances are optimized with a series of sequential optimization of the feedback controller and the feedforward controller, the TDF controller is frequently designed through two steps: a feedback controller design and a feedforward controller design.

#### 4.2.2 A basic design of the TDF controller

We show a basic approach to designing the TDF control system. This approach of the design is a natural way that can be occurred immediately from the previous analysis. That is, a sequential design of the feedback controller and the feedforward controller can satisfy each of the performances in the TDF control system.

Thus, first of all, let's design the feedback controller by way of solving the mixed sensitivity problem. The generalized plant of the problem is written in Figure 4.2 where the meanings of symbols are identical to that in Figure 3.2. Here, the  $H_{\infty}$  problem of the generalized plant  $G_{ms}(s)$  is solved. Since the order of the resultant controller is no less than that of  $G_{ms}(s)$ , if we denote the order of each system in boxes as deg (•), this design derives  $H_{\infty}$  controllers of



Figure 4.3: Feedforward controller design

order

$$\deg (K(s)) \ge \deg (P(s)) + \deg (W_S(s)) + \deg (W_T(s))$$

In the second step, let's design the feedforward controller by way of solving the model matching problem. The generalized plant of the problem is written in Figure 4.3 where  $W_M(s)$  is a transfer function of an ideal tracking performance,  $W_\rho(s)$  is a weighting function of the tracking error. Here, the  $H_\infty$  problem of the generalized plant  $G_{mmp}(s)$  is solved. Thus, the design derives  $H_\infty$  controllers of order

$$\deg (F(s)) \geq \deg (K(s)) + \deg (P(s)) + \deg (W_M(s)) + \deg (W_\rho(s))$$

$$= 2 \deg (P(s)) + \deg (W_S(s)) + \deg (W_T(s))$$

$$+ \deg (W_M(s)) + \deg (W_\rho(s)) .$$

In order to evaluate the order of the TDF controller, let us represent the state-space equations of the feedback controller and the feedforward controller as follows.

$$F: \left\{ \begin{array}{l} \dot{x}_f = A_F x_f + B_F r\\ u_f = C_F x_f + D_F r \end{array} \right. K: \left\{ \begin{array}{l} \dot{x}_k = A_K x_k + B_K \tilde{y}\\ u_k = C_K x_k + D_K \tilde{y} \end{array} \right.$$

Then the dynamics of the TDF controller is composed of both of the dynamics of the controllers.

$$\begin{cases} \begin{bmatrix} \dot{x}_f \\ \dot{x}_k \end{bmatrix} = \begin{bmatrix} A_F & O \\ O & A_K \end{bmatrix} \begin{bmatrix} x_f \\ x_k \end{bmatrix} + \begin{bmatrix} B_F \\ O \end{bmatrix} r + \begin{bmatrix} O \\ B_K \end{bmatrix} \tilde{y} \\ u = \begin{bmatrix} C_F & C_K \end{bmatrix} \begin{bmatrix} x_f \\ x_k \end{bmatrix} + D_F r + D_K \tilde{y}$$

$$(4.8)$$

As shown in (4.8), the McMillan degree of the TDF controller is the sum of the degrees of these controllers. Thus the TDF controller (F(s), K(s)) of order

$$\deg (F(s), K(s)) \geq 3\deg (P(s)) + 2\deg (W_S(s)) + 2\deg (W_T(s)) + \deg (W_M(s)) + \deg (W_\rho(s)).$$

can be constructed. From this fact, the natural design of the TDF controller yields high-order controller. This is a basic motivation for the low-order TDF controller design.

#### 4.3 An idea of sharing common dynamics

As shown in the previous section, the McMillan degree of the TDF controller that is designed in the natural way becomes very high. Sometimes we are faced with the situation such that the degree of the TDF controller is too high to implement it as hardware. So it is important to consider designing of the low-order TDF controller. One of the effective ways to design the low-order TDF controller is to share common dynamics between these controllers[5, 6, 24]. In this section we define the share of common dynamics in the TDF controller, and show a basic design method of the low-order TDF controller.

#### **4.3.1** Sharing common dynamics between K(s) and F(s)

Here, let us define a low-order TDF controller in which the feedback controller and the feedforward controller share common dynamics.

**Definition 4.3.1** The TDF controller which shares common dynamics between K(s) and F(s) is defined as follows.

$$\begin{bmatrix} F_s(s) & K_s(s) \end{bmatrix} = \begin{bmatrix} A & B_F & B_K \\ \hline C & D_F & D_K \end{bmatrix},$$
(4.9)

where both of the pairs  $(A, B_K)$  and  $(A, B_F)$  are controllable, and the pair (A, C) is observable. In the above description, the subscript "s" means "share".

It can be seen that the order of the TDF controller  $(F_s(s), K_s(s))$  is

$$\deg\left(F_{s}(s),K_{s}(s)\right) = \deg\left(F_{s}(s)\right) = \deg\left(K_{s}(s)\right),$$

hence by sharing common dynamics between the feedback controller and the feedforward controller, the order of the TDF controller can be reduced.

In the TDF control system in which the controllers share common dynamics, the internal stability is satisfied if  $K_s(s)$  satisfies the internally stability of the feedback loop.

**Lemma 4.3.1** Let us suppose  $K_s(s) \in \mathcal{K}$ , then the TDF controller given as (4.9) satisfies the stability of the TDF control system.

**Proof.** Let us give a coprime factorization of  $K_s(s)$  over  $\mathcal{RH}_{\infty}$  as

$$K_s(s) = M_K^{-1}(s)N_K(s)$$
  

$$\tilde{M}_K(s) = \left[ \begin{array}{c|c} A + HC & H \\ \hline C & I \end{array} \right]$$
(4.10)

$$\tilde{N}_K(s) = \begin{bmatrix} A + HC & B_K + HD_K \\ \hline C & D_K \end{bmatrix},$$
(4.11)



Figure 4.4: A generalized plant for a TDF controller design

where  $\tilde{M}_K(s) \in \mathcal{RH}_{\infty}$  and  $\tilde{N}_K(s) \in \mathcal{RH}_{\infty}$  are left coprime, and  $\operatorname{Re}(\lambda_i(A + HC)) < 0, \forall i$  is satisfied. Then  $F_s(s)$  can be also factorized over  $\mathcal{RH}_{\infty}$  as follows

$$F_s(s) = \tilde{M}_K^{-1}(s)\tilde{N}_F(s) \qquad (4.12)$$
  
$$\tilde{N}_F(s) = \left[ \begin{array}{c|c} A + HC & B_F + HD_F \\ \hline C & D_F \end{array} \right],$$

where  $\tilde{M}_K(s) \in \mathcal{RH}_{\infty}$  and  $\tilde{N}_K(s) \in \mathcal{RH}_{\infty}$  are left coprime. The equation (4.12) implies

$$\tilde{M}_K(s)F_s(s) = \tilde{N}_F(s) \in \mathcal{RH}_\infty,$$

hence from Lemma 4.2.1 the TDF system is internally stable.

From Lemma 4.3.1, in order to satisfy the internal stability of the TDF control system in which the controllers share the common dynamics, it is only needed to consider the internal stability of the feedback controller.

#### 4.3.2 A basic design of a low-order TDF controller

Here, we show a basic method for designing the low-order TDF controller where one dynamics is shared by the feedback controller and the feedforward controller. Firstly, let it be designed a feedback controller of the form:

$$K_s(s) = \begin{bmatrix} A_K & B_K \\ \hline C_K & D_K \end{bmatrix}.$$
(4.13)

Then, construct a generalized plant as in Figure 4.4, where  $G_{mmp}(s)$  is the generalized plant of the model matching problem and  $G_{tdf}(s)$  is the generalized plant for the TDF controller design. The meaning of each subsystem is same as that in Figure 4.3. The generalized plant  $G_{tdf}(s)$  can be represented as follows:

$$G_{tdf}(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & D_{11} & D_{12} \\ \hline C_{2v} & O & \hline O & O \\ O & I & O \end{bmatrix}.$$
 (4.14)

Since the feedback controller has already been designed, the state-space parameters in (4.13) are fixed. In order to design the feedforward controller which shares the same dynamics with  $K_s(s)$  in (4.13), we put the state-space data of the feedforward controller as follows:

$$F_s(s) = \begin{bmatrix} A_K & B_F \\ \hline C_K & D_F \end{bmatrix}, \tag{4.15}$$

where  $B_F$  and  $D_F$  are the parameters which are to be designed.

By using those state-space representations in (4.13), (4.14) and (4.15), the state-space data of the closed loop system  $T_{zw}(s)$  is represented as follows:

$$T_{zw}(s) = \mathcal{F}_l \left( G_{tdf}(s), \begin{bmatrix} F_s(s) & K_s(s) \end{bmatrix} \right) \\ = \begin{bmatrix} A_{cl} & B_{cl} \\ \hline C_{cl} & D_{cl} \end{bmatrix},$$

where

$$A_{cl} = \begin{bmatrix} A + B_2 D_K C_{2v} & B_2 C_K \\ B_K C_{2v} & A_K \end{bmatrix}$$
$$B_{cl} = \begin{bmatrix} B_1 + B_2 D_F \\ B_F \end{bmatrix} = E + J\mathcal{R}$$
$$C_{cl} = \begin{bmatrix} C_1 + D_{12} D_K C_{2v} & D_{12} C_K \end{bmatrix}$$
$$D_{cl} = D_{11} + D_{12} D_F = H + L\mathcal{R}.$$

Here, it can be seen that the matrices  $A_{cl}$  and  $C_{cl}$  are composed of the fixed parameters, and both of  $B_{cl}$  and  $D_{cl}$  include design parameters  $B_F$  and  $D_F$ . Hence, we put the variable composed of  $B_F$  and  $D_F$  as

$$\mathcal{R} \triangleq \left[ \begin{array}{c} B_F \\ D_F \end{array} \right],$$

and the other fixed parameters are defined as follows:

$$E \triangleq \begin{bmatrix} B_1 \\ O \end{bmatrix}, J = \begin{bmatrix} O & B_2 \\ I & O \end{bmatrix}, H \triangleq D_{11}, L \triangleq \begin{bmatrix} O & D_{12} \end{bmatrix}.$$

Under the above preparation, the following result is obtained.

**Theorem 4.3.1** In the TDF control system of Figure 4.4, the closed loop system is stable and it satisfies a condition such that

$$\|T_{zw}(s)\|_{\infty} < \gamma,$$

100

if and only if there exists a positive definite matrix X which satisfies the following inequality:

$$\begin{bmatrix} XA_{cl}^T + A_{cl}X & E + J\mathcal{R} & XC_{cl}^T \\ (E + J\mathcal{R})^T & -\gamma I & (H + L\mathcal{R})^T \\ C_{cl}X & H + L\mathcal{R} & -\gamma I \end{bmatrix} < 0.$$
(4.16)

**Proof.** By using the bounded real lemma, it can be immediately deduced.

**Remark 4.3.1** The matrix inequality in (4.16) is a Linear Matrix Inequality, hence by using some convex programming method we can solve the inequality.

#### 4.4 Trade-off analysis in the low-order TDF control system

In the previous section, a method of the low-order TDF controller is introduced. However, it is unclear whether sharing dynamics between F(s) and K(s) causes a conflict in the TDF performances or not. If the answer to the question is "yes", the trade-off in the performances must be considered in the design of the low-order TDF controller. The two step design is not adequate for balancing the trade-off in the designing of the low-order TDF controller.

# 4.4.1 Trade-off between the feedback performance and the feedforward performance

In the above section, the TDF controller in which F(s) and K(s) share common dynamics is defined and it is represented in the state-space formula. Then, this section evaluates degradation in the performance of the feedforward controller when the performance of the feedback controller is specified first.

**Lemma 4.4.1** Let us represent  $K_s(s) \in \mathcal{K}$  as a right coprime factorization over  $\mathcal{RH}_{\infty}$ .

$$K_s(s) = (Y(s) - Q(s)\tilde{N}(s))^{-1}(X(s) + Q(s)\tilde{M}(s))$$
  

$$Y(s) - Q(s)\tilde{N}(s) \triangleq \tilde{M}_K(s), \quad X(s) + Q(s)\tilde{M}(s) \triangleq \tilde{N}_K$$
(4.17)

Then all the class of  $F_s(s) \in \mathcal{F}$  which shares a dynamics with  $K_s(s)$  is given by

$$\mathcal{F}s(s) = \left\{ \tilde{M}_K^{-1}(s)\tilde{R}(s) \mid \forall Q(s) \in \mathcal{RH}_{\infty}, \forall B_R \in \mathbb{R}^{n \times p}, \forall D_R \in \mathbb{R}^{m \times p} \right\},$$
(4.18)

where  $\tilde{R}(s)$  is represented as follows

$$\tilde{R}(s) = \begin{bmatrix} A + HC & B_R \\ \hline C & D_R \end{bmatrix}$$

and  $\tilde{M}_K(s)$  and  $\tilde{R}(s)$  are left coprime.

**Proof.** (necessity: ) Let us represent  $K(s) \in \mathcal{K}$  and  $F(s) \in \mathcal{F}$  with coprime factorizations as follows

$$K(s) = \tilde{M}_{K}^{-1}(s)\tilde{N}_{K}(s)$$

$$F(s) = \tilde{M}_{K}^{-1}(s)R(s), \quad R(s) \in \mathcal{RH}_{\infty}.$$

$$(4.19)$$

If R(s) is denoted as  $R(s) = \begin{bmatrix} A_R & B_R \\ \hline C_R & D_R \end{bmatrix}$ , then by using (4.10), F(s) can be represented as follows

$$F(s) = \begin{bmatrix} A & HC_R & HD_R \\ O & A_R & B_R \\ \hline -C & C_R & D_R \end{bmatrix}$$
$$= \begin{bmatrix} A & A + HC_R - A_R & HD_R - B_R \\ O & A_R & B_R \\ \hline -C & -C + C_R & D_R \end{bmatrix}.$$
(4.20)

From the equation (4.20), in order to share common dynamics between F(s) and K(s), it is necessary to satisfy  $A_R = A + HC$ ,  $C_R = C$ . Hence the feedforward controller and feedback controller share common dynamics only if F(s) is given as follows

$$F(s) = \tilde{M}_K^{-1}(s)\tilde{R}(s). \tag{4.21}$$

(sufficiency: ) It is easily verified that F in (4.21) has the same dynamics of  $K_s(s)$  and satisfies the internal stability of TDF system from Lemma 4.3.1.

From the above result it is possible to show the class of the transfer functions which can be attained with the TDF controller in (4.9).

**Theorem 4.4.1** (The class of tracking performance) With the TDF controller, which shares the same dynamics, any tracking performance that belongs to the class:

$$\tilde{\mathcal{T}}_{yr} = \left\{ N(s)\tilde{R}(s) \mid \forall B_R \in \mathbb{R}^{n \times p}, \forall D_R \in \mathbb{R}^{m \times p} \right\}$$
(4.22)

can be attained.

**Proof.** Substituting (4.18) into (4.3) leads the class.

From equations (4.10), (4.11) and (4.17), it is apparent that the pair (A+HC, C) depends on the parameter  $Q(s) \in \mathcal{RH}_{\infty}$ . Hence the class of  $\tilde{R}(s)$ :

$$\tilde{\mathcal{R}} \triangleq \left\{ \tilde{R}(s) \mid \forall B_R \in \mathbb{R}^{n \times p}, \forall D_R \in \mathbb{R}^{m \times p} \right\}$$

also depends on Q(s) and the inclusion

$$\mathcal{R} \subset \mathcal{RH}_{\infty}$$

holds. By comparing (4.7) and (4.22), it is verified that the tracking performance is restricted by the feedback controller design if the TDF controller share the same dynamics.

Let us define the transfer function of the requiring tracking performance as  $M_{yr}(s)$  and let  $T_{yr}(R(s)) \triangleq N(s)R(s)$  be the tracking performance, then an objective function for the evaluation of the tracking performance can be selected as follows

$$f(R(s)) \triangleq M_{yr}(s) - T_{yr}(R(s))$$
.

102

Then the class  $\tilde{\mathcal{R}}$  is restricted by the design of the feedback controller. Hence the following inequality

$$\min_{R(s)\in\mathcal{RH}_{\infty}} \left\| f\left(R(s)\right) \right\| \le \min_{\tilde{R}(s)\in\tilde{\mathcal{R}}} \left\| f\left(\tilde{R}(s)\right) \right\|$$

where  $\| \bullet \|$  indicates any norm of a transfer function, holds. The above inequality indicates that if the feedback performance is specified in the first time, the tracking performance may be sacrificed.

#### 4.5 Summary

In this chapter, based on the basic analysis and synthesis of the TDF control system, we have motivated to design the low-order TDF controller. Then, we have introduced designing of a low-order TDF controller where a feedback controller and a feedforward controller share common dynamics. Then, we have analyzed the trade-off of the low-order TDF control system. From the analysis it is shown that the independent property of the feedback performance and the feedforward performance in the basic TDF control system is not maintained in the low-order TDF control system. Then, it is pointed out that the low-order TDF controller should be constructed with a simultaneous design of the feedback controller and the feedforward controller. In the next chapter we will introduce the simultaneous design method of the low-order TDF controller.

## Chapter 5

# A synthesis of low-order TDF controller

#### 5.1 Introduction

The previous chapter has analyzed the trade-off between the feedback performance and the tracking performance in a low-order TDF control system where the controllers share common dynamics. It is clarified that there exists a conflict between those performances, and that the independence in the properties of the feedback controller and the feedforward controller is not maintained in the low-order TDF control system. It is difficult to consider the trade-off in a sequential approach, where a feedback controller and a feedforward controller are designed separately, the low-order TDF control should be designed with a simultaneous approach, where the feedback controller are designed simultaneous approach.

Based on the analysis, the present chapter proposes a method for designing the loworder TDF controller in a simultaneous approach. Remarkable point of this chapter is the reduction of the low-order TDF control problem to the non-standard  $H_{\infty}$  problem of case 2. This formulation enables the TDF controller to be derived simultaneously. As a result, the feedforward controller and the feedback controller are designed such that the order is lower than that of the controller designed with the sequential approach based on the standard  $H_{\infty}$ problem. Furthermore, a reduced-order controller, which has dynamics of lower-order than the dynamics of a generalized plant, is derived by an algebraic operation using free parameters of the general solution.

#### 5.2 Problem descriptions and comparison

This section describes some problems for the low-order TDF controller design and compares the properties of the problems with each other.

Consider a system shown in Figure 5.1, which is a generalized plant for the TDF control system design. Where P(s) is a plant, K(s) is a feedback controller and F(s) is a feedforward controller. The signal  $d \in \mathbb{R}^p$  indicates a disturbance input,  $u_f \in \mathbb{R}^m, u_k \in \mathbb{R}^m$  are the



Figure 5.1: A generalized plant

outputs from the feedforward and feedback controllers, and  $r \in \mathbb{R}^p, y \in \mathbb{R}^p$  respectively indicate a reference input and control output. The signals  $z_i, i = 1, 2, 3$  are used to evaluate the control performances of the TDF control system. Concretely,  $z_1 \in \mathbb{R}^p$  is a index of the tracking performance,  $z_2 \in \mathbb{R}^p$  and  $z_3 \in \mathbb{R}^p$  are the indices of the feedback performances.  $W_M(s), W_p(s), W_T(s)$  and  $W_S(s)$  are the weighting functions for each specification. The generalized plant is arranged and replaced with Figure 5.2.



Figure 5.2: A generalized plant

#### 5.2.1 Two-step design

First of all, as seen in the basic TDF controller design, the two-step design is one of the most popular ways to construct the TDF control system.

Problem 5.2.1 (Two-step design I) Firstly, let's obtain a feedback controller

$$K^*(s) = \left[\begin{array}{c|c} A_{K^*} & B_{K^*} \\ \hline C_{K^*} & D_{K^*} \end{array}\right] \in \mathcal{K}, A_{K^*} \in \mathbb{R}^{n_k \times n_k}$$

which satisfies an inequality for the feedback performances

$$\left\|T_{z_{3}d}^{z_{2}}\left(K(s)\right)\right\|_{\infty} < \gamma,\tag{5.1}$$

where  $\mathcal{K}$  denotes the class of the internally stabilizing feedback controllers,  $T_{z_3 d}^{z_2}(K(s))$  denotes a transfer function matrix:

$$T_{z_{2d}}^{z_{2d}}\left(K(s)\right) \triangleq \left[\begin{array}{c} T_{z_{2d}}\left(K(s)\right) \\ T_{z_{3d}}\left(K(s)\right) \end{array}\right],$$

and  $T_{ab}(s)$  denotes a transfer function from the signal b to a.

Then, find the feedforward controller F(s) which is the solution of the minimization problem described as follows

$$\underset{F(s)\in\mathcal{F}}{\text{minimize}} \quad \|T_{z_1r}\left(F(s),K^*(s)\right)\|_{\infty},$$

where  $\mathcal{F}$  indicates the class of the stabilizing feedforward controllers.

**Remark 5.2.1** Assume that the order of F(s) is  $n_f$ , then the maximum order of the TDF controller is  $n_k + n_f$ . As seen in chapter 4, solving Problem 5.2.1 derives a high-performance but high-order TDF controller.

In the next problem, we intend to derive a low-order TDF controller.

Problem 5.2.2 (Two-step design II) Let's obtain a feedback controller

$$K^*(s) = \left[\begin{array}{c|c} A_{K^*} & B_{K^*} \\ \hline C_{K^*} & D_{K^*} \end{array}\right] \in \mathcal{K}, A_{K^*} \in \mathbb{R}^{n_k \times n_k}$$

which satisfies an inequality in (5.1).

Then, find the feedforward controller F(s) which is the solution of the minimization problem described as follows

$$\min_{F(s)\in\mathcal{F}(K^*(s))} \|T_{z_1r}\left(F(s),K^*(s)\right)\|_{\infty},$$
(5.2)

where the class  $\mathcal{F}(K^*(s))$  indicates

$$\mathcal{F}(K^*(s)) \triangleq \left\{ \left[ \begin{array}{c|c} A_K^* & B_F \\ \hline C_K^* & D_F \end{array} \right] | B_F \in \mathbb{R}^{n_k \times p}, D_F \in \mathbb{R}^{m \times p} \right\} \subset \mathcal{F}.$$
(5.3)

**Remark 5.2.2** The TDF controller that is derived by solving Problem 5.2.2 shares common dynamics between the feedback controller and the feedforward controller. Thus the order of the TDF controller is  $n_k$ , which is lower than that of the TDF controller derived by solving Problem 5.2.1.

Lemma 5.2.1 The optimal values of Problem 5.2.1 and Problem 5.2.2 satisfy

$$\min_{F(s)\in\mathcal{F}} \|T_{z_{1}r}(F(s),K^{*}(s))\|_{\infty} \leq \min_{F(s)\in\mathcal{F}(K^{*}(s))} \|T_{z_{1}r}(F(s),K^{*}(s))\|_{\infty}.$$

**Proof.** From the inclusion in (5.3), the above inequality is immediately verified.

**Remark 5.2.3** By solving Problem 5.2.2, it is possible to design the low-order controller in which common dynamics is shared between K(s) and F(s). From (5.2) and (5.3) it is seen that  $A_{K^*}$  and  $C_{K^*}$ , which are designed in the first step, are the fixed parameters of  $F(s) \in$  $\mathcal{F}(K^*(s))$ . Thus, the freedom in the designing parameters of F(s) is restricted by designing of the feedback controller, hence the selection of the feedback controller may affect the minimum value of the objective function in (5.2) if the feedback controller which satisfies the restriction (5.1) is not unique. This also holds when the feedforward controller is designed in the first step. Hence, essentially it is important to make a compromise between those performances simultaneously.

#### 5.2.2 Simultaneous design

Based on the discussion in the above section, this section considers a simultaneous TDF controller design where the feedback controller and the feedforward controller are designed at the same time. In the  $H_{\infty}$  control, it is possible to design the multi-input and multi-output controller, hence the problem of the simultaneous TDF controller design can be formulated as a double-input and single-output controller design. Then, in the resultant TDF controller the feedback controller and the feedforward controller share common dynamics. Hence the designing parameters of F(s) come to be restricted by the parameters of K(s). Firstly, let's consider the following problem.

**Problem 5.2.3 (Constrained optimization problem)** Find  $F(s) \in \mathcal{F}$  and  $K(s) \in \mathcal{K}$  which achieve the specifications such that

$$\begin{array}{l} \underset{F(s)\in\mathcal{F},K(s)\in\mathcal{K}}{\text{minimize}} & \|T_{z_1r}\left(F(s),K(s)\right)\|_{\infty}\\ \text{subject to} & \left\|T_{z_2d}\left(K(s)\right)\right\|_{\infty} < \gamma. \end{array}$$

**Remark 5.2.4** Problem 5.2.3 is a minimization problem with a constraint. In the problem, the feedforward controller and the feedback controller which satisfy the optimal tracking performance subject to the constraint of the feedback performance are simultaneously designed. Thus, in this problem the controllers are designed in a single step, hence the order of the resultant TDF controller isn't so high as the TDF controller obtained in the two-step design of Problem 5.2.1.

Our interest in this formulation is the improvement of the tracking performance in comparison with the design via Problem 5.2.2.

**Lemma 5.2.2** Let us denote the solutions of each problem  $F_i^*(s)$  and  $K_i^*(s)$ , where subscript *i* indicates the number of Problem 5.2.*i*. Then the following inequality holds.

$$\|T_{z_1r}(F_3^*(s), K_3^*(s))\|_{\infty} \le \|T_{z_1r}(F_2^*(s), K_2^*(s))\|_{\infty}$$
(5.4)

#### 5.2. PROBLEM DESCRIPTIONS AND COMPARISON

**Proof.**  $(F_i^*(s), K_i^*(s)), i = 2, 3$  are represented as follows

$$(F_{2}^{*}(s), K_{2}^{*}(s)) = \underset{(F(s), K^{*}(s)) \in \tilde{\Omega}}{\operatorname{argmin}} \|T_{z_{1}r}(F(s), K^{*}(s))\|_{\infty}$$
$$(F_{3}^{*}(s), K_{3}^{*}(s)) = \underset{(F(s), K(s)) \in \Omega}{\operatorname{argmin}} \|T_{z_{1}r}(F(s), K^{*}(s))\|_{\infty},$$

where classes of  $\tilde{\Omega}$  and  $\Omega$  are defined as

$$\begin{split} \tilde{\Omega} &\triangleq \left\{ \left( F(s), K^*(s) \right) | F(s) \in \mathcal{F} \left( K^*(s) \right), K^*(s) : fixed \right\} \\ \Omega &\triangleq \left\{ \left( F(s), K(s) \right) | F(s) \in \mathcal{F}, K(s) \in \mathcal{K}, \left\| T_{\frac{z_2}{z_3}d} \left( K(s) \right) \right\|_{\infty} < \gamma \right\}. \end{split}$$

It can be easily verified that the inclusion:

 $\tilde\Omega\subset\Omega$ 

holds, hence it is also verified that the inequality (5.4) holds.

In the  $H_{\infty}$  problem, it is hard to solve Problem 5.2.3 directly. Hence we solve the following problem instead of solving Problem 5.2.3.

**Problem 5.2.4 (Satisficsing problem)** Let us give parameters  $\gamma_1$  and  $\gamma_2$  a priori. Find  $F(s) \in \mathcal{F}$  and  $K(s) \in \mathcal{K}$  which satisfy

$$\left\| T_{z_1r} \left( F(s), K(s) \right) \right\|_{\infty} < \gamma_1$$
$$\left\| T_{z_2d}^{z_2} \left( K(s) \right) \right\|_{\infty} < \gamma_2.$$

**Remark 5.2.5** Problem 5.2.4 is a satisficing problem and this problem reduces to Problem 5.2.3 if the parameter  $\gamma_1$  is minimized. In this formulation, by adjusting the parameters  $\gamma_1$  and  $\gamma_2$ , the trade-off between the feedback performances and the feedforward performance can be taken into account.

**Remark 5.2.6** Problem 5.2.4 is hard to solve directly. Hence we consider solving the stabilizing controller (F(s), K(s)) that satisfies

$$\left\| \begin{array}{c} \frac{\gamma}{\gamma_1} T_{z_1 r} \left( F(s), K(s) \right) \\ \frac{\gamma}{\gamma_2} T_{z_3 d}^{z_2} \left( K(s) \right) \end{array} \right\|_{\infty} < \gamma.$$

If the controller to this problem is obtained, then it is assured that the controller is a solution for Problem 5.2.4.

Thus Problem 5.2.2, Problem 5.2.3 and Problem 5.2.4 aim at deriving the low-order TDF controller in the sense of sharing common dynamics. In these problems, the objective function  $||T_{z_1r}(F(s), K(s))||_{\infty}$  is minimized with the constraint such that F(s) and K(s) have common A-matrix.

On the other hand, as seen in chapter 4, if the constraint is not imposed on the controllers, the objective function can be represented as

$$T_{z_1r}(F(s), K(s)) = W_{\rho}(s) (W_M(s) - T_{yr}(F(s), K(s)))$$
  
=  $W_{\rho}(s) (W_M(s) - N(s)R(s)),$  (5.5)

where  $N(s) \in \mathcal{RH}_{\infty}$  is a coprime factor in the coprime factorization of  $P(s) = N(s)M^{-1}(s)$ , and  $R(s) \in \mathcal{RH}_{\infty}$  is a free parameter. Then, it can be seen that  $T_{z_1r}(F(s), K(s))$  is a function of R(s), which is a design parameter of F(s). This indicates that the constraint which is related to the feedback performances doesn't affect the objective function (5.5) if the controllers F(s) and K(s) are not correlated with each other. Hence, solving Problem 5.2.1 accomplishes the derivation of the highest performance-TDF controller at the sacrifice of simplicity of the controller.

#### 5.3 Low-order TDF controller design

In order to design the low-order controller that holds better performances, we must take into account the selection of the problem formulation. It is the point of the discussion in the previous section. In this section, we focus on a point such that " How to obtain the low-order solution to the problem which we have selected ".

#### 5.3.1 Control specification and construction of the generalized plant

Firstly, we describe the specifications of the TDF control system. The feedback performances are specified with making the norms of the sensitivity function and the co-sensitivity function small. The tracking performance is specified with making the norm of the error between the control output and a desirable response of the TDF system small. Let us consider the generalized plant shown in Figure 5.3, where  $W_S(s), W_T(s)$  and  $\rho, \eta$  are the weighting functions. In order to evaluate the feedback performances, the controlled variables  $z_1$  and  $z_2$ are selected as follows

$$z_1 = W_S(s)\tilde{S}(s)w$$
$$z_2 = W_T(s)\tilde{T}(s)w,$$

where  $\tilde{S}(s)$  and  $\tilde{T}(s)$  are defined as:

$$\tilde{S}(s) \triangleq S(s) (I + P(s) (K(s) - F(s)))$$

$$\tilde{T}(s) \triangleq T(s)K^{-1}(s)F(s),$$

and S(s) and T(s) are represented as follows

$$S(s) = (I + P(s)K(s))^{-1}$$
$$T(s) = (I + P(s)K(s))^{-1}P(s)K(s)$$



Figure 5.3: A generalized plant

Here it should be noted that  $\tilde{S}(s)$  and  $\tilde{T}(s)$  satisfy the equation

$$S(j\omega) + T(j\omega) = I, \quad \forall \omega$$

hence their norm cannot be reduced in the same frequency range. By introducing the weights  $W_S(s)$  and  $W_T(s)$  into the controlled variable as (5.6) and (5.6) the trade-off between S(s) and T(s) is taken into account.

On the other hand, the controlled variables  $z_3$  and  $z_4$  are represented as follows

$$z_3 = \rho \left( W_M(s) - \tilde{T}(s) \right) w$$
  
$$z_4 = \eta \hat{S}(s) F(s) w,$$

where

$$\hat{S}(s) \triangleq (I + K(s)P(s))^{-1}$$

and  $W_M(s)$  is a transfer function of an ideal tracking performance. These are adopted to specify the tracking performance such that the controlled output y tracks the filtered reference input, which is a desirable and feasible closed loop response under the restriction of control input energy.

In order to introduce integrators into the controller the weighting function  $W_S(s)$  is chosen to have the integrators. Since all the modes of weighting function  $W_S(s)$  are unobservable from the measured output y, and the assumption A1-2) is not satisfied in the generalized plant described in Figure 5.3. As we have discussed in chapter 3, the states of the weighting function are supposed to be observed to satisfy the assumption A1-2). This means that the controller includes the observed mode of the weighting function. Thus, the generalized



Figure 5.4: A generalized plant

plant described in Figure 5.4 is constructed and the controlled variables  $z_i$ , (i = 1, 2, 3, 4) are evaluated simultaneously. By tuning the weighting functions it is possible to consider the trade-off between those specifications. In the generalized plant the  $H_{\infty}$  control problem is formulated such that: find all the internally stabilizing controllers that satisfy

$$\|W_{S}(s)\tilde{S}(s)\|_{\infty} < 1 \|W_{T}(s)\tilde{T}(s)\|_{\infty} < 1 \|\rho\left(W_{M}(s) - \tilde{T}(s)\right)\|_{\infty} < 1 \|\eta\hat{S}(s)F(s)\|_{\infty} < 1$$
(5.6)

simultaneously. However, this problem is difficult to solve directly, hence let us consider the following criterion alternatively.

$$\left\| \begin{array}{c} W_{S}(s)S(s) \\ W_{T}(s)\tilde{T}(s) \\ \rho\left(W_{M}(s) - \tilde{T}(s)\right) \\ \eta \hat{S}(s)F(s) \end{array} \right\|_{\infty} < 1$$

$$(5.7)$$

If the inequality in (5.7) is satisfied then all the conditions in (5.6) is satisfied.

#### 5.3.2 State-space representation of the generalized plant

Now, let us define the state-space realizations of transfer functions in Figure 5.4, and represent the generalized plant with the state-space form.

#### 5.3. LOW-ORDER TDF CONTROLLER DESIGN

The real plant P(s) is assumed to be a system of  $n_p$ -dimensional, *m*-input and *p*-output system, and is represented as follows:

$$P(s) \triangleq \left[ \begin{array}{c|c} A_P & B_P \\ \hline C_P & O \end{array} \right], A_P \in \mathbb{R}^{n_p \times n_p}, B_P \in \mathbb{R}^{n_p \times m}, C_P \in \mathbb{R}^{p \times n_p}, \tag{5.8}$$

where we assume without loss of generality that the p outputs of the system are linearly independent – or equivalently that the matrix  $C_P$  is of full row rank, and that  $(A_P, B_P, C_P)$ is controllable and observable.

The realizations of weighting functions are represented as follows:

$$\begin{bmatrix} -\frac{W_S(s)}{\tilde{W}_S(s)} \end{bmatrix} \triangleq \begin{bmatrix} \frac{A_W}{C_W} & B_W \\ \hline C_W & O \\ \hline \alpha I_{n_w} & O \\ \hline O \end{bmatrix}, A_W \in \mathbb{R}^{n_w \times n_w}$$
$$W_M(s) \triangleq \begin{bmatrix} \frac{A_M}{C_M} & B_M \\ \hline C_M & O \end{bmatrix}, A_M \in \mathbb{R}^{n_m \times n_m}$$
$$W_T(s)P(s) \triangleq \begin{bmatrix} \frac{A_P}{C_T} & B_P \\ \hline C_T & D_T \end{bmatrix},$$

where the parameter  $\alpha$  is defined as follows:

$$\alpha \triangleq \begin{cases} 1, & \exists i \in \mathbb{Z}, \operatorname{Re}(\lambda_i(A_W)) = 0\\ 0, & \forall i \in \mathbb{Z}, \operatorname{Re}(\lambda_i(A_W)) < 0 \end{cases}$$

Then the generalized plant G(s) can be represented as follows.

$$G(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix} = \begin{bmatrix} A_W & -B_W C_P & O & B_W & O \\ O & A_P & O & O & B_P \\ \hline O & O & A_M & B_M & O \\ \hline C_W & O & O & O & O \\ O & C_T & O & O & D_T \\ O & -\rho C_P & \rho C_M & O & O \\ \hline O & C_P & O & O & O \\ O & O & O & I_p & O \end{bmatrix}$$
(5.9)

The criterion in inequality (5.7) is equal to the following inequality:

$$\left\|\mathcal{F}_l\left(G(s), K_{\infty}(s)\right)\right\|_{\infty} < 1.$$
(5.10)

Since, in this generalized plant G(s),  $D_{21}$  is column full rank and the assumptions A1 and A3 are satisfied, on the other hand, the assumption **A2-2**) is not satisfied, the problem of finding the internally stabilizing controller which satisfies the inequality (5.10) is the aforementioned non-standard  $H_{\infty}$  control problem of case 2.

#### 5.3.3 Parametrization of all $H_{\infty}$ controllers

Firstly, let us introduce a change of the coordinates in the system P(s) represented with the realization in (5.8). Since  $C_P$  is assumed to be a matrix of full row rank, by using a nonsingular matrix:

$$T \triangleq \left[ \begin{array}{c} C_P \\ (C_P^{\perp}) \end{array} \right]$$

the coordinates can be transformed, and then the state-space realization of P(s) is represented as follows:

$$\begin{bmatrix} \overline{A}_P & \overline{B}_P \\ \overline{C}_P & O \end{bmatrix} \triangleq \begin{bmatrix} T\overline{A}_P T^{-1} & T\overline{B}_P \\ \overline{C}_P T^{-1} & O \end{bmatrix}$$
$$= \begin{bmatrix} A_{11} & A_{12} & B_{P_1} \\ A_{21} & A_{22} & B_{P_2} \\ \overline{I_P} & O & O \end{bmatrix}.$$

Hence, we assume without generality that the parameters  $A_P$  and  $C_P$  are in the form

$$\begin{bmatrix} A_P \\ \hline C_P \end{bmatrix} = \begin{bmatrix} -A_{11} & A_{12} \\ \hline A_{21} & A_{22} \\ \hline I_p & O \end{bmatrix}, A_{11} \in \mathbb{R}^{p \times p}, A_{22} \in \mathbb{R}^{(n_p - p) \times (n_p - p)},$$
(5.11)

where it is well known that  $(A_{22}, A_{12})$  is observable [42].

**Lemma 5.3.1** In (5.11), it is assumed without loss of generality that real parts of all the eigenvalues of  $A_{22}$  are negative.

**Proof.** Since  $(A_{22}, A_{12})$  is observable, there exists a nonsingular matrix:

$$J = \left[ \begin{array}{cc} I_p & O\\ L & I_{(n_p - p)} \end{array} \right]$$

such that J satisfies

$$\begin{bmatrix} J^{-1}A_PJ\\ -\overline{C}_PJ \end{bmatrix} = \begin{bmatrix} A_{11} + A_{12}L & A_{12}\\ -LA_{11} - LA_{12}L + A_{21} + A_{22}L & -LA_{12} + A_{22}\\ \hline I_p & O \end{bmatrix},$$

where  $\operatorname{Re}(\lambda_i(-LA_{12}+A_{22})) < 0, \forall i = 1, 2, \dots, n_p - p$ . Hence, in the form of (5.11), it is assumed that real parts of all the eigenvalues of  $A_{22}$  are negative.

From the definition in section 2.3,  $D_{21}^{\dagger}, D_{21}^{\perp}$  can be given as follows.

$$D_{21}^{\dagger} = \begin{bmatrix} O & O & I_p \end{bmatrix}, D_{21}^{\perp} = \begin{bmatrix} I_{n_w} & O & O \\ O & I_p & O \end{bmatrix}$$

Then,  $A - B_1 D_{21}^{\dagger} C_2$  and  $D_{21}^{\perp} C_2$  can be represented as follows.

$$\begin{bmatrix} A - B_1 D_{21}^{\dagger} C_2 \\ - \overline{D_{21}} \overline{C_2}^{\dagger} \end{bmatrix} = \begin{bmatrix} A_W & -B_W C_P & O \\ O & A_P & O \\ O & O & A_M \\ - \overline{\alpha} \overline{I_{n_w}} & \overline{O} & \overline{O} \\ O & C_P & O \end{bmatrix}$$

Let us select  $L_H$  such that the observable subspace of the pair  $(A - B_1 D_{21}^{\dagger} C_2, D_{21}^{\perp} C_2)$  is stabilized by  $D_{21}^{\perp} C_2$ , i. e., let us select  $L_H$ :

$$L_{H} = \begin{bmatrix} h_{1} & B_{W} \\ O & h_{2} \\ O & -A_{21} \\ l_{1} & l_{2} \end{bmatrix}$$

such that

$$\operatorname{Re}(\lambda_i(A_W + \alpha h_1)) < 0, \forall i = 1, 2, \dots, n_w, \operatorname{Re}(\lambda_i(A_{11} + h_2)) < 0, \forall i = 1, 2, \dots, p \quad (5.12)$$

are satisfied, and  $l_1 \in \mathbb{R}^{n_m \times n_w}, l_2 \in \mathbb{R}^{n_w \times p}$  are arbitrary matrices.

Then,  $A_{ZH}$  is represented as follows

where  $\mathbf{Re}(\lambda_i(A_{22})) < 0, \forall i = 1, 2, ..., n_p - p.$ 

Based on the above preparation, the  $H_{\infty}$  controller for the generalized plant (5.9) can be obtained from the following lemma.

**Lemma 5.3.2** The  $H_{\infty}$  problem for G(s) in (5.9) is solvable if and only if both of the AREs:

$$XA_{ZF} + A_{ZF}^T X + XRX + (D_{12}^{\perp}C_1)^T D_{12}^{\perp}C_1 = 0$$
(5.14)

$$YA_{ZH}^{T} + A_{ZH}Y + Y\left\{C_{1}^{T}C_{1} - (D_{21}^{\dagger}C_{2})^{T}D_{21}^{\dagger}C_{2}\right\}Y = O,$$
(5.15)

where  $A_{ZF}$  and R are defined as :

$$A_{ZF} \triangleq A - B_2 D_{12}^{\dagger} C_1 R \triangleq B_1 B_1^T - B_2 D_{12}^{\dagger} (B_2 D_{12}^{\dagger})^T,$$

and  $A_{ZH}$  is represented in (5.13), have stabilizing solutions  $X \ge O$  and  $Y \ge O$ , and which satisfy  $\rho(XY) < 1$ . If the solvability condition is satisfied, the class of the  $H_{\infty}$  controllers are parametrized as follows:

$$K_{\infty}(s) = \mathcal{F}_{l} \left( \begin{bmatrix} A_{Y} + \hat{B}_{2}F_{\infty} & H_{\infty} & -\hat{B}_{2}\Sigma \\ \hline -F_{\infty} & O & \Sigma \\ -D_{21}^{\dagger}\hat{C}_{2}Z & D_{21}^{\dagger} & O \\ D_{21}^{\pm}\hat{C}_{2}Z & D_{21}^{\pm} & O \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix} \right),$$
(5.16)

where

$$\begin{split} A_{Y} &= A + YC_{1}^{T}C_{1} + H_{\infty}C_{2} \\ \hat{B}_{2} &= B_{2} + YC_{1}^{T}D_{12} \\ \hat{C}_{2} &= D_{21}B_{1}^{T}X + C_{2} \\ F_{\infty} &= \left\{ -D_{12}^{\dagger}C_{1} - D_{12}^{\dagger} \left(B_{2}D_{12}^{\dagger}\right)^{T}X \right\} Z \\ H_{\infty} &= -B_{1}D_{21}^{\dagger} - Y \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger} + L_{H}D_{21}^{\perp} \\ Z &= (I - YX)^{-1} \\ \Sigma &= \left(D_{12}^{T}D_{12}\right)^{-\frac{1}{2}}, \end{split}$$

and  $N(s) \in \mathcal{BH}_{\infty}$  and  $W(s) \in \mathcal{RH}_{\infty}$  are the free parameters.

**Proof.** By applying the result of Theorem 2.4.2 for G(s) in (5.9), the result is immediately obtained.

**Remark 5.3.1** The McMillan degree of the central solution, which is a controller given with the free parameters are set to zero, is equal to the degree of the generalized plant, and is  $n = n_w + n_p + n_m$ .

**Remark 5.3.2** It should be also noted that the class of the non-standard  $H_{\infty}$  controllers is represented with two free parameters  $N(s) \in \mathcal{BH}_{\infty}$  and  $W(s) \in \mathcal{RH}_{\infty}$ .

Continuously, we review the  $H_{\infty}$  controller given in Lemma 5.3.2 as an integral-type TDF controller. In this paragraph it is assumed that the  $H_{\infty}$  controller implies the central solution. From Figure 5.4, let us extract the closed loop system composed of  $P(s), K_{\infty}(s)$ and  $\tilde{W}_S(s)$ . The closed loop system is described in Figure 5.5. Here, it is assumed that  $\tilde{W}_S(s)$  has integrators, hence the parameter  $\alpha$  is set to 1. If we denote  $K_{\infty}(s)$  according to the dimensions of  $y_0, y_1$  and  $y_2$  as

$$K_{\infty}(s) = \begin{bmatrix} K_1(s) & K_2(s) & K_3(s) \end{bmatrix}$$

the control input variable u can be represented as follows

$$u = K_{\infty}(s) \begin{bmatrix} y_0 \\ y_1 \\ y_2 \end{bmatrix}$$
$$= \begin{bmatrix} K_1(s) & K_2(s) & K_3(s) \end{bmatrix} \begin{bmatrix} \tilde{W}_S(s) (r-y) \\ y \\ r \end{bmatrix}$$
$$= \begin{bmatrix} F(s) & K(s) \end{bmatrix} \begin{bmatrix} r \\ y \end{bmatrix},$$



Figure 5.5: Interpretation of the closed loop system as a TDF control system

where F(s) and K(s) are defined as follows

$$F(s) \triangleq K_1(s)W_S(s) + K_3(s)$$
$$K(s) \triangleq K_2(s) - K_1(s)\tilde{W}_S(s).$$

Thus the TDF controller which satisfies prescribed  $H_{\infty}$  performances can be represented. It should be noted that the TDF controller is integral-type. It can be seen that the McMillan degree of the TDF controller is

$$\deg(F(s), K(s)) = 2n_w + n_p + n_m.$$
(5.17)

Compared with the order of the TDF controller evaluated in section 4.2.2, in this synthesis the TDF controller which is also an integral-type can be designed with lower order.

#### 5.3.4 Reduced-order TDF controller design

As we have considered in chapter 2, it is possible to derive the reduced-order  $H_{\infty}$  controller in the non-standard problem, where the dimension of the observed output is greater than that of the disturbance input. This subsection derives the reduced-order TDF controller based on the way that we presented in chapter 2.

The following lemma is useful for later discussion.

**Lemma 5.3.3** Assume that the ARE in (5.15) has a positive semi-definite stabilizing solution. Then the stabilizing solution of the ARE can be represented as follows:

where  $Y_r \ge O$  is a stabilizing solution of the reduced-order ARE:

$$Y_r A_M^T + A_M Y_r + \rho^2 Y_r C_M^T C_M Y_r = 0. (5.19)$$

**Proof.** The ARE in (5.15) can be represented as follows

$$YA_Y^T + A_{ZH}Y = O, (5.20)$$

where

$$A_{Y} = A_{ZH} + Y \left\{ C_{1}^{T} C_{1} - \left( D_{21}^{\dagger} C_{2} \right)^{T} D_{21}^{\dagger} C_{2} \right\} \in \mathbb{R}$$

is a stable matrix. Let U be a row-basis of the stable subspace of  $A_{ZH}$ . Then the matrix U can be selected as

$$U = \begin{bmatrix} I_{n_w} & O & O & O \\ O & I_p & O & O \\ O & O & I_{n_p-p} & O \end{bmatrix}.$$

Then the matrix U satisfies the following equation:

$$UA_{ZH} = \Lambda U,$$

where

$$\Lambda \triangleq \begin{bmatrix} A_W + \alpha h_1 & O & O \\ O & A_{11} + h_2 & A_{12} \\ O & O & A_{22} \end{bmatrix} \in \mathbb{R}^{(n_w + n_p) \times (n_w + n_p)}$$

is a stable matrix. Pre-multiplying the equation (5.20) by U, the following equation is obtained

$$(UY) A_Y^T + \Lambda (UY) = O.$$

Since  $\operatorname{\mathbf{Re}}(\lambda_i(\Lambda)) < 0$  and  $\operatorname{\mathbf{Re}}(\lambda_i(A_Y)) < 0$  hold, the solution  $Y \ge O$  satisfies UY = O and Y must be in the form (5.18). By substituting  $Y \ge O$  in (5.18) into the ARE in (5.15),  $Y_r \ge O$  must be a solution of the reduced-order ARE in (5.19).

Let us represent the controller given in (5.16) with DHMT as follows

$$K_{\infty}(s) = \mathcal{DHM}\left( \begin{bmatrix} A_{Y} & -\hat{B}_{2} & -H_{\infty} \\ \hline \Sigma^{-1}F_{\infty} & \Sigma^{-1} & O \\ -D_{21}^{\dagger}\hat{C}_{2}Z & O & D_{21}^{\dagger} \\ -D_{21}^{\dagger}C_{2} & O & D_{21}^{\dagger} \end{bmatrix}, \begin{bmatrix} N(s) & W(s) \end{bmatrix} \right)$$
$$= \begin{bmatrix} A_{Y} & -\hat{B}_{2} \\ \hline C_{K}(s) & \Sigma^{-1} \end{bmatrix}^{-1} \begin{bmatrix} A_{Y} & -H_{\infty} \\ \hline C_{K}(s) & N(s)D_{21}^{\dagger} + W(s)D_{21}^{\bot} \end{bmatrix},$$

where  $C_K(s)$  is given as follows

$$C_K(s) \triangleq \Sigma^{-1} F_{\infty} - N(s) D_{21}^{\dagger} \hat{C}_2 Z - W(s) D_{21}^{\perp} C_2.$$
(5.21)

By using the result of Lemma 5.3.3 the matrix  $A_Y$  can be represented as follows

$$A_{Y} = \begin{bmatrix} A_{W} + \alpha h_{1} & O & O & O \\ O & A_{11} + h_{2} & A_{12} & O \\ O & O & A_{22} & O \\ \alpha l_{1} & l_{2} - \rho^{2} Y_{r} C_{M}^{T} & O & A_{M} + \rho^{2} Y_{r} C_{M}^{T} C_{M} \end{bmatrix},$$
 (5.22)

where  $Y_r \geq O$  is a stabilizing solution of the reduced-order ARE in (5.19), and  $l_1 \in \mathbb{R}^{n_m \times n_w}$ and  $l_2 \in \mathbb{R}^{n_w \times p}$  are free parameters. In the equation (5.21),  $C_K(s)$  is explicitly represented with the free parameter  $W(s) \in \mathcal{RH}_{\infty}$ . Hence an adequate selection of the free-parameters of  $l_1, l_2$  and W(s) yields pole-zero cancellations in the controller, and the order of the controller can be reduced.

**Theorem 5.3.1** Under the same solvability condition as in Lemma 5.3.2, the class of reduced order  $H_{\infty}$  controllers is parametrized as follows:

$$K_{\infty}^{r}(s) = \mathcal{DHM}\left(\begin{bmatrix} \frac{\tilde{A}_{Y} & -\hat{B}_{P_{2}} & -H_{\infty_{2}} \\ \hline \Sigma^{-1}F_{\infty_{2}} & \Sigma^{-1} & \Sigma^{-1}\hat{F}_{\infty_{1}}D_{21}^{\perp} \\ -C_{D_{2}} & O & D_{21}^{\dagger} - \hat{C}_{D_{1}}D_{21}^{\perp} \end{bmatrix}, N(s)\right),$$
(5.23)

where  $N(s) \in \mathcal{BH}_{\infty}$  is a free parameter, and  $\tilde{A}_Y, \hat{B}_{P_2}, H_{\infty_2}, \hat{F}_{\infty_1}, F_{\infty_2}, \hat{C}_{D_1}, C_{D_2}$  are defined as follows:

$$\begin{split} \tilde{A}_Y &= \begin{bmatrix} A_{22} & O \\ O & A_M + \rho^2 Y_r C_M^T C_M \end{bmatrix} \in \mathbb{R}^{(n - (n_w + p)) \times (n - (n_w + p)))} \\ \hat{B}_{P_2} &= \begin{bmatrix} B_{P_2} \\ O \end{bmatrix} \in \mathbb{R}^{(n - (n_w + p)) \times m} \\ H_{\infty_2} &= \begin{bmatrix} O & -A_{21} & O \\ O & \rho^2 Y_r C_M^T & -B_M \end{bmatrix} \\ \hat{F}_{\infty_1} &= F_{\infty} \begin{bmatrix} \alpha^{-1} I_{n_w} & O \\ O & I_p \\ O & O \end{bmatrix} \\ F_{\infty_2} &= F_{\infty} \begin{bmatrix} O \\ I_{n - (n_w + p)} \end{bmatrix} \\ \hat{C}_{D_1} &= D_{21}^{\dagger} \hat{C}_2 Z \begin{bmatrix} O \\ I_n \\ O \end{bmatrix} \\ C_{D_2} &= D_{21}^{\dagger} \hat{C}_2 Z \begin{bmatrix} O \\ I_{n - (n_w + p)} \end{bmatrix}. \end{split}$$

**Proof.** By setting the free parameters as

$$\begin{cases} l_1 = O\\ l_2 = \rho^2 Y_r C_M^T\\ W(s) = \left(\Sigma^{-1} F_\infty - N(s) D_{21}^{\dagger} \hat{C}_2 Z\right) V \end{cases},$$

where V is a full column rank matrix defined as

$$V \triangleq \begin{bmatrix} \alpha^{-1}I_{n_w} & O \\ O & I_p \\ O & O \\ O & O \end{bmatrix} \in \mathbb{R}^{(n_p + n_w + n_m) \times (n_w + p)},$$

the matrices  $A_Y$  and  $C_K(s)$  satisfy

$$\begin{cases} A_Y V = V \begin{bmatrix} A_W + \alpha h_1 & O \\ O & A_{11} + h_2 \end{bmatrix} \\ C_K(s) V = O \end{cases}$$

Hence the pair  $(A_Y, C_K(s))$  is unobservable, and the order of the controller is reduced by the dimension of rank  $(V) = n_w + p$ . Then the reduced order  $H_\infty$  controller is derived.

**Remark 5.3.3** The McMillan degree of the central solution of the  $H_{\infty}$  controller in (5.16) is reduced by the dimension of rank  $(V) = n_w + p$ , and is equal to  $n_p - p + n_m$ .

**Remark 5.3.4** It should be also noted that the class of the reduced-order  $H_{\infty}$  controllers is represented with free parameter  $N(s) \in \mathcal{BH}_{\infty}$ .

**Remark 5.3.5** In this theorem the reduced order TDF controller is represented with DHMT, and this implies that the controller is in a form of coprime factorization over  $\mathcal{RH}_{\infty}$ . This form is useful for reducing the order of the controller by approximation, because no model approximation methods can be used directly for unstable system.

We review the reduced-order  $H_{\infty}$  controller given in Theorem 5.3.1 as an integral-type TDF controller. In this paragraph it is assumed that the  $H_{\infty}$  controller implies the central solution. As we have discussed in the previous section, let us extract the closed loop system composed of  $P(s), K_{\infty}^{r}(s)$  and  $\tilde{W}_{S}(s)$ . The closed loop system is shown in Figure 5.6. If we denote  $K_{\infty}^{r}(s)$  according to the dimensions of  $y_{0}, y_{1}$  and  $y_{2}$  as

$$K_{\infty}^{r}(s) = \begin{bmatrix} K_{1}^{r}(s) & K_{2}^{r}(s) & K_{3}^{r}(s) \end{bmatrix}$$

the control input variable u can be represented as follows

$$u = K_{\infty}^{r}(s) \begin{bmatrix} y_{0} \\ y_{1} \\ y_{2} \end{bmatrix}$$
$$= \begin{bmatrix} K_{1}^{r}(s) & K_{2}^{r}(s) & K_{3}^{r}(s) \end{bmatrix} \begin{bmatrix} \tilde{W}_{S}(s) (r-y) \\ y \\ r \end{bmatrix}$$
$$= \begin{bmatrix} F^{r}(s) & K^{r}(s) \end{bmatrix} \begin{bmatrix} r \\ y \end{bmatrix},$$



Figure 5.6: Interpretation of the closed loop system as a TDF control system

where  $F^{r}(s)$  and  $K^{r}(s)$  are defined as follows

$$F^{r}(s) \triangleq K_{1}^{r}(s)\tilde{W}_{S}(s) + K_{3}^{r}(s)$$
$$K^{r}(s) \triangleq K_{2}^{r}(s) - K_{1}^{r}(s)\tilde{W}_{S}(s).$$

Thus the reduced-order TDF controller which satisfies prescribed  $H_{\infty}$  performances can be represented. It should be noted that the reduced-order TDF controller is integral-type. It can be seen that the McMillan degree of the reduced-order TDF controller is

$$\deg(F^{r}(s), K^{r}(s)) = n_{w} + n_{p} + n_{m} - p.$$
(5.24)

Compared with the order of the TDF controller evaluated in section 4.2.2, in this synthesis the TDF controller which is also integral type can be designed with further lower order. It is also apparent that

$$\deg\left(F(s), K(s)\right) - \deg\left(F^{r}(s), K^{r}(s)\right) = n_{w} + p$$

holds. This order reduction is a benefit which is yielded by the formulation of the nonstandard  $H_{\infty}$  control.

#### 5.4 A numerical example

#### 5.4.1 Magnetic levitation system

This section examines the effectiveness of the proposed method by numerical examples. We use the model of the magnetic levitation system that we have used in chapter 2. Let us consider the system shown in the Figure 2.6. The purpose of controlling the magnetic levitation system in this chapter is to stabilize the system regardless of variation of the mass, and



Figure 5.7: Steel balls

besides, to obtain a good tracking property with an integral-type TDF controller. The order of the controller should be obtained with lower order.

#### 5.4.2 Description of perturbed models

In the linear model of equation (2.94), three types of steel balls are assumed. Mass of each steel ball is set as

$$M_1 = 0.6M, \quad M, \quad M_2 = 1.7M.$$

Then the transfer function of each model is obtained as

$$P_1(s) = \frac{-111.71}{(s-61.6)(s+14.8)(s+60.1)}, \text{ where mass is } M_1;$$

$$P(s) = \frac{-67.03}{(s-47.8)(s+46.3)(s+14.9)}, \text{ where mass is } M;$$

$$P_2(s) = \frac{-39.4}{(s-36.8)(s+15.0)(s+35.1)}, \text{ where mass is } M_2.$$

We call P(s) nominal model,  $P_1(s)$  and  $P_2(s)$  perturbed models.

Our purpose of controlling the magnetic levitation system is to stabilize the system regardless of variation of mass with one controller. For this purpose we consider to design a robust controller which stabilizes any system in

$$\mathcal{P}_{real} \triangleq \{ P(s; \alpha M) \mid \alpha = [0.6, 1.7] \}.$$
(5.25)

For this system we set a boundary of the perturbation such that a partial system from the input u to the output  $y_3$  varies in a set

$$\mathcal{P} = \{ (1 + \Delta(s)) P(s) : |\Delta(j\omega)| \le |W_T(j\omega)|, \forall \omega \}, \qquad (5.26)$$

where  $W_T(s)$  is a function of the relative error bound between the perturbed model and P(s). In this thesis, by trial and error, we select  $W_T(s)$  as

$$W_T(s) = 0.6 \times 10^{-10} (10^{-2}s^3 + s^2 + 10^8 s + 1),$$

such that the plant set  $\mathcal{P}$  includes set of perturbed plants  $\mathcal{P}_{real}$ . Figure 5.8 plots the gain of the weighting function  $W_T(s)P(s)$  with violet line, variations of the models  $P_1(s)$  and  $P_2(s)$  from the nominal model P(s) with lines of blue and red. Here, it can be verified that the gain of the boundary is not less than that of any variation.



Figure 5.8: Gain plots of perturbations and weighting function

#### 5.4.3 Designing of low-order integral-type TDF controller

We design an integral-type TDF controller based on the method that is proposed in section 5.3. The generalized plant of equation (5.9) is made by using the weighting functions:

$$W_S(s) = \frac{32700}{s(s+1000)} \tag{5.27}$$

$$W_T(s) = 0.6 \times 10^{-10} (10^{-2}s^3 + s^2 + 10^8 s + 1)$$
(5.28)

$$W_M(s) = \frac{130^2}{s^2 + 247s + 130^2}$$
(5.29)

$$\eta = 1.5 \times 10^{-4} \tag{5.30}$$

$$\rho = 0.2.$$
(5.31)

Then 7th order generalized plant is obtained. For the generalized plant, we first obtained a 7th order  $H_{\infty}$  controller whose order is same as that of the generalized plant. By selecting an adequate free parameter W(s) we obtained a 4th order  $H_{\infty}$  controller, and converted the controller into an integral-type. Then we got an integral-type TDF controller of order 6. The frequency response of the TDF controller is shown in Figure 5.9-(a). From the figure it can be seen that the feedback controller and the feedforward controller share common dynamics and they include an integrator. The frequency responses of sensitivity function S(s) and cosensitivity function T(s) are shown in Figure 5.9-(b). It can be seen that the reduced-order controller satisfies the feedback performances such that

$$\begin{cases} \|W_S(s)S(s)\|_{\infty} < 1\\ \|W_T(s)T(s)\|_{\infty} < 1 \end{cases}$$



Figure 5.9: Frequency responses of (a) 6th order TDF controller, and (b) closed loop transfer functions.



Figure 5.10: Step responses of (a) a feedback control system composed of the nominal plant and the 6th order feedback controller, and (b) a TDF control system composed of the nominal plant and the 6th order TDF controller.



Figure 5.11: Step responses of the perturbed systems

Figure 5.10 illustrates step responses of closed loop systems: (a) a feedback control system composed of the nominal plant and the 6th order feedback controller, (b) a TDF control system composed of the nominal plant and the 6th order TDF controller. From this figure, it can be seen that the tracking performance can be improved by adding the feedforward controller to the feedback control system. In order to verify the robustness of the TDF system, step responses of the closed loop systems, in which the perturbed plants are used, are illustrated in Figure 5.11; in (a) the nominal model is used, in (b) a perturbed model  $P_1(s)$  is used, in (c) a perturbed model  $P_2(s)$  is used. It can be seen that the closed loop system holds stability within the prescribed variation of the plant.

#### 5.4.4 Comparison with another method

To evaluate the performances of the TDF controller that is obtained with the proposed method, other TDF controllers are also designed with the other method, which is an independent sequential approach in which the feedback controller and the feedforward controller are respectively designed with the standard  $H_{\infty}$  control. First, a feedback controller is obtained by solving the mixed sensitivity problem. The generalized plant of the problem is illustrated in Figure 3.2. Two types of sensitivity weighting functions are adopted. One of the sensitivity functions is type of non-optimal sensitivity function, where the gain of the weight  $W_S(s)$  is not so high that there is room for improving the robustness. Here, the weight is used as the following one.

$$W_{S_1}(s) = \frac{4110}{s(s+1000)}$$

We call a mixed sensitivity problem in which  $W_{S_1}(s)$  is adopted type1 problem. The other type of the problem is with sensitivity function of maximal gain. The weight is used as

$$W_{S_2}(s) = \frac{482}{s(s+1.0)},$$

which has higher gain than  $W_{S_1}(s)$  has in low frequency range. We call this type of problem type 2 problem. On the other hand, the weighting function  $W_T(s)$  is used the same one that is used in equation (5.28) in each type problem. Since the order of each generalized plant is five, the order of each resultant feedback controller is also five. The feedback properties: frequency responses of the sensitivity function and co-sensitivity function, which are obtained in those types of problems, are illustrated in Figure 5.12. In (a), the result of the type 1 problem is plotted, in (b) the result of the type 2 problem is plotted.

Second, using each feedback controller, a feedforward controller is designed by solving the model matching problem whose generalized plant is illustrated in Figure 5.13. The weighting functions that are used in the generalized plant are as follows

$$W_M(s) = \frac{130^2}{s^2 + 247s + 130^2}$$
$$W_\rho(s) = \frac{2.0}{s}$$
$$\eta = 4.0 \times 10^{-5}.$$

Since the order of the generalized plant is 11, an 11th order feedforward controller is obtained in each type of problem. Hence, each resultant TDF controller is 16th order, which is higher by 10th order than that of the TDF controller designed with the proposed method. Then, by using the balanced truncation method [1, 52], each 16th order controller is approximated with lower order TDF controllers. Here, in order to preserve the steady-state property of the feedforward controller F(s), the truncated feedforward controller  $\tilde{F}_r(s)$  is scaled as follows.

$$F_r(s) = \tilde{F}_r(s) \times \frac{F(0)}{\tilde{F}_r(0)}$$

The approximated TDF controllers have variation according to the combination of orders of feedback controller and feedforward controller.



Figure 5.12: The feedback properties of the controllers obtained by solving the problems: (a) Type 1 problem, (b) Type 2 problem. Green line :  $|S(j\omega)|$ ; Blue line :  $|T(j\omega)|$ ; Red  $\times$  :  $|W_S^{-1}(j\omega)|$ ; Magenta + :  $|W_T^{-1}(j\omega)|$ .



Figure 5.13: The generalized plant of the model matching problem



Figure 5.14: The step responses

Step response of each closed loop system, which is composed of the nominal plant and each controller derived from the type 1 problem, are plotted. In (a), a 16th order TDF controller is used. In (b), a 6th order TDF controller, which is given with a first order reduced order feedforward controller and a 5th order feedback controller, is used. In (c), a 6th order TDF controller, which is given with a second order reduced order feedforward controller and a 4th order reduced order feedback controller, is used. In (d), a 6th order TDF controller, which is given with a 3rd order reduced order feedforward controller and a 3rd order reduced order feedforward controller, is used.



Figure 5.15: The step responses

Step response of each closed loop system, which is composed of the nominal plant and each controller derived from the type 2 problem, are plotted. In (a), a 16th order TDF controller is used. In (b), a 6th order TDF controller, which is given with a first order reduced order feedforward controller and a 5th order feedback controller, is used. In (c), a 6th order TDF controller, which is given with a second order reduced order feedforward controller and a 4th order reduced order feedback controller, is used. In (d), a 7th order TDF controller, which is given with a second order reduced order feedforward controller, which is given with a second order reduced order feedforward controller, which is given with a second order reduced order feedforward controller, and a 5th order reduced order feedforward controller, which is given with a second order reduced order feedforward controller and a 5th order reduced order feedforward controller, and a 5th order reduced order feedforward controller, and a 5th order reduced order feedforward controller and a 5th order reduced order feedback controller, is used.

Figure 5.14 plots step responses of closed loop systems in which 16th order TDF controller and its reduced order controllers derived from the type 1 problem are adopted. Figure 5.15 plots step responses of closed loop systems in which 16th order TDF controller and its reduced order controllers derived from the type 2 problem are adopted. From these figures, it can be seen that each 16th order TDF controller has better tracking property than that of the 6th order TDF controller that is obtained with the proposed method. It is pretty natural that a higher order controller performs better than a lower order controller. On the other hand, compared with the 6th order controller obtained with the proposed method, none of the reduced order controllers perform better.

#### 5.5 Summary

In this chapter, we have considered designing a low-order TDF controller. Based on the analysis in the previous chapter, formulations of designing the low-order TDF controller are compared with each other. The result of the comparison has suggested that a method of simultaneously designing the feedback controller and the feedforward controller is the best way to derive the low-order TDF controller. Then we have chosen the way of simultaneous approach. Another reason why we have adopted the simultaneous approach is that the formulation of the simultaneous approach reduces to the non-standard  $H_{\infty}$  problem, where the way of designing the reduced-order controller has just been clarified. Thus we have proposed a method of designing the low-order TDF controller whose order is lower than the order of the generalized plant by the sum of the dimensions of the measurement output of the real plant and the order of the weighting function of the sensitivity function. In this method, the TDF controller can be designed with the integral-type by using the result of chapter 3. Effectiveness of the proposed method is verified with a numerical example of controlling the magnetic levitation system.

## Chapter 6

## $H_{\infty}$ controller approximation

#### 6.1 Introduction

In the previous chapters, we have considered the *direct* approach to deriving the low-order  $H_{\infty}$  controller. On the other hand, in this chapter we consider the *indirect* approach where an approximation method is used for reducing the order of an  $H_{\infty}$  controller.

In this chapter, we treat a specialized solution of the numerically solved  $H_{\infty}$  controller, which is represented by the linear fractional transformation of an optimized Youla parameter [69] in  $\mathcal{RH}_{\infty}$ . The numerical approach [3, 43, 54] is useful for solving the  $H_{\infty}$  problem with a constraint of specifying the closed loop poles. In many cases, the numerical approach derives an extremely high-order Youla parameter and the resultant  $H_{\infty}$  controller becomes high order. However, it is unknown how to reduce the order of the  $H_{\infty}$  controller without deviating the constraint. In this chapter, we consider the possibility of reducing the order of the  $H_{\infty}$  controller by approximating the Youla parameter.

When reducing the order of the high-order  $H_{\infty}$  controller which satisfies the constraint of the closed loop pole location, it is required to reduce the order of the controller such that the specified closed-loop property is preserved. For this requirement, in this study we firstly propose a model approximation method for a system whose poles are located in a specified domain. Since this approximation method assures that the poles of the approximated system locate in the domain in which the poles of the original system locate, it is useful to apply this method for the  $H_{\infty}$  controller reduction problem which we consider in this chapter. Thus we consider the problem of reducing the order of the  $H_{\infty}$  controller based on the approximation. One of advantages of this reduction method is that the reduced-order controller maintains the closed-loop properties of the original controller, such as internal stability and closed-loop pole specification. A sufficient condition for the closed loop transfer function satisfies the constraint of the  $H_{\infty}$  norm is derived. The effectiveness of this  $H_{\infty}$  controller reduction method is verified with a numerical example.


Figure 6.1: Closed loop system

# 6.2 A numerical approach to $H_{\infty}$ controller design

#### 6.2.1 Youla parametrization

Consider the following generalized plant G(s) which is a transfer function matrix from  $(w^T, u^T)^T$  to  $(z^T, y^T)^T$ , where vectors w, u, z and y represent the exogenous signals such as perturbation and disturbances, the control input signals from actuators or controllers, the controlled output signals to be evaluated and the measured output signals, respectively.

$$G(s) = \begin{bmatrix} G_{11}(s) & G_{12}(s) \\ G_{21}(s) & G_{22}(s) \end{bmatrix}$$

The generalized plant G(s) is composed of the plant and all weighting functions which specify the robustness and the other closed loop specifications. When the controller K(s) is connected between y and u, the closed loop transfer function from w to z is represented by the following equation.

$$T_{zw}(s) = \mathcal{F}_l\left(G(s), K(s)\right)$$

The block diagram of the closed loop system is shown in Figure 6.1.

It is known that if one controller  $K_{nom}(s)$  which stabilizes the closed loop system is given, all the controllers which stabilize the closed loop system can be represented by using a stable and proper transfer function Q(s). Assume that a plant P(s) is a minimal scalar system, and is represented with realization

$$P(s) = \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix}.$$

If we select a feedback gain F and an observer gain H such that matrices  $A_F \triangleq A + BF$  and  $A_H \triangleq A + HC$  are stable, then all of the controllers which stabilize P(s) can be represented as follows:

$$K(s) = \frac{X(s) + Q(s)M(s)}{Y(s) - Q(s)N(s)},$$
(6.1)

where  $Q(s) \in \mathcal{RH}_{\infty}$  is a free parameter, and transfer functions N(s), M(s), X(s) and Y(s)are represented as follows

$$\begin{bmatrix} -N(s)\\ -\overline{M}(s) \end{bmatrix} = \begin{bmatrix} A_F & B\\ \hline C + DF & D\\ \hline F & \hline 1 \end{bmatrix}$$
$$\begin{bmatrix} X(s) & Y(s) \end{bmatrix} = \begin{bmatrix} A_H & H & -B - HD\\ \hline F & 0 & 1 \end{bmatrix}.$$

This representation of the stabilizing controllers is called Youla parametrization. It should be noted that K(s) in (6.1) stabilizes the closed loop system composed of P(s) and K(s), if and only if  $Q(s) \in \mathcal{RH}_{\infty}$ .

In general, it is known that by using the Youla parametrization, arbitrary closed loop transfer function  $T_{zw}(s)$  can be represented with the following affine combination of stable transfer functions:

$$T_{zw}(s) = T_1(s) - T_2(s)Q(s)T_3(s) \triangleq F_{affine}(Q(s)),$$
 (6.2)

where  $T_1(s)$ ,  $T_2(s)$  and  $T_3(s)$  are certain stable transfer functions related to the plant, the weighting function and the stabilizing controller, and Q(s) represents a free parameter. The closed loop system is stable, if and only if we choose Q(s) from the set of all real rational stable functions.

For example, let's consider a problem of sensitivity minimization. In this problem the transfer function is selected as

$$T_{zw}(s) = W_S(s) \frac{1}{1 + P(s)K(s)},$$

where  $W_S(s) \in \mathcal{RH}_{\infty}$  is a weighting function. If we note the fact that equations:

$$P(s) = \frac{N(s)}{M(s)}, \quad X(s)N(s) + Y(s)M(s) = 1$$

hold, it is verified that the transfer function  $T_{zw}(s)$  can be represented as follows

$$T_{zw}(s) = T_1(s) - T_2(s)Q(s)T_3(s),$$

where

$$T_1(s) \triangleq W_S(s)M(s)Y(s)$$
$$T_2(s) \triangleq W_S(s)M(s)N(s)$$
$$T_3(s) \triangleq 1.$$

The representation of the closed loop system in (6.2) is useful for designing an  $H_{\infty}$  controller which specifies the location of the closed loop poles. Let us denote  $\mathcal{RH}_{\infty}^D$  as a class of stable proper transfer functions whose poles lie in the disk domain Dom(D). Figure 6.2



Figure 6.2: Disk domain

illustrates Dom(D). The set of closed loop poles is the union of the poles of the transfer functions  $T_1(s)$ ,  $T_2(s)$ ,  $T_3(s)$  and Q(s). Since the poles of transfer functions  $T_1(s)$ ,  $T_2(s)$  and  $T_3(s)$  can be selected from arbitrary complex numbers on the left half plane by the selection of F, H and weighting functions, adding a constraint such that  $Q(s) \in \mathcal{RH}^D_{\infty}$ , the  $H_{\infty}$  controller which locates closed loop poles in a specified domain Dom(D) can be obtained. We call the problem of finding  $Q(s) \in \mathcal{RH}^D_{\infty}$  which satisfies the constraint of  $H_{\infty}$ -norm the  $H_{\infty}$  problem with pole specification. The problem is formulated as follows.

#### Problem 6.2.1 The $H_{\infty}$ problem with pole specification

Suppose that the transfer functions  $T_i(s) \in \mathcal{RH}^D_{\infty}$  are given. Then, find a stable system Q(s) that achieves

$$||T_1(s) - T_2(s)Q(s)T_3(s)||_{\infty} < 1$$

such that

$$Q(s) \in \mathcal{RH}^D_{\infty}.$$

If such a Q(s) is obtained, the controller in a request is given by substituting Q(s) into the equation in (6.1).

#### 6.2.2 A numerical approach to controller design

In order to solve the problem stated above numerically, we must search a solution over an infinite-dimensional function space  $\mathcal{RH}_{\infty}$ . Then, we focus on an approximation stated as follows.

Let us define a sequence of transfer functions  $Q_k(s) \in \mathcal{RH}_{\infty}, k = 1, 2, ..., N$ , and let us denote transfer functions as follows

$$R_0 \triangleq T_1(s)$$
$$R_k \triangleq T_2(s)Q_k(s)T_3(s).$$

Then, restricting the closed loop transfer function  $T_{zw}(s)$  to a finite-dimensional function space as

$$H_N(x) \triangleq R_0 + \sum_{k=1}^N x_k R_k, \quad x_k \in \mathbb{R},$$

we optimize a function

$$\phi(x) \triangleq \phi\left(H_N(x)\right),$$

where  $\phi$  is a function which specifies some performance of the control system. This restriction is called Ritz Approximation. In this approximation, it is assured that  $\phi(x)$  converges to an optimal value as  $N \to \infty$  [3, 54]. To each  $x \in \mathbb{R}^N$  there corresponds the controller

$$K(x) = \mathcal{F}_l\left(K_0(s), \sum_{k=1}^N x_k Q_k(s)\right)$$

that achieves the closed loop specification  $\phi(H_N(x)) < \gamma$ .

Now, in order to solve Problem 6.2.1, we put the free parameter Q(s) as

$$Q(s) = \sum_{k=1}^{N} x_k Q_k(s) \triangleq \sum_{k=1}^{N} x_k \left(\frac{\alpha}{s+\alpha}\right)^k,$$
(6.3)

where  $\alpha > 0$  is chosen such that  $Q_k(s)$  satisfies  $Q_k(s) \in \mathcal{RH}^D_{\infty}$  and  $x_k, (k = 1, 2, ..., N)$  are finite-dimensional parameters. In this problem the function  $\phi$  indicates the  $H_{\infty}$ -norm. Since the function of the  $H_{\infty}$ -norm satisfies

$$\|\beta H_a + (1-\beta)H_b\|_{\infty} \le \beta \|H_a\|_{\infty} + (1-\beta)\|H_b\|_{\infty}, \quad \forall \beta \in [0,1],$$

where  $H_a$  and  $H_b$  are arbitrary transfer functions in  $\mathcal{RH}_{\infty}$ , it is apparent that  $\phi$  is a convex function. Hence, if we put the free parameter Q(s) as (6.3), Problem 6.2.1 can be reduced to a finite-dimensional convex optimization problem.

#### 6.2.3 A defect of the numerical approach

Thus, Problem 6.2.1 can be solved by using a convex optimization method such as ellipsoid algorithm [3]. However, sometimes the order of the free parameter becomes very high, causing the controller also high order. In order to understand this mechanism, let us represent the controller in (6.1) as

$$K(s) = \mathcal{F}_l\left(K_0(s), Q(s)\right),$$

where  $K_0(s)$  is defined as

$$K_0(s) \triangleq \left[ \begin{array}{cc} X(s)Y^{-1}(s) & M(s) + X(s)Y^{-1}(s)N(s) \\ Y^{-1}(s) & Y^{-1}(s)N(s) \end{array} \right],$$

and let us introduce the state-space form of  $K_0(s)$  and Q(s) as follows

$$K_0(s) = \begin{bmatrix} A_k & B_{k_1} & B_{k_2} \\ \hline C_{k_1} & D_{k_{11}} & D_{k_{12}} \\ C_{k_2} & D_{k_{21}} & D_{k_{22}} \end{bmatrix}$$
$$Q(s) = \begin{bmatrix} A_q & B_q \\ \hline C_q & D_q \end{bmatrix}.$$

Then the state-space form of the controller  $K(s) = \mathcal{F}_l(K_0(s), Q(s))$  can be represented as follows.

$$K(s) = \begin{bmatrix} A_k + B_{k_2} D_q E C_{k_2} & B_{k_2} (I + D_q E D_{K_{22}}) C_q & B_{k_1} + B_{k_2} D_q E D_{k_{21}} \\ B_q E C_{k_2} & A_q + B_q E D_{k_{22}} C_q & B_q E D_{k_{21}} \\ \hline C_{k_1} + D_{k_{12}} D_q E C_{k_2} & D_{k_{12}} (I + D_q E D_{k_{22}}) C_q & D_{k_{11}} + D_{k_{12}} D_q E D_{k_{21}} \end{bmatrix}, \quad (6.4)$$

where  $E = (I - D_{k_{22}}D_q)^{-1}$  and  $K_0(s)$ . From this representation of K(s) it can be seen that the McMillan degree of K(s) can be estimated as follows

$$\deg\left(K(s)\right) \le \deg\left(K_0(s)\right) + \deg\left(Q(s)\right). \tag{6.5}$$

In the above relation, the inequality holds when pole-zero cancellations occur between  $K_0(s)$ and Q(s). In such a case, the order of the controller is reduced to some degree. However, it is not always expected that the pole-zero cancellation reduces the order of the controller to some required order, rather it is ordinary that the pole-zero cancellation doesn't occur. Thus, in this numerical approach the increase in the order of the resultant controller is one of difficulties, hence the controller order reduction is required.

#### 6.3 The controller reduction by the approximation of Q(s)

The relation in (6.5) informs us an upper bound of the order of the resultant controller. In this study we consider to reduce the order of the upper bound in order to reduce deg (K(s)). In this context, it is useful to reduce the order of Q(s). When approximating Q(s), it is important to consider the preservation of the closed loop specifications such as the closed loop stability, the designation of closed loop pole position and the  $H_{\infty}$  norm constraint. Suppose that Q(s) is approximated by  $Q_r(s)$ . The conditions of the stability and the pole position are satisfied if  $Q_r(s) \in \mathcal{RH}_{\infty}^D$ . Then we have only to consider the constraint on the norm in order to derive the low-order  $H_{\infty}$  controller which satisfies the pole specification. Now, it is required to consider a model approximation method which is closed on  $\mathcal{RH}_{\infty}^D$ .

#### 6.3.1 Balanced truncation

Now, let us introduce one of the most useful model approximation method. Balanced truncation [52, 56, 43] is one of the most effective model approximation methods for reducing the order of stable systems. We consider to approximate the system Q(s) based on the Balanced truncation. Consider a linear time invariant stable system Q(s) described by

$$Q(s) = \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix},$$

where it is assumed that the McMillan degree of Q(s) is n, and the pairs (A, B), (A, C) are controllable and observable. The controllability Gramian of Q(s) is defined as

$$\Sigma_c \triangleq \int_0^\infty e^{At} B B^T e^{A^T t} dt,$$

and is given by the unique positive definite solution to the Lyapunov equation

$$\Sigma_c A^T + A\Sigma_c + BB^T = O.$$

The observability Gramian of G(s) is defined as

$$\Sigma_o \triangleq \int_0^\infty e^{A^T t} C^T C e^{At} dt,$$

and is given by the unique positive definite solution to the Lyapunov equation

$$\Sigma_o A + A^T \Sigma_o + C^T C = O.$$

Positive square roots of eigenvalues of  $\Sigma_o \Sigma_c$  are called Hankel singular values of Q(s), and they are invariant under a nonsingular state transformation of Q(s). Let  $\sigma_1 \ge \sigma_2 \ge \cdots \ge \sigma_n > 0$ be the Hankel singular values of Q(s). The system is said to be *internally balanced* if the Gramians satisfy the equation

$$\Sigma_c = \Sigma_o = \Sigma$$

where  $\Sigma = \text{diag}(\sigma_1, \sigma_2, \cdots, \sigma_n)$ . Let us partition  $\Sigma$  accordingly into

$$\Sigma = \left[ \begin{array}{cc} \Sigma_a & O \\ O & \Sigma_b \end{array} \right]$$

where  $\Sigma_a = \text{diag}(\sigma_1, \sigma_2, \cdots, \sigma_r)$  and  $\Sigma_b = \text{diag}(\sigma_{r+1}, \sigma_{r+2}, \cdots, \sigma_n)$ .

Suppose that Q(s) is *internally balanced*. Let us partition A, B, C conformably with the partitions of  $\Sigma$ .

$$Q(s) = \begin{bmatrix} A_{11} & A_{12} & B_1 \\ A_{21} & A_{22} & \overline{B_2} \\ \hline C_1 & C_2 & D \end{bmatrix}, \quad A_{11} \in \mathbb{R}^{r \times r}, A_{22} \in \mathbb{R}^{(n-r) \times (n-r)}$$

Then an approximated system of Q(s) is given by an r-state reduced-order system:

$$Q_r(s) = \begin{bmatrix} A_{11} & B_1 \\ \hline C_1 & D \end{bmatrix}.$$

Concerned about the approximation the following lemma is known [53].

**Lemma 6.3.1** Let Q(s) be stable and minimal system with n states. Let r < n and let  $Q_r(s)$  be a reduced-order system obtained by approximating Q(s) with Balanced truncation. Then the following items hold.

- 1)  $Q_r(s)$  is stable and minimal system.
- 2)  $||Q(s) Q_r(s)||_{\infty} \leq 2 \operatorname{trace} \Sigma_b.$

**Remark 6.3.1** Balanced truncation is a method to approximate a system in  $\mathcal{RH}_{\infty}$  with a reduced-order system in  $\mathcal{RH}_{\infty}$ . If  $Q(s) \in \mathcal{RH}_{\infty}^D$  is approximated by  $Q_r(s)$ , it is only assured that  $Q_r(s) \in \mathcal{RH}_{\infty}$ . Hence the reduced-order system may not satisfy the pole specification of the original system.

In the following section we will consider a model reduction method which preserves the specified domain in which the poles locate.

#### 6.3.2 Model approximation with constraint of pole position

Besides the internal stability, the closed loop pole specification to a specified domain is important in controller design [21]. The domain is specified to the left hand side divided by the dashed lines which is shown in Figure 6.2. If the closed loop poles are confined to this domain, then the system modes damp asymptotically at desired rates. However, it is difficult to address this domain directly. Hence, we approximate this domain by Dom(D).

This section proposes a model approximation method that preserves the domain in which the poles locate. Here, the domain is limited to a disk domain. Outline of our idea concerned with the approximation method is listed as follows.

- 1. The coordinate of the original system is transformed using an affine transformation and a bilinear transformation. Then the disk domain on the complex left half plane is transformed to the complex left half plane of the changed coordinate. Poles of the original system are mapped into the left half plane in the new coordinate.
- 2. The original system in the new coordinate is approximated to a low-order one using Balanced truncation. Then it should be noted that the poles of the reduced-order system in the new coordinate still locate in the left half plane.
- 3. Again the reduced-order system is transformed using the inverse of the affine transformation and the bilinear transformation. Then the poles of the reduced-order system in the original coordinate locate in the specified disk on the complex left half plane.

Thus, we can approximate a system, which has poles on a specified domain with a reducedorder system, which still has poles on a specified domain. Now let us move on to details.



Figure 6.3: s-plane and  $\tilde{s}$ -plane

Firstly, let us introduce some important transformations. The affine transformation is defined as

$$\frac{s+\alpha}{\beta},$$

where  $\alpha$  and  $\beta$  are some real numbers. This transforms Dom(D) to the domain within the unit disk on the complex plane. The bilinear transformation is defined as

$$\tilde{s} = \frac{v-1}{v+1}.$$

This transforms the domain within the unit disk on the complex plane to the complex left half plane. The mixed transformation which transforms Dom(D) to the complex left half plane is described by

$$\tilde{s} = \frac{s + \alpha - \beta}{s + \alpha + \beta},\tag{6.6}$$

where  $\alpha > \beta \neq 0$ . This transformation is called a Linear Fractional Transformation (LFT), and is a one-to-one mapping of  $S^2$  onto  $S^2$ . LFT has a lot of attractive properties, one of which is given by the following lemma.

**Lemma 6.3.2** Let Q(s) be analytic outside Dom(D). Let the function with the coordinate transformed by LFT be defined as

$$R(\tilde{s}) \triangleq Q\left(\frac{(-\alpha - \beta)\tilde{s} + \alpha - \beta}{\tilde{s} - 1}\right).$$

Then  $R(\tilde{s})$  is analytic in the complex right half plane.

**Proof.** The reader may refer to the Riemann mapping theorem [58].

Thus, by transforming the coordinate of  $Q(s) \in \mathcal{RH}^D_{\infty}$  with LFT in (6.6) we can obtain another system  $R(\tilde{s}) \in \mathcal{RH}_{\infty}$ . In the later discussion, for notational ease  $R(\tilde{s})$  is denoted by R(s). Then we approximate R(s) by a reduced-order system  $R_r(s)$ . Assume that R(s) is a stable *n*-th order system. Using Balanced truncation, we can obtain an *r*-the order reduced-order system. Let  $R_r(s)$  be the reduced-order system obtained by Balanced truncation. From Lemma 6.3.1  $R_r(s) \in \mathcal{RH}_{\infty}$ , hence  $R_r(s)$  is analytic in the complex right half plane.

Then transform the coordinate of  $R_r(s)$  by the inverse of the LFT:

$$s = \frac{(-\alpha - \beta)\tilde{s} + \alpha - \beta}{\tilde{s} - 1}.$$
(6.7)

Let the transformed system be  $Q_r(s)$ . Then the *r*-the order reduced-order system  $Q_r(s)$  is analytic outside Dom(D), i. e.,  $Q_r(s) \in \mathcal{RH}^D_{\infty}$ . Here, we obtain the following theorem.

**Theorem 6.3.1** Let Q(s) be analytic outside Dom(D). Then the reduced-order system  $Q_r(s)$  is analytic outside Dom(D).

**Proof.** The system R(s) is analytic in the complex right half plane. The reduced-order system  $R_r(s)$  which is obtained by Balanced truncation is also analytic in the complex right half plane. Therefore  $Q_r(s)$  obtained by the inverse transformation of LFT in (6.6) is analytic outside Dom(D), because the inverse transformation is also LFT, that is, a one-to-one mapping of  $S^2$  onto  $S^2$ .

Thus the procedure to approximate Q(s) is listed as follows.

**Procedure 6.3.1** Assume that an original system  $Q(s) \in \mathcal{RH}_{\infty}^{D}$  is obtained. Then the following procedure derives an approximated system  $Q_r(s) \in \mathcal{RH}_{\infty}^{D}$ .

step 1: Transform  $Q(s) \in \mathcal{RH}_{\infty}^D$  to  $R(\tilde{s}) \in \mathcal{RH}_{\infty}$  with LFT in (6.6).

step 2: Approximate  $R(\tilde{s}) \in \mathcal{RH}_{\infty}$  by  $R_r(\tilde{s}) \in \mathcal{RH}_{\infty}$  with Balanced truncation.

step 3: Transform  $R_r(\tilde{s}) \in \mathcal{RH}_\infty$  to  $Q_r(s) \in \mathcal{RH}_\infty^D$  with LFT in (6.7).

Thus, we can reduce the dynamical order of  $Q(s) \in \mathcal{RH}^D_{\infty}$  with preserving the specified domain in which poles locate.

In the second place, we look into the error of the approximation discussed in the above. The following theorem gives an upper bound for the approximation error.

**Theorem 6.3.2** Assume that the  $H_{\infty}$  norm error bound of  $R(s) - R_r(s)$  is known to be  $\gamma$ , *i.e.* the following inequality holds.

$$\left\|R(s) - R_r(s)\right\|_{\infty} < \gamma \tag{6.8}$$

Then the error bound of  $Q(s) - Q_r(s)$  is evaluated with the following inequality.

$$\|Q(s) - Q_r(s)\|_{\infty} < \gamma \tag{6.9}$$

**Proof.** Define  $E_R(s)$ ,  $E_Q(s)$  as follows.

$$E_R(s) \triangleq R(s) - R_r(s)$$
$$E_Q(s) \triangleq Q(s) - Q_r(s)$$

The systems Q(s) and  $Q_r(s)$  are analytic outside Dom(D), hence  $E_Q(s)$  is also analytic outside Dom(D). By the maximum modulus principle [58] of the analytical functions we obtain an inequality

$$\sup_{Dom(C)} \left\{ E_Q(s) \right\} > \sup_{\omega} \left\{ E_Q(j\omega) \right\}.$$
(6.10)

The LFT in (6.6) transforms the value of the function  $E_Q(s)$  on Dom(C) to the value of the function  $E_R(j\omega)$  on the imaginary axis. Hence the following equation is obtained.

$$\sup_{\omega} \left\{ E_R(j\omega) \right\} = \sup_{Dom(C)} \left\{ E_Q(s) \right\}.$$
(6.11)

Since  $R(s) \in \mathcal{RH}_{\infty}$  and  $R_r(s) \in \mathcal{RH}_{\infty}$  are analytic in the complex right half plane,  $E_R(s)$  is also analytic in the complex right half plane. The  $H_{\infty}$  norm of the function  $E_R(s)$  equals the maximal value of the function  $E_R(j\omega)$  on the imaginary axis.

$$||E_R(s)||_{\infty} = \sup_{\omega} \{E_R(j\omega)\}$$
(6.12)

From equations. (6.11) and (6.12), the following equation is obtained

$$\|E_R(s)\|_{\infty} = \sup_{Dom(C)} \{E_Q(s)\}.$$
(6.13)

If the following inequality

$$\left\|E_R(s)\right\|_{\infty} < \gamma$$

holds, then from equation (6.13), the inequality is obtained

$$\sup_{Dom(C)} \left\{ E_Q(s) \right\} < \gamma. \tag{6.14}$$

Then from the inequalities (6.10) and (6.14), the following inequality

$$\sup_{\omega} \left\{ E_Q(j\omega) \right\} = \left\| E_Q(s) \right\|_{\infty} < \gamma$$

holds.

**Remark 6.3.2** From Lemma 6.3.1, an upper bound of the reduction error  $E_R(s)$  is given by the following inequality.

$$||E_R(s)||_{\infty} \leq 2$$
trace,  $\Sigma_2$ 

where  $\Sigma_2$  is defined as

$$\Sigma_2 \triangleq \operatorname{diag}\left(\sigma_{r+1}, \sigma_{r+2}, \cdots, \sigma_n\right)$$

and  $\sigma_i$ ,  $(1 \leq i \leq n)$  are the Hankel singular values of R(s). Hence, using the result of Theorem 6.3.2, the error bound of  $E_Q(s)$  is evaluated using the Hankel singular values of R(s) as follows

$$||E_Q(s)||_{\infty} < 2 \operatorname{trace} \Sigma_2.$$

#### 6.3.3 The $H_{\infty}$ controller reduction

In this section, by using the approximation method proposed in the previous section we consider to reduce the order of the  $H_{\infty}$  controller which is represented by LFT with a high-order Youla parameter. In order to reduce the order of the  $H_{\infty}$  controller, it is useful to reduce the order of the Youla parameter when it is high order. We apply the approximation method proposed in the above section to reduce the order of the Youla parameter.

Suppose that  $Q^*(s) \in \mathcal{RH}^D_{\infty}$ , which is an *n*-th order solution to Problem 6.2.1 is given, and define the value of the objective function

$$\lambda^* \triangleq \|T_1(s) - T_2(s)Q^*(s)T_3(s)\|_{\infty} < 1.$$

We approximate  $Q^*(s)$  with r-th order system  $Q_r(s) \in \mathcal{RH}^D_{\infty}$  by Procedure 6.3.1, then define the value of the objective function

$$\lambda_r \triangleq \|T_1(s) - T_2(s)Q_r(s)T_3(s)\|_{\infty}.$$

Then the following lemma holds.

Lemma 6.3.3 The inequality holds.

$$\lambda_r - \lambda^* \le \|T_2(s)\|_{\infty} \cdot \|Q^*(s) - Q_r(s)\|_{\infty} \cdot \|T_3(s)\|_{\infty}$$
(6.15)

**Proof.** It can immediately verified by the following inequality.

$$\lambda_{r} = \|T_{1}(s) - T_{2}(s)Q^{*}(s)T_{3}(s) + T_{2}(s)(Q^{*}(s) - Q_{r}(s))T_{3}(s)\|_{\infty}$$
  

$$\leq \|T_{1}(s) - T_{2}(s)Q^{*}(s)T_{3}(s)\|_{\infty} + \|T_{2}(s)(Q^{*}(s) - Q_{r}(s))T_{3}(s)\|_{\infty}$$
  

$$\leq \lambda^{*} + \|T_{2}(s)\|_{\infty} \cdot \|Q^{*}(s) - Q_{r}(s)\|_{\infty} \cdot \|T_{3}(s)\|_{\infty}$$

By using the above lemma, we can derive the next result.

**Theorem 6.3.3** Let us denote the Hankel singular values of R(s) as  $\sigma_i^R$ ,  $i = 1, 2, \dots, n$ . If an inequality

$$\lambda^* + 2 \|T_2(s)\|_{\infty} \cdot \text{trace } \Sigma_2^R \cdot \|T_3(s)\|_{\infty} \le 1, \tag{6.16}$$

where  $\Sigma_2^R$  implies

$$\Sigma_2^R \triangleq \operatorname{diag} \left(\sigma_{r+1}^R, \sigma_{r+2}^R, \cdots, \sigma_n^R\right),$$

holds, then  $Q_r(s)$  is a solution to Problem 6.2.1.

**Proof.** From Lemma 6.3.1, an inequality

$$||R(s) - R_r(s)||_{\infty} \leq 2 \operatorname{trace} \Sigma_{R2}$$

holds, hence from Theorem 6.3.2 an inequality

$$\left\|Q(s) - Q_r(s)\right\|_{\infty} < 2 \operatorname{trace} \Sigma_{R2} \tag{6.17}$$

holds. Then, from Lemma 6.3.3 and (6.17),

$$\lambda_r < \lambda^* + 2 \|T_2(s)\|_{\infty} \cdot \operatorname{trace} \Sigma_{R2} \cdot \|T_3(s)\|_{\infty}$$

holds. Hence, if the inequality (6.16) holds,  $\lambda_r < 1$  is satisfied. Then,  $Q_r(s) \in \mathcal{RH}^D_{\infty}$  is a solution to Problem 6.2.1.

Substituting  $Q_r(s)$  into the representation of the controller in (6.1) derives an approximated controller. Then it is assured that the approximated controller satisfies the constraint of the pole position in the closed loop. A sufficient condition for the approximated controller satisfies the constraint of the  $H_{\infty}$ -norm is given in Theorem 6.3.3. Then we have to consider next is to ascertain whether the order of the approximated controller is really reduced or not. In the state space representations of the  $H_{\infty}$  controller in (6.4), if Q(s) is replaced with  $Q^*(s)$ , the resultant controller is not necessarily minimal, hence the McMillan degree of the controllers may reduce to some order:

$$\deg\left(\mathcal{F}_l\left(K(s), Q^*(s)\right)\right) \le \deg\left(K_0(s)\right) + \deg\left(Q^*(s)\right).$$

Hence the pole-zero cancellation [61, 36] may cause a case such that the order of the approximated controller is higher than that of the original controller.

In order to avoid such a situation, we look into the pole-zero cancellation. Let us denote the controller  $K_0(s)$  as

$$K_0(s) = \left[ \begin{array}{cc} K_{11}(s) & K_{12}(s) \\ K_{21}(s) & K_{22}(s) \end{array} \right].$$

Then it is known the property concerned about the pole-zero cancellation.

**Lemma 6.3.4** Every mode of the unobservable subspace in  $\mathcal{F}_l(K_0(s), Q(s))$  is the invariant zero of  $K_{12}(s)$ . Every mode of the uncontrollable subspace in  $\mathcal{F}_l(K_0(s), Q(s))$  is the invariant zero of  $K_{21}(s)$ .

**Proof.** See reference [37, 36].

By using the above lemma, we obtain the following result.

**Theorem 6.3.4** Suppose that sum of the numbers of the invariant zeros of  $K_{12}(s)$  and  $K_{21}(s)$  is TZ > 0. Then, if the inequality

$$\deg\left(Q^*(s)\right) - \deg\left(Q_r(s)\right) > TZ$$

holds, the following inequality holds.

$$\deg\left(\mathcal{F}_l\left(K_0(s), Q^*(s)\right)\right) > \deg\left(\mathcal{F}_l\left(K_0(s), Q_r(s)\right)\right) \tag{6.18}$$

holds.

**Proof.** Suppose that the inequality

$$\deg\left(Q^*(s)\right) - \deg\left(Q_r(s)\right) > TZ \tag{6.19}$$

holds. From Lemma 6.3.4, the pole-zero cancellations in  $\mathcal{F}_l(K_0(s), Q^*(s))$  may happen within its unobservable mode and uncontrollable mode. Hence, an inequality

$$\deg\left(\mathcal{F}_{l}\left(K_{0}(s), Q^{*}(s)\right)\right) \ge \deg\left(K_{0}(s)\right) + \deg\left(Q^{*}(s)\right) - TZ$$
(6.20)

holds. From (6.19) and (6.20), the following inequality holds.

$$\deg\left(\mathcal{F}_{l}\left(K_{0}(s), Q^{*}(s)\right)\right) > \deg\left(K_{0}(s)\right) + \deg\left(Q_{r}(s)\right)$$
(6.21)

On the other hand,  $\deg \left(\mathcal{F}_l\left(K_0(s), Q_r(s)\right)\right)$  satisfies

$$\deg\left(\mathcal{F}_l\left(K_0(s), Q_r(s)\right)\right) \le \deg\left(K_0(s)\right) + \deg\left(Q_r(s)\right). \tag{6.22}$$

Hence, from (6.21) and (6.22), (6.18) holds.

#### 6.4 A numerical example

Consider the following 17-th order system  $Q^*(s)$  which is obtained as a solution to Problem 6.2.1. The pole positions of  $Q^*(s)$  are illustrated in Figure 6.4. Let the domain in which the poles are located be denoted Dom(D) with center -100 + j0 and radius 99.5 on the complex plane, hence the parameters  $\alpha$  and  $\beta$  are set as  $\alpha = 100, \beta = 99.5$ . The 12-th order reduced-order system which is obtained by the proposed approximation method is denoted by  $Q_{12prop}(s)$  and the 12-th order reduced-order system obtained by Balanced truncation is denoted by  $Q_{12BT}(s)$ . The pole positions of these systems are illustrated in Figure 6.5 and Figure 6.6, respectively. From these figures, it is apparent that all the poles of  $Q_{12prop}(s)$ are located within Dom(D), while some poles of  $Q_{12BT}(s)$  are located outside Dom(D). The poles of the closed loop transfer functions  $F_{affine}(Q_{12prop}(s))$  and  $F_{affine}(Q_{12BT}(s))$  are

144



Figure 6.4: Pole position of  $Q^*(s)$ 

illustrated in Figure 6.7 and Figure 6.8, respectively. It is also apparent that the closed loop pole specification is satisfied by using the proposed model reduction method, on the other hand, the specification is not satisfied by using Balanced truncation only. The  $H_{\infty}$  norm of the approximated closed loop transfer function is evaluated by using the following inequality.

$$\begin{aligned} \|F_{affine} \left(Q_{12prop}(s)\right)\|_{\infty} &= \|T_{1}(s) - T_{2}(s)Q(s)T_{3}(s) + T_{2}(s) \left(Q(s) - Q_{12prop}(s)\right)T_{3}(s)\|_{\infty} \\ &\leq \|T_{1}(s) - T_{2}(s)Q(s)T_{3}(s)\|_{\infty} + \|T_{2}(s) \left(Q(s) - Q_{12prop}(s)\right)T_{3}(s)\|_{\infty} \\ &\leq \|F_{affine} \left(Q(s)\right)\|_{\infty} + \|T_{2}(s)\|_{\infty} \cdot \|T_{3}(s)\|_{\infty} \cdot \|Q(s) - Q_{12prop}(s)\|_{\infty} \end{aligned}$$

$$(6.23)$$

We can check the  $H_{\infty}$  norm condition by using the above inequality and the result of Theorem 6.3.2. In fact, the  $H_{\infty}$  norm of the approximated closed loop transfer function is evaluated as

$$\|F_{affine}\left(Q_{12prop}(s)\right)\|_{\infty} < 0.965.$$
 (6.24)

Hence, the reduced-order  $H_{\infty}$  controller  $\mathcal{F}_l(K_0(s), (Q_{12prop}(s)))$  which satisfies the pole specification can be obtained.

#### 6.5 Summary and discussion

In this chapter we have proposed a controller reduction method for a  $H_{\infty}$  controller which is designed with the numerical approach based on the optimization of the Youla parameter. This method is based on the approximation of the Youla parameter that is designed to satisfy the constraints of the closed loop system. Thus the parameter holds information of



Figure 6.5: Pole position of  $Q_{12prop}(s)$ 

Table 6.1: A summary of deg  $(Q_r)$ ,  $\lambda^* + 2 ||T_2||_{\infty} \cdot \operatorname{trace} \Sigma_2^R \cdot ||T_3||_{\infty}$ ,  $\lambda$  and the McMillan degrees of the controllers

$\deg(Q_r)$	$\lambda^* + 2 \ T_2\ _{\infty} \cdot \operatorname{trace} \Sigma_2^R \cdot \ T_3\ _{\infty}$	$\lambda$	$\deg\left(\mathcal{F}_{l}\left(K_{0},Q_{r}\right)\right)$
17	—	0.802	20(19)
16	0.803	0.802	20(18)
15	0.805	0.802	19(17)
14	0.814	0.802	18(16)
13	0.839	0.804	17 (15)
12	0.965	0.807	16(14)
11	1.21	0.826	15 (13)
10	2.47	1.09	13(12)

the constraints of the closed loop system, hence it is natural to approximate the parameter without losing the information of the constraints. However there was a difficulty in reducing the order of the controller. Since this method approximates the Youla parameter, the order of the controller is not necessary reduced sufficiently if the order of the parameter is reduced. In order to avoid this situation we have given a sufficient condition for the order of the approximated controller is really reduced.

Thus we have given a  $H_{\infty}$  controller reduction method. The advantages of the method are that the approximated controller certainly satisfies the internal stability and the constraint of the pole position in the closed loop. For the constraint of the  $H_{\infty}$ -norm, a sufficient condition for satisfying the constraint is given. The effectiveness of this result was verified with a numerical example.



Figure 6.6: Pole position of  $Q_{12BT}(s)$ 



Figure 6.7: Pole position of  $F_{affine}(Q_{12prop}(s))$ 



Figure 6.8: Pole position of  $F_{affine}(Q_{12BT}(s))$ 

# Chapter 7 Conclusion

In general, as the structure of a design problem becomes more complex, the number of choices of formulations to solve the problem increases. As we have seen in chapters 4 and 5, the complexity and the optimality of the solution depend on the formulation chosen. In the designing of robust controllers based on the  $H_{\infty}$  problem, there are many approaches to solving the design problems. However, these problems are usually formulated as the standard  $H_{\infty}$  problem and their solutions sometimes become complex and result in high-order controllers. It is therefore important, not only to solve a problem formulated beforehand but also to consider a new formulation.

This thesis has considered the formulation of the  $H_{\infty}$  control problem in the designing of low-order controllers from several viewpoints. Two types of approaches to the problem are considered: a *direct* approach based on the derivation of the reduced-order  $H_{\infty}$  controllers, and an *indirect* approach based on the approximation of a high-order controller. For each approach, some methods for designing low-order controllers have been proposed.

1. In the *direct* approach, by getting a hint to reduce the dynamical order of a controller from the minimal-order-observer design, we have treated the non-standard  $H_{\infty}$  control problems, where the dimensions of the control inputs are greater than those of the controlled outputs, or the dimensions of the measurement outputs are greater than those of the disturbance inputs. As a result, generalized classes of the reduced-order  $H_{\infty}$  controllers are expressed with free parameters. The reduced-order controllers are classified into two types: a minimal-order-observer-type and a dual type of the minimal order observer. Thus, this study has shown that the low-order controllers can be obtained by reducing some controller design problem to the non-standard  $H_{\infty}$  problem. While the formulation of non-standard  $H_{\infty}$  problem has been avoided in previous studies, it is expected that the non-standard  $H_{\infty}$  problem is positively utilized in the controller designs. This thesis has, in fact, clarified the controller design problems that can be formulated as the non-standard problems and has shown the advantages of formulating them in this way. This thesis has gone on to clarify some classes of the problems that are reducible to the non-standard  $H_{\infty}$  problem and has proposed methods for deriving robust controllers.

- Based on the non-standard H<sub>∞</sub> problem, a new formulation for designing an integral-type H<sub>∞</sub> controller is proposed, and new method for deriving the low-order controller is shown. The advantages of this formulation are due to the simplicity of the derivation. If the plant doesn't include integrators, it is not necessary to solve the so-called pseudo stabilizing solution of the ARE. On the other hand, if the plant includes the integrators, although it is still necessary to solve the pseudo stabilizing solution, it is not necessary to transform the non-standard problem to the standard problem and to separate the integrators from the plant. Moreover, the results can be extended to TDF controller design which is also reducible to the non-standard H<sub>∞</sub> control problem.
- A TDF robust controller design problem is reduced to the non-standard H<sub>∞</sub> problem, and a new method for designing a low-order TDF controller is proposed. This method has many advantages over conventional design methods of TDF controller. One is that the method makes it possible to design both the feedback controller and the feedforward controller simultaneously, hence the trade-off between the performances of the controllers can be considered. In this thesis, it is pointed out that there exists trade-off between the performances of the controllers if those dynamics are common. The other is that the order of the TDF controller can be reduced to the order which is lower than that of the generalized plant.
- 2. With regard to the *indirect* approach, this thesis has treated a specialized solution of the  $H_{\infty}$  controller, which is represented with an optimized Youla parameter. Since the order of the parameter is extremely high, the order of the controller is also high. In this thesis, we have proposed a model approximation method for a linear time invariant stable system whose poles are located in a specified disk on the left half plane. This method preserves the domain in which the poles of the original system are located. An upper bound of the model approximation error was evaluated using the Hankel singular values of the original system. By using the approximation method for reducing the order of the Youla parameter, a method for reducing the order of the  $H_{\infty}$  controller, which satisfies a constraint of the closed loop pole position, is proposed.

# Bibliography

- B. D. O. Anderson and Y. Liu. Controller reduction: Concepts and approaches. *IEEE Trans. Automatic Control*, 34(8):802–812, 1989.
- [2] D. S. Bernstein and D. C. Hyland. The optimal projection equations for fixed-order dynamic compensation. *IEEE Trans. Automatic Control*, 29(11):1034–1037, 1985.
- [3] S. P. Boyd and C. H. Barratt. Linear Controller Design –Limits of performance–. Prentice Hall, 1991.
- [4] R. Y. Chiang and M. G. Safnov. Real K<sub>m</sub>-synthesis via generalized multipliers. Proc. of ACC, pages 2417–2418, 1994.
- [5] Y. Chida. Two-degree-of-freedom control using feedback control dynamical characteristics (in Japanese). Trans. of IEE Japan, 117-C(4):439–445, 1997.
- [6] Y. Chida, Y. Yamagichi, H. Soga, T. Kida, I. Yamaguchi, and S. Ishikawa. Ets-vi attitude control by  $H_{\infty}$  control and two-degree-of-freedom control (in Japanese). Proc. of SICE symposium on control theory, pages 233–238, 1996.
- [7] Urs Christen, Martin F. Weilenmann, and Hans P. Geering. Design of  $H_2$  and  $H_{\infty}$  controllers with two degrees of freedom. *Proc. of ACC*, pages 2391–2395, 1994.
- [8] E. G. Collins, L. D. Davis, and S. Richter. Design of reduced-order, H<sub>2</sub> optimal controllers using a homotopy algorithm. *International Journal of Control*, 61(1):97–126, 1995.
- [9] E. G. Collins and D. Shadhukhan. A comparison of descent and continuation algorithms for H<sub>2</sub> optimal, reduced-order control design. *International Journal of Control*, 69(5):647–662, 1998.
- [10] J. C. Doyle, B. A. Francis, and A. R. Tannenbaum. Feedback control theory. Macmillam Publishing Company, 1992.
- [11] J. C. Doyle, K. Glover, P. P. Khargonekar, and B. A. Francis. State-space solutions to standard H<sub>2</sub> and H<sub>∞</sub> control problems. *IEEE Trans. Automatic Control*, 34(8):831–847, 1989.

- [12] B. A. Francis. A cource in  $H_{\infty}$  control theory. Springer-Verlag, 1987.
- [13] Y. Fujisaki and M. Ikeda. Synthesis of two-degrees-of-freedom optimal servosystems (in Japanese). Trans. of SICE, 27(8):907–914, 1991.
- [14] K. Furuta, S. Kawaji, T. Mita, and S. Hara. Mecanical system control (in Japanese). Ohmsha Publishing Co., LTD., 1984.
- [15] K. Furuta, O. Nishimura, and Y. Shinata. Alternative design of  $H_{\infty}$  controller. *Proc. of CDC*, pages 732–733, 1996.
- [16] P. Gahinet and P. Apkarian. A linear matrix inequality approach to  $H_{\infty}$  control. International Journal of Robust and Nonlinear Control, 4:421–448, 1994.
- [17] A. Giusto, A. T. Neto, and E. B. Castelan.  $H_{\infty}$  and  $H_2$  design techniques for a class of prefilters. *IEEE Trans. Automatic Control*, 41(6):865–870, 1996.
- [18] K. Glover. All Hankel-norm approximations of linear multivariable systems and their  $\mathcal{L}_{\infty}$ -error bounds. International Journal of Control, 39(6):1115–1193, 1984.
- [19] K. Glover and J. C. Doyle. State-space formulae for all stabilizing controllers that satisfy an  $H_{\infty}$ -norm bound and relations to risk sensitivity. *Syst. Contr. Lett.*, 11:167–172, 1988.
- [20] P. J. Goddard. Performance-Preserving Controller Approximation. PhD thesis, Cambridge University, 1995.
- [21] W. M. Haddad and D. S. Bernstein. Controller design with regional pole constraints. *IEEE Trans. Automatic Control*, 37(1):54–69, 1992.
- [22] T. Hagiwara, Y. Ohtani, and M. Araki. A design method of LQI servo systems with two degrees of freedom (in Japanese). *Trans. of ISCIE*, 4(12):501–510, 1991.
- [23] S. Hara, H. Fujioka, T. Kosugiyama, and T. Asai. A synthesis for robust tracking systems based on  $H_{\infty}$  control. *Trans. of SICE*, 32(4):502–509, 1996.
- [24] M. Hirata, K. Liu, and T. Mita. A design method for two-degree-of-freedom controller using the dynamical model of the feedback controller (in Japanese). Trans. of IEE of Japan, 116-D(1):71-78, 1996.
- [25] S. Hosoe and M. Araki (editorial). Design of Control Systems  $H_{\infty}$  control and its applications (in Japanese). Asakura Publishing Co., LTD., 1994.
- [26] J. Hozumi, H. Fujioka, and S. Hara.  $H_{\infty}$  control synthesis for robust servo systems an LMI approach –. *Proc. of Asian Control Conf.*, pages 33–36, 1994.

- [27] J. Hozumi, S. Hara, and H. Fujioka. Robust servo problem with  $H_{\infty}$  constraint. International Journal of Control, 66(6):803–823, 1997.
- [28] D. C. Hyland and D. S. Bernstein. The optimal projection equations for model reduction and the relationships among the methods of wilson. *IEEE Trans. Automatic Control*, 30(12):1201–1211, 1985.
- [29] M. Ikeda. Modern control (in Japanese). SICE seminar text, 1992.
- [30] Z. Iwai, A. Inoue, and S. Kawaji. Observer (in Japanese). Corona Publising Co., Ltd., 1988.
- [31] T. Iwasaki. LMI and Control (in Japanese). Shoukodo Publishing Co., LTD., 1997.
- [32] T. Iwasaki and R. E. Skelton. All controllers for the general  $H_{\infty}$  control problem. Automatica, 30(8):1307–1317, 1994.
- [33] T. Iwasaki and R. E. Skelton. Parametrization of all stabilizing controllers via quadratic lyapunov functions. J. Optimiz. Theory Appl., 85:291–307, 1995.
- [34] JAPAN E. M. Co., Ltd. An explanation note of the magnetic levitation system (in Japanese), 1995.
- [35] H. Kimura, Y. Lu, and R. Kawatani. On the structure of  $H_{\infty}$  control systems and related extensions. *IEEE Trans. Automatic Control*, 36(6):653–667, 1991.
- [36] D. J. N. Limebeer and G. D. Halikias. A controller degree bound for  $\mathcal{H}_{\infty}$  optimal control problems of the second kind. *SIAM J. Contr. and Opt.*, 26(3):647–677, 1988.
- [37] D. J. N. Limebeer and Y. S. Hung. An analysis of the pole-zero cancellations of the first kind. SIAM J. Contr. and Opt., 25(6):1457–1493, 1987.
- [38] D. J. N. Limebeer, E. M. Kasenally, and J. D. Perkins. On the design of robust two degree of freedom controllers. *Automatica*, 29(1):157–168, 1993.
- [39] K. Liu and T. Mita. Generalized  $H_{\infty}$  control theory. *Proc. of ACC*, pages 2245–2249, 1992.
- [40] K. Liu and T. Mita. On robust steady-state performance of feedback control systems (in Japanese). Trans. of SICE, 29(3):312–318, 1993.
- [41] D. G. Luenberger. Observers for multivariable systems. *IEEE Trans. Automatic Control*, 11(8):190–197, 1966.
- [42] D. G. Luenberger. An introduction to observers. *IEEE Trans. Automatic Control*, 16(6):596–602, 1971.

- [43] H. Maeda, S. Kodama, and Z. Z. Man.  $H_{\infty}$  control with prescribed stability domain (in Japanese). Trans. of SICE, 28(10):1194–1200, 1992.
- [44] H. Maeda and T. Sugie. System control theory for advanced control (in Japanese). Asakura Publishing Co., LTD., 1990.
- [45] J. B. Matson, T. Mita, and B. D. O. Anderson. Controller parametrization for nonstandard  $\mathcal{H}_{\infty}$  and  $\mathcal{H}_2$  problems. *Proc. of the Asian Control Conference*, pages 21–24, 1994.
- [46] M. Mercadal. Homotopy approach to optimal, linear quadratic, fixed architecture compensation. J. Guidance, 14(6):1224–1233, 1991.
- [47] T. Mita.  $H_{\infty}$  control (in Japanese). Shoukodo Publishing Co., LTD., 1994.
- [48] T. Mita, M. Hirata, and S. B. Villas-Boas. Design of  $H_{\infty}$  controllers for plants having poles on the  $j\omega$  axis –  $H_{\infty}$  motion control – (in Japanese). Trans. of IEE of Japan, 115-D(10):1253–1262, 1995.
- [49] T. Mita, K. Liu, and K. Kuriyama.  $H_{\infty}$  control with weighting functions having  $j\omega$  poles (in Japanese). Trans. of SICE, 29(11):1320–1329, 1993.
- [50] T. Mita and J. B. Matson. A nonstandard  $H_2$  problem and derivation of all  $H_2$  optimal controllers (in Japanese). Trans. of. SICE, 31(4):489–496, 1995.
- [51] T. Mita, J. B. Matson, and B. D. O. Anderson. A complete and simple parametrization of controllers for a nonstandard  $H_{\infty}$  problem. *Automatica*, 34(3):369–374, 1998.
- [52] B. C. Moore. Principal component analysis in linear systems: Controllability, observability, and model reduction. *IEEE Trans. Automatic Control*, 26(1):17–32, 1981.
- [53] D. Mustafa and K. Glover. Controller reduction by  $\mathcal{H}_{\infty}$  balanced truncation. *IEEE Trans. Automatic Control*, 36(6):668–682, 1991.
- [54] A. Ohara and T. Sugie. Control systems synthesis via convex optimization (in Japanese). Trans. of ISCIE, 38(3):139–146, 1994.
- [55] A. Packard and J. C. Doyle. The complex structured singular value. Automatica, 29(1):71–109, 1993.
- [56] L. Pernebo and L. M. Silverman. Model reduction via balanced state space representations. *IEEE Trans. Automatic Control*, 27(2):382–387, 1982.
- [57] S. Richter. A homotopy algorithm for solving the optimal projection equations for fixedorder dynamic compensation: Existance, convergence and global optimality. *Proc. ACC*, pages 1527–1531, 1987.

- [58] W. Rudin. Real and Complex Analysis. McGraw-Hill, 1987.
- [59] M. Sampei, H. Itoh, and T. Mita.  $H_{\infty}$  control of a plant with redundant degree of freedom (in Japanese). Proc. of 20th SICE Symposium on Control Theory, pages 101–104, 1991.
- [60] M. Sampei, T. Mita, and M. Nakamichi. An algebraic approach to  $H_{\infty}$  output feedback control problems. *Syst. Contr. Lett.*, 14:13–24, 1990.
- [61] J. Sefton and K. Glover. Pole/zero cancellations in the general  $H_{\infty}$  problem with reference to a two block design. Syst. Contr. Lett., 14:295–306, 1990.
- [62] R. E. Skelton. *Dynamic systems control*. New York: Wiley, 1988.
- [63] R. E. Skelton, T. Iwasaki, and K. Grigoriadis. A unified algebraic approach to linear control design. Taylor and Francis, 1998.
- [64] A. A. Stoorvogel. The  $H_{\infty}$  control problem : A state space approach. Prentice Hall, 1992.
- [65] A. A. Stoorvogel, A. Sabeli, and B. M. Chen. A reduced order observer based controller design for  $H_{\infty}$  optimization. *IEEE Trans. Automatic Control*, 39(2):355–360, 1994.
- [66] T. Sugie and T. Yoshikawa. General solution of robust tracking problem in two-degreeof-freedom control systems. *IEEE Trans. Automatic Control*, 31(6):552–554, 1986.
- [67] M. Vidyasagar. Control system synthesis: A factorization approach. The MIT Press, 1985.
- [68] Xin Xin, L. Guo, and C. Feng. Reduced-order controllers for continuous and discretetime singular  $H^{\infty}$  control problems based on LMI. *Automatica*, 32(11):1581–1585, 1996.
- [69] D. C. Youla, H. A. Jabr, and J. J. Bongiorno Jr. Modern wiener-hopf design of optimal controllers – part ii: The multivariable case. *IEEE Trans. Automatic Control*, 21(3):319– 338, 1976.
- [70] X. Yu and C. S. Hsu. Luenberger observer-based  $H_{\infty}$ -compensator design for nonstandard problems. *IEEE Trans. Automatic Control*, 41(9):1364–1367, 1996.
- [71] F. Zhang and S. Hosoe.  $H_{\infty}$  control problem with partial measurable states (in Japanese). Trans. of SICE, 29(2):169–178, 1993.
- [72] K. Zhou. Frequency-weighted  $\mathcal{L}_{\infty}$  norm and optimal hankel norm model reduction. *IEEE Trans. Automatic Control*, 40(10):1687–1699, 1995.
- [73] K. Zhou, J. C. Doyle, and K. Glover. Robust and Optimal Control. Prentice Hall, 1996.

# Appendix A Preliminary results

## A.1 Invariant zeros

The invariant zeros are characterized by the following lemma.

Lemma A.1.1 (Invariant zeros) Let us consider the system

$$T(s) = \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix},$$

and define a system matrix as follows

$$Q(s) \triangleq \left[ \begin{array}{c|c} A - sI & B \\ \hline C & D \end{array} \right].$$

A complex number  $z_0 \in \mathbb{C}$  is called an invariant zero of T(s) if it satisfies

$$\operatorname{rank}(Q(z_0)) < \operatorname{normal} \operatorname{rank}(Q(s)),$$

where the normal rank of Q(s) means the maximally possible rank of  $Q(s), s \in \mathbb{C}$ .

If D is a matrix of full rank, the invariant zeros are defined by the following three cases.

- (i) When D is full column rank, the invariant zeros of the system T(s) are the unobservable modes of the pair  $(A BD^{\dagger}C, D^{\perp}C)$ .
- (ii) When D is full row rank, the invariant zeros of the system T(s) are the uncontrollable modes of the pair  $(A BD^{\dagger}C, BD^{\perp})$ .
- (iii) When D is a square matrix, the invariant zeros of the system T(s) are the eigenvalues of the matrix  $A BD^{-1}C$ .

## A.2 Parametrization of stabilizing controllers

Let us consider the system

$$G(s) = \begin{bmatrix} G_{11}(s) & G_{12}(s) \\ G_{21}(s) & G_{22}(s) \end{bmatrix} = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix},$$
(A.1)

where  $(A, B_2)$  is stabilizable and  $(A, C_2)$  is detectable.

Lemma A.2.1 Choose the matrices F and H such that the matrices

$$A_F = A + B_2 F$$
$$A_H = A + HC_2$$

are both stable. Then the class of all the stabilizing controllers for  $G_{22}(s)$  is represented as follows:

$$K(s) = \mathcal{F}_l \left( \begin{bmatrix} A + B_2 F + HC_2 & -H & -B_2 \\ F & O & -I \\ -C_2 & I & O \end{bmatrix}, Q(s) \right), \quad Q(s) \in \mathcal{RH}_{\infty}.$$
(A.2)

The class of closed loop transfer functions  $G_{zw}(s) = \mathcal{F}_l(G(s), K(s))$  is represented as follows:

$$G_{zw}(s) = \mathcal{F}_l \left( \begin{bmatrix} A_F & -B_2F & B_1 & B_2 \\ O & A_H & B_H & O \\ \hline C_F & -D_{12}F & O & D_{12} \\ O & C_2 & D_{21} & O \end{bmatrix}, -Q(s) \right), \quad Q(s) \in \mathcal{RH}_{\infty},$$
(A.3)

where  $B_H$  and  $C_F$  are defined as follows.

$$B_H = B_1 + HD_{21}$$
$$C_F = C_1 + D_{12}F$$

#### A.3 A matrix equation

Let us define the matrix equation as follows

$$PA_1 + A_2 P = B, (A.4)$$

where  $A_1 \in \mathbb{R}^{n \times n}$  and  $A_2 \in \mathbb{R}^{m \times m}$  are square matrices.

**Lemma A.3.1** If  $A_1$  and  $A_2$  satisfy

$$\lambda_i(A_1) + \lambda_j(A_2) \neq 0, \ \forall i \in [1, 2, \dots, n], \forall j \in [1, 2, \dots, m]$$
(A.5)

the matrix equation in (A.4) has a unique solution. Specifically in the case B = O, the unique solution is P = O.

#### A.4 Star product

Let us consider the systems:

$$G(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & D_{11} & D_{12} \\ \hline C_2 & D_{21} & D_{22} \end{bmatrix}, \quad K(s) = \begin{bmatrix} A_K & B_{K1} & B_{K2} \\ \hline C_{K1} & D_{K11} & D_{K12} \\ \hline C_{K2} & D_{K21} & D_{K22} \end{bmatrix}.$$

Then the star product of G(s) and K(s) are defined as

$$G(s) \star K(s) = \left[ \begin{array}{c|c} \bar{A} & \bar{B} \\ \hline \bar{C} & \bar{D} \end{array} \right],$$

where

$$\begin{split} \bar{A} &= \begin{bmatrix} A + B_2 \tilde{R}^{-1} D_{K11} C_2 & B_2 \tilde{R}^{-1} C_{K1} \\ B_{K1} R^{-1} C_2 & A_K + B_{K1} R^{-1} D_{22} C_{K1} \end{bmatrix} \\ \bar{B} &= \begin{bmatrix} B_1 + B_2 \tilde{R}^{-1} D_{K11} D_{21} & B_2 \tilde{R}^{-1} D_{K12} \\ B_{K1} R^{-1} D_{21} & B_{K2} + B_{K1} R_{-1} D_{22} D_{K12} \end{bmatrix} \\ \bar{C} &= \begin{bmatrix} C_1 + D_{12} D_{K11} R^{-1} C_2 & D_{12} \tilde{R}^{-1} C_{K1} \\ D_{K21} R^{-1} C_2 & C_{K2} + D_{K21} R^{-1} D_{22} C_{K1} \end{bmatrix} \\ \bar{D} &= \begin{bmatrix} D_{11} + D_{12} D_{K11} R^{-1} D_{21} & D_{12} \tilde{R}^{-1} D_{K12} \\ D_{K21} R^{-1} D_{21} & D_{K22} + D_{K21} R^{-1} D_{22} D_{K12} \end{bmatrix} \\ R &= I - D_{22} D_{K11}, \quad \tilde{R} = I - D_{K11} D_{22}. \end{split}$$

# A.5 Inner function

A transfer function N(s) is inner if  $N \in \mathcal{RH}_{\infty}$  and  $N^T(-s)N(s) = I$ .

**Lemma A.5.1** Let us consider the system N(s) as follows:

$$N(s) = \begin{bmatrix} A & B \\ \hline C & D \end{bmatrix}.$$

Suppose that D is of full column rank, (A, C) is detectable. Then N(s) is inner if there exists  $P \ge O$  such that

$$\begin{cases} PA + A^T P + C^T C = O\\ D^T C + B^T P = O\\ D^T D = I \end{cases}$$
(A.6)

## A.6 Lossless system

Let us consider the stable system

$$\Xi(s) = \begin{bmatrix} \Xi_{11}(s) & \Xi_{12}(s) \\ \Xi_{21}(s) & \Xi_{22}(s) \end{bmatrix} \in \mathcal{RH}_{\infty}^{m \times p}, \quad m \ge p.$$

Suppose that  $\Xi(s)$  satisfies

$$\begin{cases} \Xi^{T}(-s)\Xi(s) = I\\ \Xi_{22}(\infty) = O\\ \det(\Xi_{21}(j\omega)) \neq 0; \forall \omega \end{cases}$$
(A.7)

then the system  $\Xi(s)$  is called lossless.

**Lemma A.6.1** Suppose that the system  $\Xi(s)$  is lossless, and that the system Q(s) is proper. Then the following items are equivalent.

1. A closed loop system  $(\Xi(s), Q(s))$  is internally stable and Q(s) satisfies

$$|\mathcal{F}_l\left(\Xi(s),Q(s)\right)\|_{\infty} < 1$$

2. Q(s) satisfies

$$\|Q(s)\|_{\infty} < 1.$$

#### A.7 Disturbance feedforward problem

Let us consider the standard  $H_{\infty}$  problem for

$$G(s) = \begin{bmatrix} A & B_1 & B_2 \\ \hline C_1 & O & D_{12} \\ C_2 & D_{21} & O \end{bmatrix},$$
 (A.8)

where the assumptions A1, A2 and A3 are satisfied. If the matrix  $D_{21}$  is nonsingular and the matrix  $A - B_1 D_{21}^{\dagger} C_2$  is stable, the  $H_{\infty}$  problem is called DF (Disturbance Feedforward) problem. The solution of DF problem is listed in the following lemma.

Lemma A.7.1 The DF problem is solvable if and only if an ARE:

$$X\left(A - B_2 D_{12}^{\dagger} C_1\right) + \left(A - B_2 D_{12}^{\dagger} C_1\right)^T X + X\left\{B_1 B_1^T - B_2 D_{12}^{\dagger} \left(B_2 D_{12}^{\dagger}\right)^T\right\} X + \left(D_{12}^{\perp} C_1\right)^T D_{12}^{\perp} C_1 = O$$

has a nonnegative definite stabilizing solution X. If the condition is satisfied, the  $H_{\infty}$  controller is represented as follows:

$$K(s) = \mathcal{F}_l \left( \begin{bmatrix} A - B_1 D_{21}^{-1} C_2 + B_2 F_{\infty} & B_1 D_{21}^{-1} & B_2 \Sigma \\ F_{\infty} & O & \Sigma \\ - (D_{21}^{-1} C_2 + B_1^T X) & D_{21}^{-1} & O \end{bmatrix}, N(s) \right),$$
(A.9)

where  $N(s) \in \mathcal{BH}_{\infty}$ , and  $F_{\infty}$  and  $\Sigma$  are defined as

$$F_{\infty} = -D_{12}^{\dagger}C_1 - D_{12}^{\dagger} \left(B_2 D_{12}^{\dagger}\right)^T X$$
$$\Sigma = \left(D_{12}^T D_{12}\right)^{-\frac{1}{2}}.$$

# A.8 ARE (Algebraic Riccati Equation)

Let a matrix  $A \in \mathbb{R}^{n \times n}$  and a symmetric matrices  $R \in \mathbb{R}^{n \times n}$  and  $Q \in \mathbb{R}^{n \times n}$  be given. If a solution of the following ARE:

$$XA + A^T X + XRX + Q = O (A.10)$$

stabilizes the matrix

$$A_X = A + RX,$$

then the solution X is called a stabilizing solution of the ARE (A.10).

**Definition A.8.1** Associated with the ARE (A.10), let us define a Hamiltonian matrix:

$$H_X \triangleq \begin{bmatrix} A & R \\ -Q & -A^T \end{bmatrix} \in \mathbb{R}^{2n \times 2n}.$$
(A.11)

**Lemma A.8.1** If  $H_X$  doesn't have poles on the imaginary axis, then  $H_X$  has n stable eigenvalues and n unstable eigenvalues.

Suppose that  $H_X$  doesn't have poles on the imaginary axis. Then let  $\Lambda_- \in \mathbb{R}^{n \times n}$  be a matrix whose eigenvalues are all of stable eigenvalues of  $H_X$ . A full column rank matrix of  $V = \begin{bmatrix} X_1 \\ X_2 \end{bmatrix} \text{ satisfies}$ 

$$H_X V = V \Lambda_-, \tag{A.12}$$

where V is a basis of the mode  $\Lambda_{-}$ . Then if det  $(X_1) \neq 0$ , we denote

$$H_X \in \text{Dom}\left(\text{Ric}\right)$$
 (A.13)

$$H_X \in \text{Dom} (\text{Ric})$$

$$\text{Ric} (H_X) = X_2 X_1^{-1}.$$
(A.13)
(A.14)

**Lemma A.8.2** The stabilizing solution of the ARE (A.10) is given by  $X = \text{Ric}(H_X)$ .

#### A.9 Solution to Full control problem

**Lemma A.9.1** In order that the FC problem is solvable, it is necessary that the following ARE:

$$Y\left(A - B_{1}D_{21}^{\dagger}C_{2}\right)^{T} + \left(A - B_{1}D_{21}^{\dagger}C_{2}\right)Y + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}Y + B_{1}D_{21}^{\perp}\left(B_{1}D_{21}^{\perp}\right)^{T} = O$$
(A.15)

has a nonnegative definite stabilizing solution Y. By stabilizing, we mean that the following matrix is stable.

$$A_{Y} \triangleq A - B_{1}D_{21}^{\dagger}C_{2} + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}$$

**Proof.** See the paper [11].

## A.10 Reduction mode of an ARE for the standard problem

Let us consider an ARE:

$$Y\left(A - B_{1}D_{21}^{\dagger}C_{2}\right)^{T} + \left(A - B_{1}D_{21}^{\dagger}C_{2}\right)Y + Y\left\{C_{1}^{T}C_{1} - \left(D_{21}^{\dagger}C_{2}\right)^{T}D_{21}^{\dagger}C_{2}\right\}Y + B_{1}D_{21}^{\perp}\left(B_{1}D_{21}^{\perp}\right)^{T} = O, \quad (A.16)$$

which is required to be solved in the standard  $H_{\infty}$  problem. For this ARE, the following lemma holds.

**Lemma A.10.1** Assume that the ARE in (A.16) has a positive definite stabilizing solution Y, i.e., there exists a solution Y > O for the ARE and the solution stabilizes the matrix:

$$A_{Y_s} \triangleq A - B_1 D_{21}^{\dagger} C_2 + Y \left\{ C_1^T C_1 - \left( D_{21}^{\dagger} C_2 \right)^T D_{21}^{\dagger} C_2 \right\}.$$

Then it is shown that

$$\forall U, \det \left( UU^T \right) \neq 0; \begin{cases} U\left( A - B_1 D_{21}^{\dagger} C_2 \right) = \Lambda_- U, \lambda_i \left( \Lambda_- \right) < 0, \forall i \\ UB_1 D_{21}^{\perp} = O \end{cases}$$
$$\ker \left( Y^T \right) = \operatorname{range} \left( U^T \right).$$

**Proof.** Let U be a full row rank matrix which satisfies UY = O. Pre-multiplying by U and post-multiplying by  $U^T$ , the ARE is written as

$$UYA_{Y}^{T}U^{T} + U\left(A - B_{1}D_{21}^{\dagger}C_{2}\right)YU^{T} + UB_{1}D_{21}^{\perp}\left(UB_{1}D_{21}^{\perp}\right)^{T} = O,$$

hence  $\boldsymbol{U}$  satisfies

$$UB_1D_{21}^{\perp} = O$$

Again, pre-multiplying the ARE by U, it can be seen that U satisfies

$$U\left(A - B_1 D_{21}^{\dagger} C_2\right) Y = O,$$

hence  $U\left(A - B_1 D_{21}^{\dagger} C_2\right)$  can be represented as

$$U\left(A - B_1 D_{21}^{\dagger} C_2\right) = \Lambda_s U,$$

where  $\Lambda_s$  is an appropriate nonsingular matrix. Then, for the stable matrix  $A_Y$  the matrix U satisfies

$$UA_Y = \Lambda_s U.$$

From this equation, it can be seen that the matrix  $\Lambda_s$  is a stable matrix because the matrix U is of full row rank. Thus, the lemma has been proved.

#### A.11 Solutions to the mixed-sensitivity problem

The solutions to the mixed-sensitivity problems, which are the  $H_{\infty}$  control problems for the generalized plant  $G_{ms}(s)$  in (3.6) described in Figure 3.2, are listed according to some conditions.

Lemma A.11.1 (A case where  $W_s(s) \in \mathcal{RH}_{\infty}$  and P(s) includes no  $j\omega$ -poles) Assume that  $W_s(s) \in \mathcal{RH}_{\infty}$  and  $P(j\omega) \neq \infty, \forall \omega \in \mathbb{R}$  hold. If the AREs:

$$X\left(A - B_{2}D_{12}^{\dagger}C_{1}\right) + \left(A - B_{2}D_{12}^{\dagger}C_{1}\right)^{T}X + X\left\{B_{1}B_{1}^{T} - B_{2}D_{12}^{\dagger}\left(B_{2}D_{12}^{\dagger}\right)^{T}\right\}X + \left(D_{12}^{\perp}C_{1}\right)^{T}D_{12}^{\perp}C_{1} = O \qquad (A.17)$$

$$Y\left(A - B_{1}\tilde{D}_{21}^{\dagger}\tilde{C}_{2}\right)^{T} + \left(A - B_{1}\tilde{D}_{21}^{\dagger}\tilde{C}_{2}\right)Y + Y\left\{C_{1}^{T}C_{1} - \left(\tilde{D}_{21}^{\dagger}\tilde{C}_{2}\right)^{T}\tilde{D}_{21}^{\dagger}\tilde{C}_{2}\right\}Y + B_{1}\tilde{D}_{21}^{\perp}\left(B_{1}\tilde{D}_{21}^{\perp}\right)^{T} = O$$
(A.18)

have the stabilizing solutions  $X \ge O$  and  $Y \ge O$  and they satisfy an inequality  $\rho(XY) < 1$ , then the class of the  $H_{\infty}$  controllers is given as follows.

$$K_{ms}(s) = \mathcal{F}_l \left( \begin{bmatrix} \hat{A} & -ZH_{\infty} & Z\hat{B}_2 \left(D_{12}^T D_{12}\right)^{-\frac{1}{2}} \\ F_{\infty} & O & \left(D_{12}^T D_{12}\right)^{-\frac{1}{2}} \\ -\left(\tilde{D}_{21}\tilde{D}_{21}^T\right)^{-\frac{1}{2}}\hat{C}_2 & \left(\tilde{D}_{21}\tilde{D}_{21}^T\right)^{-\frac{1}{2}} & O \end{bmatrix}, N \right), \quad (A.19)$$

where  $N(s) \in \mathcal{BH}_{\infty}$  is a free parameter, and the other parameters are defined as

$$\hat{A} = A + B_1 B_1^T X + B_2 F_{\infty} + Z H_{\infty} \hat{C}_2$$
$$\hat{B}_2 = B_2 + Y C_1^T D_{12}$$
$$\hat{C}_2 = \tilde{C}_2 + \tilde{D}_{21} B_1^T X$$
$$F_{\infty} = -D_{12}^{\dagger} C_1 - D_{12}^{\dagger} \left( B_2 D_{12}^{\dagger} \right)^T X$$
$$H_{\infty} = -B_1 \tilde{D}_{21}^{\dagger} - Y \left( \tilde{D}_{21}^{\dagger} \tilde{C}_2 \right)^T \tilde{D}_{21}^{\dagger}$$
$$Z = (I - Y X)^{-1}.$$

Lemma A.11.2 (A case where  $W_s(s)$  includes  $j\omega$ -poles and P(s) includes no  $j\omega$ -poles) [49] If the ARE in (A.17) has a stabilizing solution  $X \ge O$ , the ARE in (A.18) has a pseudostabilizing solution  $Y \ge O$  and they satisfy the inequality  $\rho(XY) < 1$ , then by using these solutions X and Y the class of the  $H_{\infty}$  controllers is given with the expression in (A.19).

**Lemma A.11.3 (A case where** P(s) **includes**  $j\omega$ -**poles)** [48] Assume that P(s) has  $j\omega$ -poles, and can be factorized as follows

$$P(s) = \tilde{P}(s)a(s),$$

where

$$\hat{P}(j\omega) < \infty, \forall \omega \in \mathbb{R}, \quad a^{-1}(s) \in \mathcal{RH}_{\infty}.$$

In Figure 3.2, let us select  $W_s(s)$  as follows

$$W_S(s) = a(s)\hat{W}_s(s),$$

where  $\hat{W}(s)$  has  $j\omega$ -poles which are needed to be included in the controller, and satisfies

$$\hat{W}(\infty) = 0, \quad \hat{W}(s) < \infty, \forall s \in \{s : \operatorname{Re}(s) > 0\}.$$

Then, by factorizing P(s) as in Figure 3.7 let us construct the generalized plant  $\hat{G}_{ms}(s)$ .

If there exists  $\hat{K}_{ms}(s)$  which is an  $H_{\infty}$  controller for  $\hat{G}_{ms}(s)$ , the  $H_{\infty}$  controller for the original plant  $G_{ms}(s)$  exists and is given by  $a^{-1}(s)\hat{K}_{ms}(s)$ .

# Appendix B

# Publications

## **Journal Papers**

- 1. T. Watanabe, K. Yasuda and R. Yokoyama: A reduction method of  $H_{\infty}$  controller based on truncation of Youla parameter, *Trans. of SICE*, Vol. 31,No. 8, pp. 1080-1088 (1995) (in Japanese).
- 2. T. Watanabe, K. Yasuda and R. Yokoyama: Reduced order solution of non-standard  $H_{\infty}$  control problem, *Trans. of SICE*, Vol. 32, No. 1, pp. 16-25 (1996) (in Japanese).
- T. Watanabe, S. Akimoto and K. Yasuda: Simultaneous design of Two-Degree-of-Freedom controller via linear matrix inequality approach, *Trans. IEE of Japan*, Vol. 118, No. 3, pp. 310-319 (1998) (in Japanese).
- 4. T. Watanabe and K. Yasuda: Trade-off analysis of TDF control systems sharing common dynamics and simultaneous design of low-order TDF controller via non-standard  $H_{\infty}$  control, *Trans. IEE of Japan*, Vol. 118, No. 6, pp. 917-926 (1998) (in Japanese).
- 5. T. Watanabe and K. Yasuda: A synthesis of integral type control systems based on non-standard  $H_{\infty}$  control, *Trans. of SICE*, Vol. 34, No. 12, pp. 1822-1830 (1998) (in Japanese).
- 6. T. Watanabe and K. Yasuda: Reduced order design of integral-type, TDF controller based on a non-standard  $H_{\infty}$  control, submitted to *International Journal of Control* (1998)

## Conference Papers (reviewed)

 T. Watanabe, K. Yasuda and R. Yokoyama: Balanced truncation preserving poles in a specified disk, *Proc. of the 34th SICE Annual Conf.*, International Session Papers, pp. 1387-1390 (1995)

- T. Watanabe, K. Yasuda and R. Yokoyama: Balanced truncation preserving poles in a specified disk and its application to the reduction of H<sub>∞</sub> controller, Proc. of 22nd IEEE International Conference on Industrial Electronics, Control and Instrumentation, pp. 1359-1364, Taipei (1996)
- 3. T. Watanabe, K. Yasuda: Reduced order solutions to nonstandard  $H_2$  and  $H_{\infty}$  control problems, *Proc. of 35th IEEE Conf. on Decision and Control*, pp. 715-720, Kobe (1996)

#### Domestic Conference Papers (not reviewed)

- T. Watanabe, K. Yasuda and R. Yokoyama: A fundamental study on robust control for power systems –A design method of robust servo systems –, *Proc. of the Fourth Annual Conf. of Power and Energy Society, IEE of Japan*, II-No. 102, pp. 3-4 (1993) (in Japanese).
- T. Watanabe, K. Yasuda and R. Yokoyama: Robust performance of servo systems, Preprints of the 36th Japan Joint Automatic Control Conf., No. 1018, pp. 47-48 (1993) (in Japanese).
- 3. T. Watanabe, K. Yasuda and R. Yokoyama: A model reduction method of  $H_{\infty}$  controller, *Preprints of the 16th SICE Symposium on Dynamical System Theory*, pp. 35-38 (1993) (in Japanese).
- 4. T. Watanabe, K. Yasuda and R. Yokoyama: Order reduction of a sub-sptimal  $H_{\infty}$  controller, *National Convention Records, IEE of Japan*, No. 1710, pp. 35-36 (1994) (in Japanese).
- 5. T. Watanabe, K. Yasuda and R. Yokoyama: A reduction method of  $H_{\infty}$  controller, Preprints of the 23rd SICE Symposium on Control Theory, pp. 125-130 (1994) (in Japanese).
- T. Watanabe, K. Yasuda and R. Yokoyama: Balanced truncation preserving the poles in a specified disk, *National Convention Records, IEE of Japan*, No. 617, pp. 141-142 (1995) (in Japanese).
- T. Watanabe, K. Yasuda and R. Yokoyama: Parametrization on reduced order singular *H*<sub>∞</sub> controller, *National Convention Records*, *IEE of Japan*, No. 622, pp. 150-151 (1995) (in Japanese).
- T. Watanabe, K. Yasuda and R. Yokoyama: Parametrization and reduction of the nonstandard H<sub>∞</sub> controller, *Proc. of National Conf. on Systems and Control (IEEJ)*, SC-95-19, pp. 19-28 (1995) (in Japanese).

- 9. T. Watanabe, K. Yasuda and R. Yokoyama: Reduced order solutions of nonstandard  $H_{\infty}$  control problems and parametrization of reduced order nonstandard  $H_{\infty}$  controller, *Preprints of the 24th SICE Symposium on Control Theory*, pp. 45-50 (1995) (in Japanese).
- T. Watanabe, K. Yasuda and R. Yokoyama: On the algebraic riccati equation in the nonstandard H<sub>∞</sub>/H<sub>2</sub> control problems, National Convention Records, IEE of Japan, No. 676, pp. 216-217 (1996) (in Japanese).
- T. Watanabe, K. Yasuda: On two types of reduced order controllers to nonstandard H<sub>2</sub> control problem, Proc. of 25th SICE Symposium on Control Theory, pp. 157-162 (1996)
- 12. T. Watanabe and K. Yasuda: A Synthesis of Integral Type Control Systems Based on Non-standard  $H_2$ ,  $H_{\infty}$  Controls, *Proc. of the Electronics, Information and Systems Conference (IEEJ)*, pp. 323-328 (1996) (in Japanese).
- R. Kasuga, T. Watanabe, K. Yasuda and R. Yokoyama: Frequency separate control of plant with redundant actuators by nonstandard H<sub>2</sub> control, *Proc. of the 35th SICE Annual Conference Domestic Session*, pp.77-78, (1996) (in Japanese).
- T. Watanabe and K. Yasuda: A Synthesis of Integral Type Two-Degree-of-Freedom Controller via Non-standard H<sub>∞</sub> Control, Proc. of National Conf. on Systems and Control (IEEJ), SC-97-27, pp. 37-44 (1997) (in Japanese).
- 15. T. Watanabe and K. Yasuda: Simultaneous Design of Low Order Two-Degree-of-Freedom Controller via Non-standard  $H_{\infty}$  Control, *Proc. of National Conf. on Industorial Instrument and Control (IEEJ)*, IIC-97-30, pp. 25-33 (1997) (in Japanese).
- 16. T. Watanabe and K. Yasuda: Simultaneous Design of Two-Degree-of-Freedom Controller via Non-standard  $H_{\infty}$  Control, *National Convention Records (IEEJ)*, No. 657, pp. 170-171 (1997) (in Japanese).
- 17. T. Watanabe, K. Yasuda: Simultaneous Design of Feedforward and Feedback Controllers via Non-standard  $H_{\infty}$  Control, *Proc. of 26th SICE Symposium on Control Theory*, pp. 99-102 (1997)
- 18. T. Watanabe and K. Yasuda: A Synthesis of Integral Type Control Systems Based on Non-standard  $H_{\infty}$  Control, *Proc. of 26th SICE Symposium on Control Theory*, pp.189-194 (1997) (in Japanese).
- R. Kasuga, T. Watanabe and K. Yasuda: A study on H<sub>∞</sub> Reliable Control with Passive Redundancy, Proc. of the 36th SICE Annual Conference Domestic Session, pp. 191-192 (1997) (in Japanese).
- S. Akimoto, T. Watanabe and K. Yasuda: Simultaneous Design of Two Degree of Freedom Controller via LMI, Proc. of the 36th SICE Annual Conference Domestic Session, pp. 811-812 (1997) (in Japanese).
- R. Kasuga, T. Watanabe and K. Yasuda: Reliable H<sub>∞</sub> Control System Design via Matrix Inequality Approach, Proc. of National Conf. on Industrial Instrument and Control (IEEJ), IIC-97-61, pp. 27-31 (1997) (in Japanese).
- T. Watanabe, S. Akimoto and K. Yasuda, Trade-off Analysis of Low-order TDF Control System and Its Multiobjective Design, *National Convention Records (IEEJ)*, No. 621, pp. 143-144 (1998) (in Japanese).
- R. Satoh, T. Watanabe and K. Yasuda, Controller design considering approximated variation in controller, *Proc. of the 37th SICE Annual Conference Domestic Session*, pp. 31-32, (1998) (in Japanese).